## Helping people to help themselves: the development and evaluation of a non-clinical, peer-supported eTherapy model in the management of anxiety and depression in adults.

Nicola Mary Lidbetter

School of Health and Society

University of Salford

Submitted in Partial Fulfilment of the Requirements of the

**Degree of Doctor of Philosophy, 2021** 

## Contents

## Page Number

List of	tables		
List of	figures		
Abbrev	viations		
Ackno	wledgements 10-11		
Declar	ation12		
Abstra	ct13		
Papers	included in the Portfolio of Published Works 14-15		
Suppor	rting articles/books included in the Portfolio of Published Works16-18		
Introdu	action19-20		
Section	Section One: Definition and overview of anxiety and depression		
1.1	Common mental health disorders and terminology21		
1.2	Anxiety 21-27		
1.3	Depression		
1.4	Prevalence of anxiety and depression		
1.5	Treatment of anxiety and depression in England 31-36		
1.6	The role of the Third Sector in the delivery of IAPT services		
1.7	Section discussion		
1.8	Section summary		

## Section Two: eTherapy

2.1	Definition of eTherapy	39-40
2.2	eTherapy in the management of anxiety and depression	40-50

2.3	eTherapy: clinical effectiveness
2.4	eTherapy: acceptability 53-56
2.5	eTherapy: service delivery and peer support
2.6	Section discussion61-63
2.7	Section summary63
<u>Sect</u>	tion Three
3.1	Background and rationale for the published works
3.2	Thesis aim and objectives
3.3	Contribution of the included paper 68-76
3.4	The non-clinical, peer-supported eTherapy model
3.5	Section summary
3.5 <u>Sect</u>	Section summary
<ul><li>3.5</li><li>Sect</li><li>4.1</li></ul>	Section summary
<ul> <li>3.5</li> <li>Sect</li> <li>4.1</li> <li>4.2</li> </ul>	Section summary
<ul> <li>3.5</li> <li>Sect</li> <li>4.1</li> <li>4.2</li> <li>4.3</li> </ul>	Section summary
<ul> <li>3.5</li> <li>Sect</li> <li>4.1</li> <li>4.2</li> <li>4.3</li> <li>4.4</li> </ul>	Section summary
<ul> <li>3.5</li> <li>Sect</li> <li>4.1</li> <li>4.2</li> <li>4.3</li> <li>4.4</li> <li>Sect</li> </ul>	Section summary
<ul> <li>3.5</li> <li>Sect</li> <li>4.1</li> <li>4.2</li> <li>4.3</li> <li>4.4</li> <li>Sect</li> <li>5.1</li> </ul>	Section summary
<ul> <li>3.5</li> <li>Sect</li> <li>4.1</li> <li>4.2</li> <li>4.3</li> <li>4.4</li> <li>Sect</li> <li>5.1</li> <li>5.2</li> </ul>	Section summary

5.4	Reflections 128-131
5.5	Section Summary131-132
<u>Sect</u> i	ion Six
6.1	Key thesis outcomes
6.2	Recommendations for future research and practice134-136
6.3	Conclusions
<u>Refe</u>	<u>rences</u>
<u>App</u>	<u>endices</u>
Арр	endix 1: Papers included in the Portfolio of Published Works
Арр	endix 2: Letters from collaborating authors confirming contribution200-205

### List of tables

## Page Number

Table 1	Key signs and symptoms of DSM-5 <sup>®</sup> (APA, 2013) and
	ICD-10 (WHO, 2017b) anxiety disorders 23-27
Table 2	Current diagnostic criteria for Major Depressive Episode
	(MDE) according to DSM-5® (APA, 2013) criteria 28-29
Table 3	Stepped-care model: a combined summary for common
	mental health disorders (NICE, 2011)
Table 4:	Key terms associated with eTherapy40
Table 5:	eTherapy origins (Andersson, 2018a)42
Table 6:	Key IAPT recovery and associated definitions
	(Gellatly et al., 2018)44
Table 7:	Key eTherapy, CBT focused programmes used in the
	treatment of anxiety and depression
Table 8:	Peer support terms 57
Table 9:	12 principles of peer support (Basset et al., 2010) 58
Table 10:	Details of the papers and the contributions made
Table 11:	eTherapy service exclusion criteria78

List of figures

 Figure 1:
 The non-clinical, peer-supported, eTherapy model
 77

#### Abbreviations

ACT	Acceptance and Commitment Therapy
AI	Artificial Intelligence
APA	American Psychiatric Association
BA	Behavioural Activation
BDD	Body Dysmorphic Disorder
BFO	Breaking Free Online
BME	Black & Minority Ethnic
ВtВ <sup>тм</sup>	Beating the Blues
СВТ	Cognitive Behavioural Therapy
сСВТ	Computerised Cognitive Behavioural Therapy
CMD	Common Mental Disorder
CORE-OM	Clinical Outcomes in Routine Evaluation-Outcome
	Measure
dCBT	Digital Cognitive Behavioural Therapy
DHTs	Digital Health Technologies
DSM-5®	Diagnostic and Statistical Manual of Mental Disorders
	5 <sup>®</sup>
eTC	eTherapy Co-ordinator
GAD	Generalised Anxiety Disorder

GAD-7	Generalised Anxiety Disorder assessment (7 item
	measure)
GP	General Practitioner
IAB	Impact Assessment Briefing
IAPT	Improving Access to Psychological Therapies
IAPT MDS	Improving Access to Psychological Therapies
	Minimum Data Set
ICD-10/11	International Classification of Diseases 10 <sup>th</sup> /11 <sup>th</sup> Revision
IPT	Interpersonal Therapy
ISRII	International Society for Research on Internet
	Interventions
LGBTQQIAAP	Lesbian, Gay, Bisexual, Transgender, Queer,
	Questioning, Intersex, Allies, Asexual, Pansexual
LLTTF	Living Life to the Full
LOCF	Last Observation Carried Forward
MADD	Mixed Anxiety Depressive Disorder
MDD	Major Depressive Disorder
MDE	Major Depressive Episode
NHS	National Health Service

NICE	National Institute for Health and Care Excellence
	(formerly National Institute of Clinical Excellence)
OCD	Obsessive Compulsive Disorder
РСТQ-Р	Perceptions of Computerized Therapy Questionnaire -
	Patient Version
PHQ-9	Patient Health Questionnaire (9 item measure)
PTSD	Post-Traumatic Stress Disorder
PWP	Psychological Wellbeing Practitioner
PSW	Peer Support Worker
RCT	Randomised Controlled Trial
SAD	Social Anxiety Disorder
SOP	Service Operating Procedure
TAU	Treatment as Usual
TD-cCBT	Transdiagnostic computerised Cognitive Behavioural
	Therapy
TS	Third Sector
TSO	Third Sector Organisation
VR	Virtual Reality
WSAS	Work and Social Adjustment Scale
WHO	World Health Organization

#### Acknowledgements

This thesis has significance in that it completes a period of study commenced many years ago, which was cut short due to the onset of agoraphobia in my early 20s. As a direct consequence, I sadly had to discontinue a PhD level in the field of renal physiology, causing my life to dramatically change course. Unwittingly, I ended up in the field of mental health where I have remained to date, working in senior management roles, though more recently, I have ventured into the clinical world; training in Cognitive Behavioural Therapy (CBT), Compassion Focused Therapy (CFT), Eye Movement Desensitisation and Reprocessing (EMDR) and counselling.

Alongside being Chief Executive of Anxiety UK for over 25 years, I founded the user-led charity, Self Help Services, in 1995, with the aim of providing support for those experiencing anxiety and depression, through provision of accessible, peer-supported services. Throughout my career, I have instigated and overseen the development and delivery of a range of services from peer-led, drop-in support groups, primary care clinical mental health clinics, and regional crisis services – "The Sanctuary," – through to a nationally recognised eTherapy model, which is the subject of this thesis. I am passionate about ensuring that those with anxiety and depression obtain timely and appropriate access to support; something that was not available to me when I first began experiencing agoraphobia.

This thesis would not have been possible without the support and encouragement of my family: Pete, Natalia, and Kingsley. They have consistently motivated me, over the years, to make the best out of my challenges with anxiety; driving me to use this experience for the benefit of others.

Sincere thanks must also be given to my supervisory team, who have been a constant source of guidance, encouragement, and inspiration over the period of study. On this, I would like

to say a special thank you to Dr Elaine Beaumont, Professor Alison Brettle, and Dr Deborah Robertson; I am extremely grateful for your support.

#### Declaration

Signed:

This thesis includes a portfolio of publications that have been published in peer-reviewed journals, found in Appendix 1. The author's original contribution, verified by the collaborating authors, can be found in Appendix 2.

Date: 26<sup>th</sup> February 2021

#### Abstract

eTherapy interventions have widened access to evidence-based psychological therapies, particularly Cognitive Behavioural Therapy (CBT) since their introduction to primary care mental health services over a decade ago. Whilst the effectiveness of eTherapy programmes has been established and, to a degree, the acceptability of eTherapy has been strong, there are a paucity of studies undertaken in real-world settings. Even more scarce is research on service delivery models that utilise non-clinicians and instead, individuals with lived experience of mental health conditions, in the provision of support.

This portfolio of seven published works and thirteen supporting publications - two books, one book chapter, eight papers and two articles - makes a unique contribution to eTherapy literature by detailing the development and evaluation of a non-clinical, peer-supported model of eTherapy in the treatment of adults experiencing anxiety and depression in a real-world setting.

Collectively, the publications provide a body of knowledge that suggest that this novel *model* of pragmatic eTherapy service delivery is effective, acceptable, and capable of generating results equivalent to those generated by low intensity Improving Access to Psychological Therapy (IAPT) services. Furthermore, this *model* supports the widening of access to services and provision of an evidence-based, much-needed treatment for those affected by anxiety and depression, as well as for clients affected by sleep and dual diagnosis issues.

13

#### Papers included in the Portfolio of Published Works

Throughout the portfolio, research papers included in the submission will be cited in bold (e.g., **Lidbetter & Bunnell, 2013**) to distinguish such papers from other citations. The research papers listed below can be found in Appendix 1.

- Cavanagh, K., Seccombe, N., Lidbetter, N., & Bunnell, D. (2011a). Supported, serviceuser led, computerised cognitive behavioural therapy (CCBT) self-help clinics. *Journal* of Public Mental Health, 10(4), 225-233.
- (2) Cavanagh, K., Seccombe, N., & Lidbetter, N. (2011b). The Implementation of Computerized Cognitive Behavioural therapies in a Service User-Led, Third Sector Self Help clinic. *Behavioural and Cognitive Psychotherapy*, 39(4), 427-442.
- (3) Lidbetter, N., & Bunnell, D. (2013). Self Help Services: helping people to help themselves. *Mental Health and Social Inclusion*, 17(2), 76-81.
- (4) Elison, S., Ward, J., Davies, G., Lidbetter, N., Hulme, D., & Dagley, M. (2014). An outcomes study of eTherapy for dual diagnosis using Breaking Free Online. *Advances in Dual Diagnosis*, 7(2), 52-62.
- (5) Elison, S., Ward, J., Williams, C., Espie, C., Davies, G., Dugdale, S., Ragan, K., Chisnall, L., Lidbetter, N. & Smith, K. (2017). Feasibility of a UK community-based, eTherapy mental health service in Greater Manchester: repeated-measures and between-groups study of 'Living Life to the Full Interactive,' 'Sleepio<sup>TM</sup>' and 'Breaking Free Online' at 'Self Help Services.' *BMJ Open*, 7(7).
- (6) Luik, A. I., Bostock, S., Chisnall, L., Kyle, S. D., Lidbetter, N., Baldwin, N., & Espie, C. A. (2017). Treating depression and anxiety with digital cognitive behavioural therapy for insomnia: a real world NHS evaluation using standardized outcome measures. *Behavioural and Cognitive Psychotherapy*, 45(1), 91-96.

(7) Gellatly, J., Chisnall, L., Seccombe, N., Ragan, K., Lidbetter, N., & Cavanagh, K.
(2018). @ Home eTherapy service for people with common mental health problems: an evaluation. *Behavioural and Cognitive Psychotherapy*, 46(1), 115-120.

#### Supporting articles included in the Portfolio of Published Works

In addition to the seven papers constituting the published works, there are 13 supporting works including papers, a newspaper and magazine article, and two short books that provide additional information on anxiety, eTherapy, and telephone-delivered therapy in the treatment of anxiety and depression. To aid with identification, supporting works appear in bold italics throughout the thesis (e.g., *Lidbetter, 2020*) and are listed below:

- (8) Lidbetter, N., (2003). The stigmatisation of anxiety disorders. In A. Crisp (Ed.). *Every Family in the Land* (pp. 52-53). London: Royal Society of Medicine Press Ltd.
- (9) Bee, P. E., Lovell, K., Lidbetter, N., Easton, K., & Gask, L. (2010). You can't get anything perfect: "User perspectives on the delivery of cognitive behavioural therapy by telephone." *Social Science and Medicine*, 71(7), 1308-1315.
- (10) Lidbetter, N., & O'Neill, C. (2010). Anxiety: falling between primary and secondary care. *Mental Health Today (Brighton, England)*, 17
- (11) Baldwin, D. S., Anderson, I. M., Nutt, D. J., Allgulander, C., Bandelow, B., Den Boer, J. A., Christmas, D., Davies, S., Fineberg, N., Lidbetter, N., Malizia, A., McCrone, P., Nabarro, P., O'Neill, C., Scott, J., van der Wee, N. & Wittchen, H-U. (2014). Evidence-based pharmacological treatment of anxiety disorders, post-traumatic stress disorder and obsessive-compulsive disorder: A revision of the 2005 guidelines from the British Association for Psychopharmacology. *Journal of Psychopharmacology*, 28(5), 403-439.
- (12) Gellatly, J., Bower, P., McMillan, D., Roberts, C., Byford, S., Bee, P., Gilbody, S., Arundel, C., Hardy, G., Barkham, M., Reynolds, S., Gega, L., Mottram, P., Lidbetter, N., Pedley, R., Peckham, E., Connell, J., Molle, J., O'Leary, N & Lovell, K. (2014). Obsessive Compulsive Treatment Efficacy Trial (OCTET) comparing the clinical and cost effectiveness of self-managed therapies: study protocol for a randomised controlled trial. *Trials*, 15(1), 278.

- (13) Slawson, N. (2016, 24 May). Nicky Lidbetter: 'My anxiety has been a motivator.' *The Guardian*. Retrieved from https://www.theguardian.com/society/2016/may/24/icky-lidbetter-my-anxiety-hasbeen-motivator-mental-health
- (14) Lovell, K., Bower, P., Gellatly, J., Byford, S., Bee, P., McMillan, D., Arundel, C., Gilbody, S., Gega, L., Hardy, G., Reynolds, S., Barkham, M., Mottram, P., Lidbetter, N., Pedley, R., Molle, J., Peckham, E., Knopp-Hoffer, J., Price, O., Connell, J., Heslin, M., Foley, C., Plummer, F. & Roberts, C. (2017a). Clinical effectiveness, cost-effectiveness and acceptability of low-intensity interventions in the management of obsessive-compulsive disorder: the Obsessive-Compulsive Treatment Efficacy randomised controlled Trial (OCTET). *Health Technology Assessment*, 21(37), 1-132.
- (15) Lovell, K., Bower, P., Gellatly, J., Byford, S., Bee, P., McMillan, D., Arundel, C., Gilbody, S., Gega, L., Hardy, G., Reynolds, S., Barkham, M., Mottram, P., Lidbetter, N., Pedley, R., Molle, J., Peckham, E., Knopp-Hoffer, J., Price, O., Connell, J., Heslin, M., Foley, C., Plummer, F & Roberts, C. (2017b). Low-intensity cognitive-behaviour therapy interventions for obsessive-compulsive disorder compared to waiting list for therapist-led cognitive-behaviour therapy: 3-arm randomised controlled trial of clinical effectiveness. *PLoS Medicine*, 14(6), E1002337.
- (16) Kenning, C., Blakemore, A., Bower, P., Safari, M., Cuijpers, P., Brown, J. S., Hann, M., Lidbetter, N., Muñoz, R, F. & Waheed, W. (2019). Preventing depression in the community by voluntary sector providers (PERSUADE): intervention development and protocol for a parallel randomised controlled feasibility trial. *BMJ Open*, 9(10), E023791.
- (17) Lidbetter, N. (2020a). Understanding Anxiety: An Anxiety UK self help guide. Manchester: Anxiety UK. Retrieved from https://www.anxietyuk.org.uk/products/anxiety-condition/anxiety/newunderstanding-anxiety-booklet-instant-download/

- (18) Lidbetter, N. (2020b). The Caregivers Guide to Anxiety. Manchester: Anxiety UK. Retrieved from https://www.anxietyuk.org.uk/products/anxiety-condition/anxiety/thecaregivers-guide-to-anxiety-instant-download/
- (19) McMillan, D., Bee, P., Lidbetter, N., Lukoseviciute, B. (2020). Telephone therapy is convenient and it works. Let's use it more. Retrieved from https://psyche.co/ideas/telephone-therapy-is-convenient-and-it-works-lets-use-it-more
- (20) Rushton, K., Ardern, K., Hopkin, E., Welsh, C., Gellatly, J., Faija, C., Armitage, C., Lidbetter, N., Lovell, K., Bower, P. & Bee, P. (2020). 'I didn't know what to expect': Exploring patient perspectives to identify targets for change to improve telephonedelivered psychological interventions. *BMC Psychiatry*, 20, 1-13.

#### Introduction

The aim of this thesis is to describe the development and evaluation of a non-clinical, fully peer-supported eTherapy model, referred to hereinafter as the '*model*', in the management of anxiety and depression in adults in the community.

#### The key objectives were to:

- 1 Examine the development of the *model*
- 2 Review and evaluate the clinical effectiveness of the model
- 3 Evaluate the acceptability of the *model*

The thesis is split into 6 sections:

- Section one sets the context for the thesis by providing an overview of the definition, prevalence, and treatment in England available through the National Health Service (NHS) for anxiety and depression, including the role of third sector providers in the delivery of IAPT and eTherapy services.
- Section two provides an overview of eTherapy in the treatment of anxiety and depression, the evidence base and clinical effectiveness, client acceptability and service delivery models, including peer support in eTherapy.
- Section three provides the background and rationale for the published works, thesis aim and objectives, contribution to knowledge made by the included articles and an overview of the non-clinical, peer-supported eTherapy *model*.

- Section four provides an overview and critique of the publications individually and by thesis objective.
- Section five examines ethical considerations, methodology and limitations, the pragmatic paradigm and personally reflects on the thesis.
- Section six summarises key outcomes, makes recommendations for future research and service development, and provides an overall conclusion to the thesis.

#### Section One: Definition and overview of anxiety and depression

In this section, an overview of anxiety and depression (including common terminology) is provided, along with prevalence rates, treatment options available in England, and the role of the Third Sector (TS) and Third Sector Organisations (TSOs) in the provision of services. This section demonstrates the versatility of, place within current mental health services for, and need for, eTherapy, given the substantial number of individuals affected by anxiety and depression.

#### 1.1 Common mental disorders and terminology

Common mental disorders (CMDs) are depressive and anxiety disorders (McManus, 2016) including Mixed Anxiety Depressive Disorder (MADD), where anxiety and depression are experienced in limited and equal intensity (Kara, 2000).

From hereinafter, the terms 'anxiety' and 'depression' are used instead of CMD. Anxiety means all forms of 'anxiety disorder' including, but not limited to, Generalised Anxiety Disorder (GAD), perinatal anxiety, panic disorder, phobias, Post-Traumatic Stress Disorder (PTSD), and Obsessive-Compulsive Disorder (OCD). Depression means all forms of depressive disorder including Major Depressive Disorder (MDD) and dysthymia. Bipolar and psychotic depression are excluded from the thesis' scope as neither are treated in primary care IAPT services.

#### 1.2 Anxiety

Anxiety is characterised by anticipation about perceived future threats (Craske & Stein, 2016), comprising behavioural, motivational, somatic, affective, and cognitive complexities

(Price, 2003), often leading to persistent impairment in functioning and stigmatisation (Curcio & Corboy, 2020; *Lidbetter, 2003*). Adults seeking treatment usually have anxiety that interferes with their ability to cope with life events and challenges (Steimer, 2002). Though clinical levels of anxiety can occur at any age, most anxiety disorders begin during childhood, adolescence, and early adulthood (Klein & Pine, 2001).

Craske and Stein (2016), see Table 1, give key signs and symptoms of anxiety. The World Health Organization (WHO) (2017a) define anxiety as a chronic condition comprising psychological, behavioural, and physical symptomatology (*Lidbetter, 2020a*) impacting on family and friends, as well as the individual that is directly affected (*Lidbetter, 2020b*).

There are numerous anxiety disorders (*Lidbetter, 2020a*) with similar behavioural manifestations; however, enquiry about cognitions and associated beliefs typically facilitates a definitive diagnosis (Craske & Stein, 2016).

Diagnosis of anxiety in England is made by General Practitioners (GPs) via the Diagnostic and Statistical Manual of Mental Disorders 5<sup>®</sup> (DSM -5<sup>®</sup>), American Psychiatric Association [APA] (2013), and the International Classification of Diseases 11<sup>th</sup> Revision [ICD-11] (WHO, 2018a). Symptom severity is typically measured via the GAD-7 clinical outcome measure (Spitzer, Kroenke, Williams & Lowe, 2006). Whilst diagnosis may help clients gain understanding of their experience, an unintended consequence of diagnostic classification systems is that they do not always reflect lived experience (Hackman et al., 2019).

The IAPT approach itself can be viewed as a medical model of psychological therapy (Binnie, 2015). This reductionist and disablist approach is often viewed as problematic since it views clients as a complicated mixture of anatomical components and physiological systems (Aggleton & Chalmers, 2000); emphasising biochemical dysfunction as the reason behind poor health, focussing on the presence of illness (Jeffery, 2006) and attributing it to a

sole cause within the body (Wade & Halligan, 2017). Those with lived experience of mental health conditions can feel disempowered by the medical model's emphasis on healing from mental health conditions being firmly in the hands of medics, contributing to perceived power imbalances (Gutkin, 2012) as opposed to recovery being something that clients can own and lead on. Additionally, the medical model can result in the medicalisation of human experience and in stigmatisation, discrimination and labelling of those with mental health difficulties (Beresford, Nettle & Perring, 2010). This contrasts with the experiences of many who despite living with anxiety and depression, demonstrate that it is possible to lead a full and meaningful life (Lidbetter & Bunnell, 2013).

Though the British Psychological Society's Division of Clinical Psychology (DCP) gave a position statement on the need for a paradigm shift, moving away from diagnosis being based on a disease model to a relational conceptualisation where the role of relationships in shaping experience, behaviour and healing from distress is recognised (Johnstone et al., 2018), medical models of psychological therapy practice still persist (BABCP, 2020). The IAPT paradigm and epistemological framework has been adopted for the purposes of this thesis, as it remains the key initiative through which psychological therapies in England are delivered. Furthermore, whilst the approach has its limitations as detailed above, it was appropriate and relevant to use given the need to demonstrate the meeting of commissioner targets which were quantitative in nature.

Table 1: Key signs an	d symptoms of a	nxiety disorders	(Craske &	Stein, 2016).
-----------------------	-----------------	------------------	-----------	---------------

Anxiety disorder	Key signs and symptoms as per Diagnostic and
type	Statistical Manual of Mental Disorders, 5 - DSM-5 $^{ m  extsf{B}}$
	(APA, 2013) and International Classification of
	Diseases [ICD-10] (WHO, 2017b)

Separation anxiety	•	Marked fear or anxiety about separation from	
disorder		attachment figures to a degree that is developmentally	
		inappropriate	
	•	Persistent fear or anxiety about harm coming to	
		attachment figures and events that could lead to loss of,	
		or separation from, them	
	•	Reluctance to leave attachment figures	
	•	Nightmares and physical symptoms of distress	
	•	The symptoms usually develop in childhood, but can	
		develop throughout adulthood as well DSM-5® (APA,	
		2013)	
	•	A four-week duration is required for diagnosis in	
		childhood, whereas a longer duration, typically of at	
		least six months, is required in adulthood DSM- $5^{\text{®}}$	
		(APA, 2013)	
Selective or	•	Consistent failure to speak in specific social situations	
elective mutism		(e.g., school) where an expectation to speak exists,	
		despite speaking in other situations	
	•	Not limited to interactions with adults	
	•	Not explained by absence of familiarity with the	
		spoken language	
	•	Persists for at least one month (e.g., beyond the first	
		month of school)	
Specific phobia		Marked fear anxiety or avoidance of circumscribed	
specific phosia	•	objects or situations: $DSM_5^{\text{@}}$ (APA 2013) states that	
		the fears should be out of proportion to the danger	
		posed: ICD-10 (WHO, 2017b) specifies recognition	
		that the symptoms are excessive or unreasonable	
	•	Types of specific phobias include animals (e.g.	
		spiders insects dogs) the natural environment or	
		natural forces (e.g., heights, storms, water), blood	
		injection injury (e.g., needles, invasive medical	
		injection injerg (e.g., needles, invusive medical	

		procedures), situational (e.g., aeroplanes, lifts, enclosed		
		places), and other (e.g., situations that could lead to		
		choking or vomiting; in children, loud sounds, or		
		costumed characters)		
Social anxiety	•	Marked fear, anxiety, or avoidance of social		
disorder (social		interactions and situations that involve being		
phobia)		scrutinised or being the focus of attention, such as		
		being observed while speaking, eating, or performing		
		in front of others: DSM-5 <sup>®</sup> (APA, 2013) specifies that		
		the fear or anxiety should be out of proportion to the		
		threat posed; ICD-10 (WHO, 2017b) specifies		
		recognition that the symptoms are excessive or		
		unreasonable		
	•	Fear negative judgment from others and, in particular,		
		being embarrassed, humiliated, or rejected, or		
		offending others		
	•	ICD-10 (WHO, 2017b) specifies physical symptoms		
		and symptoms of blushing, fear of vomiting or		
		urgency, or fear of micturition or defecation		
	•	A subset have social anxiety in performance situations		
		only (e.g., performing in front of an audience)		
Panic disorder	•	Recurrent, unexpected (i.e., without an apparent cue)		
		panic attacks*		
	•	DSM-5 <sup>®</sup> (APA, 2013) specifies persistent concern or		
		worry about having more panic attacks or changed		
		behaviour in maladaptive ways (e.g., avoidance of		
		exercise or unfamiliar locations)		
	•	Persistent, for at least one month		
A 1.1.				
Agoraphobia	•	Marked fear, anxiety, or avoidance of two or more of		
		the following situations: DSM-5 <sup>w</sup> (APA, 2103): public		
		transportation (e.g., travelling in automobiles, buses,		
		trains, ships, aeroplanes), open spaces (e.g., carparks,		
		marketplaces, bridges), enclosed places (e.g., shops,		

		theatres, cinemas), queues or crowds, or being outside
		of home alone; for ICD-10 (WHO, 2017b), the
		situations are crowds, public places, travelling alone,
		and travelling away from home: DSM-5® (APA, 2013)
		specifies that the fear or anxiety should be out of
		proportion to the threat posed; ICD-10 (WHO, 2017b)
		specifies recognition that the symptoms are excessive
		or unreasonable
	•	DSM-5 <sup>®</sup> (APA, 2013) specifies fear that escape might
		be difficult, or help might not be available in the event
		of panic-like or other incapacitating or embarrassing
		symptoms (e.g., incontinence); ICD-10 (WHO, 2017b)
		lists panic symptoms only
Generalised anxiety	•	Marked anxiety and worry, more days than not, about
disorder		various domains, such as work and school
		performance, which the individual finds difficult to
		control, for at least six months
	•	At least three DSM-5 <sup>®</sup> (APA, 2013) or four ICD-10
		(WHO, 2017b) physical symptoms: restlessness or
		feeling keyed up or on edge, easily fatigued, difficulty
		concentrating, irritability, muscle tension, sleep
		disturbance (i.e., difficulty falling or staying asleep or
		unsatisfying sleep), and symptoms of autonomic
		arousal - ICD-10 (WHO, 2017b)
Anxiety disorders	•	Marked fear or anxiety that is the direct physiological
associated with		consequence of another medical disorder
another medical		
condition		
Substance-induced	•	Marked fear or anxiety due to substance intoxication or
or medication-		withdrawal, or due to drug treatment
induced anxiety		
disorder		

Illness anxiety	•	Preoccupation with having or acquiring a serious,
disorder		undiagnosed medical illness
$(hypochondriasis)^{\dagger}$	•	For DSM-5 <sup>®</sup> (APA, 2013) the somatic symptoms are
		either not present or only mild in intensity (if they are
		present, the diagnosis of somatic symptom
		disorder could be applied)
	•	For ICD-10 (WHO, 2017b), preoccupation might be
		with presumed deformity or disfigurement (body
		dysmorphic disorder (BDD) in DSM-5® (APA, 2013)

#### 1.3 Depression

Depression is associated with psychological, physical, and behavioural symptoms including fatigue, reduced cognitive functioning, appetite changes, suicidal ideation, feelings of guilt and/or worthlessness, sadness, loss of interest in life and daily activities, and sleep disturbances (APA, 2013).

Depressive disorders are split into two categories (WHO, 2017a):

(1) Major Depressive Disorder (MDD) – includes any number and severity of symptoms detailed above, often categorised by severity and associated with poor physical health outcomes (Knol et al., 2006; Luppino et al., 2010). Negative life events are often followed by depression (Brown & Harris, 1978; Hammen, 2005), suggesting their role in the development of depression (Joormann & Stanton, 2016).

(2) Dysthymia – mild depression which may be chronic or persistent. Symptoms are like those of a depressive episode but of a lesser intensity and lasting longer.

Depression is likely to re-occur following an episode of depression (Solomon, 2000), and is the leading cause of global disability (WHO, 2017a) and a known risk factor for suicide (Gagnon & Oliffe, 2015). It is diagnosed, like anxiety, via the DSM-5<sup>®</sup> (APA, 2013) or the International Classification of Diseases, 10<sup>th</sup> edition (WHO, 1993) with severity of depressive symptoms typically measured using the Patient Health Questionnaire (PHQ-9) clinical outcome measure (Kroenke, Spitzer & Williams, 2001). To obtain a diagnosis of depression, five or more symptoms, including lowered mood, must be experienced (Tolentino & Schmidt, 2018) – see Table 2.

Depression is often considered a heterogeneous diagnosis (National Institute for Clinical Excellence - NICE, 2009) because of frequent co-morbidity with anxiety (Tyrer, 2001) and sedentary behaviour (Zhai, 2015a), over or under sleeping (Zhai, 2015b), and long-term stress (Slavich, 2014) all increasing its risk.

#### Table 2: Diagnostic criteria for depression (APA, 2013).

Five (or more) of the following symptoms should be present nearly every day during the same two-week period, and one of the symptoms must be either a depressed mood or loss of pleasure/interest in daily activities. Symptoms must also reach clinically significant levels resulting in an adverse impact on an individual's ability to function in work and/or social life.

- Depressed mood
- Markedly diminished interest or pleasure in (almost) all activities
- Significant weight loss or gain; decrease or increase in appetite
- Insomnia or hypersomnia
- Psychomotor agitation or retardation
- Fatigue or loss of energy
- Feelings of worthlessness or excessive or inappropriate guilt
- Diminished ability to think or concentrate, or indecisiveness

• Recurrent thoughts of death, suicidal ideation, or a suicide attempt or a specific plan for completing suicide

Diagnosis of depression is challenging (Tyrer, 2009) because of chronicity and severity permeations (Klein, 2008), heterogeneity (Rush, 2007), cultural differences in experience (Haroz et al., 2017), and reliance on clinical judgement (Liu & Jiang, 2016).

ICD-10 (WHO, 2017) and DSM-5<sup>®</sup> (APA, 2013) categorise depression on the occurrence of a single episode of depression, a major depressive episode – MDE. Recurrent depression is common (Boland & Keller, 2008); defined by DMS-5<sup>®</sup> (APA, 2013) and ICD-10 (WHO, 2017b) as two or more episodes with at least two months in between each episode, during which the criteria for depression is not met. MDD in DSM-5<sup>®</sup> (APA, 2013) equates to an individual having had one or more MDE. When anxiety is present, depression is less likely to be detected within primary care (Thompson et al., 2000). Subthreshold depression - when symptoms are insufficient to meet diagnostic criteria - is often observed at community level and is a risk factor for the development of more severe depression (*Kenning et al., 2019*).

Late-life depression (adults aged 50-70 years) compared with early-life depression (adults aged 18-49 years) is similar in terms of phenomenology, though older adults are likely to have more somatic symptoms (Hegeman et al., 2012).

#### **1.4** Prevalence of anxiety and depression

Anxiety disorders are common (Kessler, Ruscio, Shear & Wittchen, 2009) with a global prevalence of 7.3% (Baxter et al., 2014), and highest prevalence in Euro-Anglo cultures (Baxter, Scott, Vos & Whiteford, 2013); possibly explained by cultural and subsequent

differences in expression (Hofman & Hinton, 2014). Those aged under 35 are more likely to experience anxiety, irrespective of country of origin and culture (Baxter et al., 2013; Somers et al., 2006; Steel et al., 2014), except in Pakistan, where anxiety affects those of middle age more (Mirza & Jenkins, 2004). In the UK, 8.2 million cases of anxiety were reported (Fineberg et al., 2013). Women are twice as likely to be diagnosed with anxiety (Martín-Merino, Ruigómez, Wallander, Johansson, & García-Rodríguez, 2009) and epidemiological surveys suggest a lifetime prevalence of 16.6-21.3% (Kessler et al., 2005; Kessler & Bromet, 2013).

Anxiety in the community is more common than depression (Buszewicz & Chew-Graham, 2011); however, numbers of correct anxiety diagnoses in primary care are low (Fernández et al., 2012; Vermani, Marcus & Katzman, 2011) where there is a high prevalence ranging from 8-20% (Kroenke, Spitzer, Williams, Monahan, & Löwe, 2007; Mergl et al., 2007; Qin et al., 2010; Serrano-Blanco et al., 2010).

Depression has a global prevalence of ~4.7% (Ferrari et al., 2013), with 5% of adults in developed countries meeting diagnostic criteria for MDD (Wells & Fisher, 2016).

In the UK, 22.5% of females aged 16 years and over in 2014 showed symptoms of anxiety or depression, compared to 16.8% of males (Evans, 2016), and women are twice as likely to experience depression than men (Kuehner, 2017). This is reflected in services where men are much less likely to access psychological therapy support (Men's Health Forum, 2020).

Both anxiety symptoms and anxiety conditions (Melartin et al., 2002) are prevalent in those experiencing depression, with as many as 80% having symptoms of anxiety and over half having a diagnosable anxiety disorder; demonstrating the high comorbidity in existence between the two conditions (Grobler, 2013).

30

Finally, anxiety and depression have been linked to reduced quality of life in those with and without additional health problems (Cerne et al., 2013; Comer et al., 2011; Mols, Schoormans, de Hingh, Oerlemans & Husson, 2018; Stevanovic, Jancic & Lakic, 2011), with depression presentations predicted to be the leading cause of disability in countries of high income by 2030 (Mathers, 2005). It is therefore critically important that those affected by anxiety and depression receive treatment in a timely manner, and that treatment options are increased.

#### 1.5 Treatment of anxiety and depression in England

Anxiety and depression are typically diagnosed and treated in primary care (Toft et al., 2005), with most seeking help from their GP (Cape & Parham, 1998), though detection of anxiety is often insufficient (Buszewicz & Chew-Graham, 2011). Only 33% with depression or anxiety receive help in England (McManus et al., 2016), with many not seeking help (Thornicroft, 2007; Wang et al., 2007). Those requiring support over and above what a GP can offer are given low intensity treatments initially (Bennett-Levy et al., 2010), however many with anxiety continue to fall between primary and secondary care, having presentations that are considered too severe for primary care yet not severe enough to warrant access to secondary care services *(Lidbetter & O'Neill, 2010)*.

NICE guidelines for anxiety and depression (NICE, 2004a, 2004b, 2006, 2009; National Collaborating Centre for Mental Health, NICE & British Psychological Society, 2011) recommend psychological therapies, which, in England, are delivered with other evidence-based treatments, such as pharmacological interventions (*Baldwin et al., 2014*), via the stepped care model (Scogin, Hanson & Welsh, 2003; Simon et al., 2001) through IAPT

services - Table 3 (NICE, 2011) - with computerised CBT / eTherapy recommended at Step

2.

# Table 3: Stepped-care model: a combined summary for common mental health disorders (NICE, 2011).

Focus of the intervention	Nature of the intervention
Step 3: Persistent subthreshold depressive symptoms or mild to moderate depression that has not responded to a low-intensity intervention; initial presentation of moderate or severe depression; GAD with marked functional impairment or that has not responded to a low-intensity intervention; moderate to severe panic disorder; OCD with moderate or severe functional impairment; PTSD.	<ul> <li>Depression: CBT, IPT, behavioural activation, behavioural couples' therapy, counselling*, short-term psychodynamic psychotherapy*, antidepressants, combined interventions, collaborative care**, self-help groups.</li> <li>GAD: CBT, applied relaxation, drug treatment, combined interventions, self-help groups.</li> <li>Panic disorder: CBT, antidepressants, self-help groups.</li> <li>OCD: CBT (including ERP), antidepressants, combined interventions and case management, self-help groups.</li> <li>PTSD: Trauma-focused CBT, EMDR, drug treatment.</li> <li>All disorders: Support groups, befriending, rehabilitation programmes, educational and employment support services; referral for further assessment and interventions.</li> </ul>
<b>Step 2:</b> Persistent subthreshold depressive symptoms or mild to moderate depression; GAD; mild to moderate panic disorder; mild to	<b>Depression:</b> Individual facilitated self-help, computerised CBT, structured physical activity, group-based peer support (self-help) programmes**, non-directive counselling delivered at home†, antidepressants, self-help groups.

moderate OCD; PTSD (including people with mild to moderate PTSD).	<ul> <li>GAD and panic disorder: Individual non-facilitated and facilitated self-help, psychoeducational groups, self-help groups.</li> <li>OCD: Individual or group CBT (including ERP), self-help groups.</li> <li>PTSD: Trauma-focused CBT or EMDR.</li> <li>All disorders: Support groups, educational and employment support services; referral for further assessment and interventions.</li> </ul>		
<b>Step 1:</b> All disorders – known and suspected presentations of common mental health disorders.	<b>All disorders:</b> Identification, assessment, psychoeducation, active monitoring; referral for further assessment and interventions.		
* Discuss with the person the uncertainty of the effectiveness of counselling and			

psychodynamic psychotherapy in treating depression.

\*\* For people with depression and a chronic physical health problem.

† For women during pregnancy or the postnatal period.

CBT, cognitive behavioural therapy; ERP, exposure and response prevention; EMDR, eye movement desensitisation and reprocessing; GAD, generalised anxiety disorder; OCD, obsessive compulsive disorder; IPT, interpersonal therapy; PTSD, post-traumatic stress disorder.

IAPT services deliver evidence-based interventions (Kendall, Pilling, Glover & Taylor, 2011), specifically, CBT for mild to moderate depression and some forms of anxiety (Clark, 2011), with stepped-care being closely aligned to waiting-list management and cost-effectiveness values (Pickersgill, 2019). Treatment is matched to client need with the least intrusive and effective intervention offered at the earliest point in time, and with each step

having clear eligibility criteria. High intensity treatments are reserved for those with greater symptom severity (Bower & Gilbody, 2005), with 70% (Clark, 2018) treated at low intensity level and typical eligibility criteria for clients being as follows:

- First presentation of the problem
- Presentation is mild/moderate depression/anger/self-esteem/anxiety (including first presentation social anxiety), mild OCD and stress
- Problem is not complex with onset < two years prior to presentation.
- IAPT caseness is met (NHS Digital, 2017) clients score 10 or more on the PHQ-9 and/or eight or more on GAD-7
- Client scores below 15 on PHQ-9 and GAD-7

Assumptions made regarding numbers of clients benefitting from low intensity interventions (NICE, 2004b) have been questioned (Lovell et al., 2008), though they are considered cheaper to provide, involve less clinician and service input (Van Straten, Hill, Richards & Cuijpers, 2015) with assessments delivered by Psychological Wellbeing Practitioners (PWPs). Some believe that those who do not respond to low intensity interventions may be negatively affected in respect of their attitude to further treatment options (Kellet & Matthews, 2008), and that treatment choice may be perceived to be limited (Lovell & Bee, 2008). Therapy modality, delivery method and service accessibility are important, yet IAPT clients often comment on its inflexibility (Marshall et al., 2016).

Though IAPT focuses on CBT - empirically supported for anxiety and depression (NICE, 2004a; 2004b; 2009) and accepted as a first-line treatment for depression (National Collaborating Centre for Mental Health, 2010) - there is a need for evidence-based psychological therapies to be expanded (Thornicroft, 2018). IAPT has been questioned as to whether it does fully improve access to all psychological therapies as it is more synonymous

with CBT (Binnie, 2015). As such, an illusion has been created regarding the widening of access to all psychological therapies (Mason & Reeves, 2018).

CBT is not for everyone; clients can struggle with the 'homework' element (Omylinska-Thurston, McMeekin, Walton & Proctor, 2019), whilst others find it hard to identify and challenge negative thoughts and feelings (Nilsson, Svensson, Sandell & Clinton, 2007). This is problematic as cognitive restructuring is a key component of the approach (Mansell, 2008). When therapists stick rigidly to a CBT protocol, this can lead to clients feeling misunderstood and invalidated (Bystedt, Rozental, Andersson, Boettcher & Carlbring, 2014); eTherapy, however, can facilitate the delivery of CBT without contamination from such variables (Knowles et al., 2014).

It is often difficult to identify the most appropriate treatment for depression (Cuijpers, 2018) and anxiety; this having been an issue for some time (Paul, 1967). Interpersonal therapy (Churchill, 2010; Cuijpers, 2016), Behavioural Activation (BA) (Ekers, 2008; Shinohara, 2013), problem-solving therapy, (Malouff, 2007), third-wave psychotherapies (Churchill, 2010), and psychodynamic therapy (Leichsenring, 2008) are all effective in the treatment of adult depression (Brettle, 2012; Cuijpers, 2018), with minor differences between modalities in terms of effectiveness for all types of depression (Brettle, 2012). Furthermore, their effects are comparable (Barth, 2013; Palpacuer, 2017). BA, like CBT, has a meta-analytic level of support for its evidence (Butler, 2006; Cuijpers, 2007). Recently, pluralistic approaches encompassing numerous modalities have been said to be essential (British Association for Counselling and Psychotherapy [BACP], 2020), raising questions about the past decade's focus on CBT. There is an also an argument for a broader approach to the management of anxiety and depression to be taken with issues such as nutrition, diet (Anxiety UK, 2020; Terry & Reeves, 2015), and sleep (Cox, Bunmi & Olatunji, 2020) considered in treatment solutions.

The therapeutic relationship, however, is key for outcomes and engagement (Ardito, 2011; Horvath, Del Re, Flückiger & Symonds, 2011), and considered essential for effective therapy (BACP, 2020). Relationship factors, such as empathy and warmth, contribute to outcomes even when therapist contact is limited (as in eTherapy), with good relationship skills generating positive outcomes irrespective of modality (Norcross & Lambert, 2018). Further research on therapist and service variability is required, instead of focusing on modalities (Pybis et al., 2017). Taken together, these variables have the potential to increase patient choice and address the high drop-out rates seen within low intensity IAPT services (Chan & Adams, 2014).

#### 1.6 The role of the Third Sector in the delivery of IAPT services

The TS is known for plugging gaps in statutory services, responding in innovative ways to national policy initiatives, and stepping in when there have been service failures (Newbigging, Mohan, Rees, Harlock & Davis, 2017). TSOs run independently of the state, have social aims, and can access hard to reach populations; successfully engaging communities (Allen, 2011). Government policy has supported provider plurality, however local commissioning strategy determines the extent of TSO commissioning (Allen, 2012) with varied views held by commissioners on this (Baird, 2018.) To secure NHS contracts, TSOs are required to meet specific criteria, particularly regarding IAPT contracts (Sweet, 2019); many report being torn between meeting their original aim versus meeting the demands of delivering contracts (Department of Health, 2016). Some have had to scale up and 'professionalise' when delivering public service contracts (Department of Health, 2016); being forced to act in a more business-like manner (Third Sector, 2020). This has changed organisational dynamics, values, and service delivery (Laurie & Bondi, 2006), leading to notions of constraint, pressure, and uniformity (Boyles & McKinnon Fathi, 2019).

36
NHS England state that the TS: 'often has an impact well beyond what statutory services alone can achieve (NHS England, 2014, p.14). Furthermore, there is acknowledgment that TSOs typically 'offer a rich range of activities in response to mental health, under the one roof' (NHS England, 2014, p. 14). The sector is renowned for being able to reach under-represented groups, and for its increasing diversity in terms of the make-up of organisations in respect of their size and constitution (Carey & Braunack-Mayer, 2009). Mental health services are no longer solely delivered by the NHS; TSOs now provide such services (Bennion, Hardy, Moore & Millings, 2017). Whilst data from TSOs is included in the IAPT dataset, there is no flag or identifier to distinguish this data from that of other service providers (NHS Digital, 2020). Consequently, there are no reports that state the ratio that TSOs comprise in the dataset regarding their overall contribution to, IAPT.

Self Help Services is a user-led charity established in 1995 by the thesis' author. The charity provides a wide range of inclusive services (Lidbetter & Bunnell, 2013) and routinely participates in community research projects such as the PERSUADE trial (*Kenning et al., 2019*). It has been commissioned to deliver low intensity IAPT services from the mid-2000s, including eTherapy services (Cavanagh, Seccombe, Lidbetter & Bunnell, 2011a; Cavanagh, Seccombe & Lidbetter, 2011b). Its user-led ethos ensures that peer support is at the heart of its service delivery, whilst innovating and responding to client need for accessible, peer-supported services. In doing this, the charity aligns itself to key policy drivers around clients being in control of their care, that services are focused on client needs and preferences; therefore, facilitating choice (Ormandy & Hulme, 2013).

## 1.7 Section discussion

IAPT has been the dominant treatment driver for anxiety and depression in England over the past decade. Whilst access to CBT has increased (Moller, Ryan, Rollings & Barkham, 2019),

fewer than one in five individuals in England experiencing anxiety and/or depression are being reached (Clark, 2018). This figure being against a target of at least 1.5 million people by 2020/1 (NHS England, 2019) and rising levels of anxiety and depression in the UK (Sample, 2020). Whilst guidelines recommend treatment is based on the stepped care model, there is limited data available as to how this is implemented in UK routine practice (Richards et al., 2012), and specifically to understand how eTherapy can be successfully implemented in services as an intervention (Drozd, Vaskinn, Bergsund & Haga, 2016; Folker et al., 2018) at low intensity level. New delivery methods and blended approaches are needed to increase access (Brown, 2018), including exploring the TS and its role in delivering services to address population need at primary care level (Chew & Osborne, 2009).

# **1.8 Section summary**

This section provided an overview of anxiety and depression, a summary of treatment available via IAPT - the vehicle through which the majority of the country's primary mental health care service offer is delivered - and an examination of the role of TS providers in the provision of NHS services. The next section examines eTherapy, which is an intervention delivered largely at low intensity level, and the role of peer support in eTherapy service delivery models.

## Section Two: eTherapy

This section provides an overview of eTherapy terms, the history and policy context for its role in the treatment of anxiety and depression, its clinical effectiveness, its client acceptability, and an examination of service delivery models, including the role of peer support.

# 2.1 Definition of eTherapy

The NHS defines digitally enabled therapy as "psychological therapy that is provided online or through mobile applications, with the support of a therapist" (NHS England, 2020a, para 2). In eTherapy, users are guided through content (usually in module format) via text, images, video resources, and sometimes case studies, either interactively or in a noninteractive manner (Rodriguez-Pulido, Castillo & Hamrioui, 2020). Automated feedback processes are used instead of a therapist being present (Marks, Cavanagh & Gega, 2007).

There is no standardised nomenclature to describe eTherapy and so it is defined and categorised in many ways (Bennion et al., 2017), including by the amount of therapist support (Newman et al., 2011) or the way the internet is used to support delivery (Barak, Klein & Proudfoot, 2009). A classification for digital health interventions has recently been constructed, which should support the development of a common, shared language for eTherapy (WHO, 2018b), in the hope that an industry standard will be developed (Borgueta, Purvis & Newman, 2018).

In this thesis, the term eTherapy covers all associated terms (see Table 4) but not interventions delivered solely through mobile phone applications (apps), though it is recognised that some eTherapy programmes are accessible in both standard and app format. Virtual Reality (VR) therapy or interventions not guided by a computer (e.g., psychological therapy that takes place with a therapist online via videoconferencing facilities or by email or instant messaging or interventions delivered in a group format [e.g., Master Your Mood -MYM (van der Zanden, Kramer, Gerrits & Cuijpers, 2012)]) are excluded.

# **Table 4: eTherapy terms**

iCBT, itherapy, computer-assisted therapy, cybertherapies, cybertherapy, computerdelivered, web-based, electronic CBT (eCBT), computerised Cognitive Behavioural therapy (cCBT), computerised therapy, online CBT, digitally-enabled therapies, computer therapy, digital IAPT, digital psychotherapy, digital cognitive behavioural therapy (dCBT), internetdelivered, internet-administered, internet-guided self-help interventions, computerised psychological therapy, internet-based therapy, internet-based interventions, computer-aided psychotherapy and therapist-guided internet interventions.

#### 2.2 eTherapy in the management of anxiety and depression

An ever-expanding range of computer programs have been developed to support individuals affected by anxiety disorders and depression, typically on the principles of CBT; a structured and directed approach that lends itself to computer-based administration (Anderson, Jacobs & Rothbaum, 2004). However, the assumption in CBT that anxiety and depression stem from faulty attitudes and beliefs has been criticised as being tantamount to blaming clients for their difficulties; failing to take into account that such conditions may instead arise from socio-economic adversity and chronic health difficulties (Knight & Thomas, 2019), and is a limitation of those eTherapy programmes that are entirely CBT-based, of which many are (Simmonds-Buckley et al., 2020). Other modalities have been digitalised including acceptance and commitment therapy - ACT (Pots et al., 2016), BA (Ly et al., 2014), interpersonal psychotherapy - IPT (Donker et al., 2013), mindfulness-based interventions

(Spijkerman, Pots & Bohlmejer, 2016), and problem-solving therapy (Kleiboer et al., 2015). Yet there is gap regarding eTherapy programmes that adopt a pluralistic approach inclusive of all modalities; said to be essential and reflective of clients being unique and needing different things (BACP, 2020).

eTherapy has the potential to offer the same benefits of CBT with less therapist involvement (Kaltenthaler et al., 2002), potentially increasing service capacity (Titov, Andrews & Sachdev, 2010b).

CBT delivered via computer was not used routinely until it was recommended (NICE, 2002), though there was a distinct lack of detail regarding eTherapy implementation and delivery model. Nevertheless, access to CBT was improved within a context of limited therapists, high costs, long waiting times, and clients' reluctance to access therapy. In 2003, the eTherapy programme, Beating the Blues (BtB<sup>TM</sup>) was reported to have positive outcomes in those with anxiety and depression (Proudfoot et al., 2003). However, whilst effective when compared to Treatment as Usual (TAU), in this case, whatever treatment the GP had prescribed, the Randomised Controlled Trial (RCT) was limited by a lack of standard treatment against which to compare BtB<sup>TM</sup> (Proudfoot et al., 2003). Additionally, as the service was delivered from GP surgeries, with a nurse providing instructions as to how to access sessions, there was no focus on the impact of the nurse's involvement. Furthermore, those not randomised to this arm of the trial did not receive this human interaction; thus, the study (Proudfoot et al., 2003) left unanswered questions as to whether nurse contact alone might have influenced results.

In 2004, a novel interactive self-help clinic was established that clients experiencing anxiety and depression accessed from home, with telephone support as required. The service reported saving clinicians' time through delegating routine aspects of therapy to a computer, utilising the programme, FearFighter<sup>™</sup> (Marks, Kenwright, McDonough, Whittaker & Mataix-Cols, 2004; Schneider, Mataix-Cols, Marks & Bachofen, 2005).

A further appraisal by NICE (2006) recommended BtB<sup>™</sup> for mild to moderate depression and FearFighter<sup>™</sup> for phobia, panic, and anxiety; and as a result, both programmes became widely available - typically accessed through GP surgeries on a stand-alone, unsupported basis, akin to the earlier model developed by Marks et al. (2004). A summary of the origins of eTherapy is detailed in Table 5 below:

## Table 5: eTherapy origins (Andersson, 2018a)

- 1. The emergence of evidence-based psychological therapies particularly CBT (Rachman, 2015).
- 2. Guided self-help literature, with trials reporting evidence for the use of such interventions when supported by clinicians (Clum & Watkins, 2008). Earlier forms of eTherapy were likened to this with online or email support (Marks et al., 2007).
- 3. Computerised testing and interventions (Marks, Shaw & Parkin, 1998).

A pilot project developed by the thesis' author in the early 2000s delivered through Self Help Services and the National Phobics' Society (now Anxiety UK), provided access to BtB<sup>™</sup> on an entirely stand-alone, unsupported basis through the 'computerised cognitive behavioural therapy' - cCBT service, for clients experiencing anxiety and depression. Whilst clients found BtB<sup>™</sup> helpful, most needed additional support and guidance, typically delivered by the then eTherapy Co-ordinator (eTC) - an individual with lived experience of anxiety/ depression, akin to a Peer Support Worker (PSW), with whom what amounted to a 'support conversation' would be had at the start and end of sessions. The outcome of this pilot project led directly to the development of a new model of eTherapy service delivery, which addressed this client need for motivational support, guidance, and contact with a human being who could relate to experiences of living with anxiety and depression. This *model*, a nonclinical, fully peer-supported eTherapy model (**Cavanagh et al., 2011a, 2011b; Elison et al., 2014, 2017; Gellatly et al., 2018; Lidbetter & Bunnell, 2013; Luik et al., 2017**), is the main subject of this thesis and is described and explored in greater detail in Section 4 and depicted in Figure 1.

In utilising the lived experience of volunteers and eTCs to provide peer support and thus operating in part as PSWs, the *model* aimed to improve client engagement and reduce dropout rates in comparison to other non-supported models, as well as by increasing accessibility by delivering the service in the community. The *model* evolved through the work that is contained in the portfolio (**Cavanagh et al., 2011a, 2011b; Elison et al., 2014, 2017; Gellatly et al., 2018; Lidbetter & Bunnell, 2013; Luik et al., 2017**) by testing effectiveness and acceptability in a real world setting (**Cavanagh et al., 2011a, b; Elison et al., 2014, 2017; Gellatly et al., 2018; Luik et al., 2017**), further developing the model into a remote *model* (**Gellatly et al., 2018**), and by adding different programmes to meet client need (**Elison et al., 2014, 2017; Luik et al., 2017**).

As IAPT grew, focus was placed on widening access to psychological therapies; eTherapy was positioned as being vital to addressing treatment barriers including costs and waiting times (Richards et al., 2018). However, many were cynical; it was not until the *model* was recognised in Manchester as having potential to assist with meeting IAPT access and recovery targets (see Table 6), that the venue *model* (where the service was delivered from venues such as Information Technology (IT) suites located within community centres) was accepted as a credible low intensity intervention (**Cavanagh et al., 2011a, 2011b**).

In IAPT, recovery is determined by reductions in outcome measures scores (see Table 6). A criticism of the approach is its reliance on the medical model (Binnie, 2015) and failure to

understand symptoms in favour of seeking to suppress or change them (Casement, 2009). Furthermore, there is a lack of focus on the wider domains of recovery (McPherson et al., 2009; Scanlon & Adlam, 2010) which are arguably more meaningful and relevant to those experiencing anxiety and depression than simply a reduction of symptomatology. The reliance on the use of quantitative outcome measures as opposed to holistic, whole-person and specifically, patient-reported outcome measures, as well as those that focus on empowerment, such as the Empowerment Scale (Rogers, Chamberlin, Langer Ellison & Crean, 1997) is also a limitation of the IAPT approach.

# Table 6: Key IAPT recovery and associated definitions (Gellatly et al., 2018).

**IAPT recovery** is the percentage of clients who score below the clinical cut-off of >9 on the PHQ-9 and >7 on the GAD-7 after the treatment period over those who scored above the clinical cut-off on the PHQ-9 or GAD-7 before treatment.

**Reliable improvement** is met if there is a decrease in one or both outcome measures that exceeds the measurement error for that measure (PHQ-9  $\leq$  6; GAD-7  $\leq$  4) and no increase in the other beyond the error of measurement.

**Reliable deterioration** is met if there is an increase in one or both scores (PHQ- $9 \le 6$ ; GAD- $7 \le 4$ ) that is more than the measurement error.

**Reliable recovery** is when a client has both reliably improved and recovered.

As leader of Self Help Services, part of my role involved influencing NHS commissioners to fund the *model*. This was achieved by demonstrating its equivalent access and positive outcomes in comparison to standard IAPT recovery rates (**Cavanagh et al., 2011b**), which led to the commissioning and adoption of the *model* in Greater Manchester and beyond. Whilst the original venue *model* (**Cavanagh et al., 2011a, 2011b**) was found to be effective and acceptable, clients articulated the need for a remote service because of access issues – leading to the @ home/remote *model* being developed a few years later (**Gellatly et al., 2018**).

Many eTherapy programmes exist (see Table 7); differing in disorder specificity, extent of mental health problem addressed (Fairburn & Patel, 2017), and platform design (Hofman, Pollitt, Broeks, Stewart & van Stolk, 2016). Transdiagnostic programmes (TD-cCBT) have also been developed (Newby, Twomey, Li & Andrews, 2016) with greater clinical utility (Fairburn & Patel, 2017). Programmes provide access to therapy content ordinarily delivered face-to-face (Teachman, 2014), with content delivery methods differing (Thew, 2020) and CBT is the most common modality used.

Further developments of the *model* accommodated clients presenting with comorbidity, including substance misuse and sleep difficulties, via the addition of programmes such as Breaking Free Online – BFO (Elison, Humphreys, Ward & Davies, 2013) and Sleepio<sup>™</sup> (Espie et al., 2012).

No comprehensive mapping exercise of eTherapy programmes has been undertaken other than an initial mapping exercise by Van Stolk, Hofman, Hafner, and Janta (2014) followed by RAND Europe's review of CCBT tools (Hofman et al, 2016). The latter was limited in that it did not review tools aimed at children, students, or the elderly, and was limited in scope, only focussing on published studies. The Center for Technology and Behavioral Health has also synthesised research on Digital Health Technologies (DHTs) for substance use disorders and co-occurring conditions including mental health issues, listing key eTherapy programmes on their website (Center for Technology and Behavioral Health, 2020).

eTherapy programme:	Reference	Composition:	Country of origin:	Suitable for/primary application:
Beating the Blues*	Proudfoot et al., 2003	Eight module	UK	Depression and anxiety

Table 7: Key programmes available globally for anxiety and depression.

Breaking Free Online*	Elison, et al., 2013	Psychosocial interventions	UK	Substance misuse
Calm Tools for Living	Roy-Byrne, et al., 2010	Eight sessions	USA	Anxiety and depression
Deprexis	Meyer et al., 2009	10 module programme	Germany	Depression
FearFighter <sup>™</sup> *	Marks et al., 2004; Schneider et al., 2005	Nine module programme	UK	Panic and phobias
GET.ON Stress	Ebert et al., 2016	Seven sessions	Germany	Work- related stress
Good Days Ahead	Wright et al., 2005	Six module programme	America	Stress, anxiety and depression
Internet-Based Self-Help Cognitive Behavioural Therapy (icBT) for Depression	Andersson, Bergstrom, Carlbring, Kaldo, & Ekselius, 2005	10-week programme	Sweden	Mild to moderate depression
Living Life to the Full Interactive) (LLTTF)- now known as 'Living Life to the Full Plus' *	Williams, n.d.	On-line, interactive, self- help skills programme comprising modules on areas of life and wellbeing commonly affected by low mood and stress	UK	Low mood, stress, and anxiety
MoodGYM	Christensen, Griffiths & Jorm, 2004	Five interactive modules	Australia	Depression and anxiety
myCompass	Proudfoot et al., 2013	12 modules	Australia	Depression, anxiety, and stress
PAXPD	Ciuca, Berger, Crișan & Miclea, 2018	12-week treatment containing 16 modules	Romania	Panic disorder and anxiety disorders
SilverCloud Health*	Sharry, Davidson, McLoughlin & Doherty, 2013;	Eight-week, condition-specific programmes, marketed under	Ireland and UK	Adults aged 16 and over. Depression, anxiety, phobia,

	Richards et al., 2015	the name 'space from'		panic, social anxiety, health anxiety, OCD, depression and anxiety, GAD, perinatal wellbeing
Sleepio <sup>TM</sup> *	Espie et al., 2012	Six-week programme with tailored clinical content and up to 12 weeks support via an online community	UK	Insomnia
Smart, Positive, Active, Realistic, X- Factor Thoughts (SPARX)	Fleming, Dixon, Frampton & Merry, 2012	Seven 'provinces' where avatars are used. Modified versions of SPARX have been created such as Rainbow SPARX	New Zealand	Mild to moderate depression in young people aged 12-19 years
The Wellbeing Course	Titov et al., 2011	Transdiagnostic program	Australia	Depression and anxiety disorders
This Way Up - The Panic Program	Wims, Titov & Andrews, 2008 –	Typically, each course comprises six 'lessons' including the 'sadness program', the 'shyness program', and 'the panic program'	Australia	Depression, GAD, mixed anxiety and depression, OCD, social anxiety, post- traumatic stress, health anxiety courses

\*indicates an eTherapy programme used in the model

Programme format varies; some maintain fidelity with the face-to-face psychotherapy standard of weekly sessions; others adopt a format seen with apps and websites (Ben-Zeev et al., 2015). Most deliver a sequence of modules where evidence-based therapies are

delivered, via the internet, through a device (Wright, Mishkind, Eells & Chan, 2019). Just as session delivery frequency varies, so does programme structure. Some are linear with clients working through an intervention in a methodical, systematic way, others are flexible and unstructured where clients select which parts of the programme they wish to access (Fairburn & Patel, 2017). The number of modules available differs according to the target condition, with anxiety disorders being reported to be lengthier (Hofman et al., 2016). Contact time ranges from < 1.5 hours through to more active involvement by the supporter/clinician, though it remains less than that of face-to-face therapy sessions (Cavanagh, Belnap, Rothenberger, Abebe & Rollman, 2018), with the intensity of support typically low (Hofman et al., 2016). Programmes can also be made available to access in a self-guided manner. The amount of supporter time required to achieve maximum benefit for clients accessing eTherapy is an area requiring further research (Andrews et al., 2018). Holländare et al. (2016) investigated common therapist behaviours in therapist-supported eTherapy via email. Most commonly therapists encouraged, affirmed, guided, and urged clients when sending emails, whilst also providing 'clarification,' 'informing' about module content,' 'emphasising' client responsibility, and self-disclosing.

The extent to which programmes are personalised or tailored differs (Twomey, O'Reilly & Bryne, 2014). The SPARX programme has been adapted for the Lesbian, Gay, Bisexual, Transgender and Queer (LGBTQQIAAP) community (Fleming et al., 2012), and another programme for those with intellectual disabilities has been developed (Cooney, Jackman, Coyle & O'Reilly, 2017). There are no programmes available for clients aged over 65 years (Bennion et al., 2017), though some have been tailored to suit client groups including older adults (Silfvernagel et al., 2012).

A digitally enabled therapy assessment programme (NICE, 2020a) assessing eTherapy programmes for use in IAPT services, and producing an IAPT Assessment Briefing (IAB),

was recently launched. Those programmes receiving positive recommendation are evaluated in practice to obtain real-world activity data, however in contrast to the *model* (Cavanagh et al., 2011a, 2011b; Elison et al., 2014, 2017; Gellatly et al., 2018; Lidbetter & Bunnell, 2013; Luik et al., 2017), clinicians provide support, maintaining fidelity with the IAPT manual (National Collaborating Centre for Mental Health, 2018). To date, only the Space from Depression (Sharry et al., 2013) has been evaluated in practice, yet many different programmes are offered in IAPT services (Bennion et al., 2017). It is unclear as to the longer-term impact of the IAPT assessment programme on the future use of programmes within services. It may impact on the model (Cavanagh et al., 2011, 2011b; Elison et al., 2014, 2017; Gellatly et al., 2018; Lidbetter & Bunnell, 2013; Luik et al., 2017), resulting in less choice if only programmes that have been evaluated in practice with an IAB are recommended (NICE, 2019b). Conversely, an evidence standards framework for DHTs developed by NICE (2019a) could lead to greater programme choice if software companies demonstrate compliance and put a case forward for their product to be used with IAPT services, without going through the IAB process. This might result in eTherapy forming a larger part of future IAPT provision in contrast to the small percentage of overall IAPT interactions that it has accounted for (Health and Social Care Information Centre, 2016). This would also contrast with the low uptake seen in some services, which has ultimately resulted in some being terminated (Brown, 2018).

The Topol Review Board, 2019 (Foley & Woollard, 2019) states that eTherapy:

can provide standalone self-help or be blended with traditional mental health interventions or online peer support networks. Such interventions are already in use and have a significant evidence base. In the future, artificial intelligence (AI) and natural language processing (NLP)-enabled chatbots may facilitate more advanced automated or semi-automated therapeutic tools (Foley & Woollard, 2019, p.5).

IAPT providers are not currently required to submit data on either eTherapy programme or service delivery model deployed; such metrics fall outside of the IAPT Minimum Data Set -IAPT MDS (NHS Digital, 2020) - yet the IAPT manual (National Collaborating Centre for Mental Health, 2018, p.44) states that, in relation to digitally-enabled therapy, it could be 'considered as part of the service model design' and:

much of the learning that is required to help people deal with emotional difficulties can be achieved by them working through materials on the internet with ongoing contact with a therapist (by telephone, secure messaging and so on) to provide encouragement, clarify misunderstandings and further enhance learning (IAPT manual, p.40).

# 2.3 eTherapy: clinical effectiveness

Many variables confound eTherapy clinical effectiveness results, including: programme type (transdiagnostic or disorder-specific); target population (community or clinical population); treatment comparisons; whether support is provided versus unguided or automated; who provides the support (clinician/non-clinician); how the service is delivered; the amount of support provided; and method of delivery.

eTherapy research is often focused on individual programmes (Hofman et al., 2016), possibly explained by software companies being keen to establish their programme's efficacy; however, TDcCBT is as effective as disorder-specific programmes (Păsărelu, Andersson, Nordgreen & Dobrean, 2017), and is practical, efficient, and relevant for a range of conditions and multiple comorbidities (Newby et al., 2016).

Meta-analyses have found eTherapy effective for anxiety and depression (Andrews, Cuijpers, Craske, McEvoy & Titov, 2010) and Social Anxiety Disorder - SAD (Kampmann,

Emmelkamp & Morina, 2016) - and systematic reviews report it as being an established treatment for depression, SAD, and panic disorder (Hedman, Ljótsson & Lindefors, 2012).

eTherapy is equally efficacious when compared to face-to-face CBT for SAD, panic disorder, and depression (Andersson et al., 2014; Carlbring, Andersson, Cuijpers, Riper & Hedman-Lagerlöf, 2018). However, an RCT comparing the clinical effectiveness of eTherapy and guided self-help (with support offered by telephone) at low intensity level for OCD, found that neither intervention led to clinically significant benefits (*Gellatly et al., 2014; Lovell et al., 2017a, 2017b*).

Self-guided eTherapy is more effective in treating depression than control conditions (Karyotaki et al., 2017), and eTherapy is clinically effective for anxiety and depression when delivered with minimal clinician guidance (Adelman, Panza, Barley, Bontempo & Bloch, 2014; Andersson & Cuijpers, 2009; Andersson, Cuijpers, Carlbring, Riper & Hedman, 2014; Andrews et al., 2018; Arnberg, Linton, Hultcrantz, Heintz, & Jonsson, 2014; Cuijpers et al., 2009; Olthuis, Watt, Bailey, Hayden & Stewart, 2016; Richards & Richardson, 2012; Titov et al., 2016).

Globally, the International Society for Research on Internet Interventions (ISRII) has debated the role of therapist support (Baumeister, Reichler, Munzinger & Lin, 2014). A number of studies (Campos et al., 2019; Karyotaki et al., 2017) point to the need to investigate supported models of eTherapy as the evidence is mixed. Berger et al. (2011) found that additional support did not lead to better outcomes but observed greater adherence with clients accessing self-guided eTherapy for depression when weekly, short phone calls were provided by an assigned coach (Mohr et al., 2013). Sijbrandij, Kunovski and Cuijpers (2016) found effects were strongest when clients accessing an eTherapy service for PTSD were supported by a therapist; proposing that clinical effectiveness can be improved by increasing the amount of therapist/clinician support, though the effect size is now believed to be smaller than previously reported (Lorenzo-Luaces, Johns & Keefe, 2018). Other studies report that therapist experience makes no difference to outcomes in the treatment of panic and social anxiety disorder (Klein et al., 2009), though experienced therapists may need less time to support clients (Andersson, Carlbring, Fumark & SOFIE Research Group, 2012).

An RCT on the effectiveness of eTherapy found that the provision of telephone support provided short-term benefits in relation to depression symptoms (Brabyn et al., 2016). Moreover, a Cochrane systematic review (Olthuis et al., 2016) assessing the effects of therapist-supported eTherapy for adults with anxiety disorders, found therapist-supported eTherapy more effective than no treatment (in this case, a waiting list). The professional background of the individual providing support is reported to be of minor importance (Baumeister et al., 2014) with no difference in outcomes found in an RCT comparing the efficacy of clinician- versus technician- (non-clinical) supported treatment for depression (Titov et al., 2010).

No difference between outcomes for clients accessing support via a web forum versus telephone calls was found, indicating that the support method may not make a difference (Titov et al., 2009). However, a meta-analysis of multi-modal CBT (CBT across a variety of delivery formats, including guided self-help CBT, telephone-based cCBT, face-to-face CBT and eTherapy CBT), found eTherapy more effective than no treatment and that practitioner support type may account for differences in results (Twomey, O'Reilly & Byrne, 2015).

Whilst eTherapy is effective in the treatment of anxiety and depression, it is inconclusive as to how long outcomes are maintained, and evidence for its long-term benefit is sparce (Hofman et al., 2016). A recent meta-analysis, however, found eTherapy efficacious and acceptable for people reporting symptoms of anxiety and depression, with benefits maintained for up to 18 months (Andrews et al., 2018); another study reported eTherapy as

effective for SAD compared to CBT group therapy, for as long as up to four years post treatment (Hedman et al., 2014a).

There is limited evidence for the effectiveness of eTherapy in Black and Minority Ethnic (BME) groups aside from the **Cavanagh et al., 2011b** study, with few studies focussing on outcomes and ethnicity/racial differences (Jonassaint et al., 2009).

Clinical effectiveness for eTherapy has been shown to be linked to treatment completion (Hobbs, Mahoney & Andrews, 2018; Karyotaki et al., 2017) with adherence greater in older clients (Hobbs et al., 2018; Mewton, Sachdev & Andrews, 2013) when provided through a GP; suggesting that the views of such healthcare practitioners are held in higher regard by older people and/or more likely to be acted on/listened to. Additionally, a meta-analysis and systematic review found mean study sample age significantly moderated the effectiveness of eTherapy in those with CMDs (Grist & Cavanagh, 2013).

# 2.4 eTherapy: acceptability

Acceptability was initially studied over a decade ago (Kaltenthaler et al., 2008), though has not been as extensively researched as other areas of eTherapy (Cavanagh et al., 2009), with a heterogeneity of pilot studies and minimal amount of research on public acceptability of the approach (Apolinário-Hagen, Kemper & Stürmer, 2017). Focus has been on improving access, rather than addressing patient experience (Knowles, 2014). Though, more recently, Rost et al. (2017) reported positive experiences in those accessing eTherapy for depression.

eTherapy should not be positioned as a one-size-fits-all approach (Perera-Delcourt & Sharkey, 2019), as personalisation and sensitisation may improve client experience (Knowles, 2014). For some, eTherapy may not be suitable (Rozental et al., 2014) and the

negative effects of the approach need to be researched (Boettcher, Rozental, Andersson & Carlbring, 2014).

Clients find certain CBT techniques in eTherapy programmes helpful, including psychoeducation and cognitive restructuring, as well as supporter interaction. However, the amount of work clients must complete, as well as frustrations with technical issues, can cause hindrance (Burke, Richards & Timulak, 2018).

Perception of eTherapy is important (Batterham et al., 2019) and can be addressed by giving clients enough information on the intervention prior to accessing the service. Positive positioning in this way gives rise to improved treatment outcomes (Cludius, Schroder & Moritz, 2018) – a phenomenon seen in telephone-delivered therapy interventions (*Bee, Lovell, Lidbetter, Easton & Gask, 2010; Rushton et al., 2020*), and something that is often neglected as a treatment response moderator. Addressing this, and the role that professionals and others play in socialising clients to eTherapy, is critical (Schröder et al., 2018). It has also been suggested that positive attitudes towards eTherapy could be improved with certification of programmes by professional psychological bodies, resulting in increased credibility (Klein et al., 2016), though acceptability is rarely measured directly (Kaltenthaler et al., 2008).

Musiat, Goldstone and Tarrier (2014) reported clients being unenthusiastic about eTherapy, but aware of its advantages in terms of accessibility, advising that policy makers must address public perceptions of the approach. Observability is also something that has been reported to be low in eTherapy (Carper, McHugh & Barlow, 2013), and which too may impact on acceptability and explain reported low intentions from clients for future use (Musiat et al., 2014).

It is recognised that eTherapy could pose a challenge in respect of the therapeutic alliance (Cavanagh et al., 2018) since it involves minimal contact (Newman, Szkodny, Llera & Przeworski, 2011). High alliance ratings have, however, been reported, as is also the case for telephone-delivered therapy (*McMillan, Bee, Lidbetter & Lukoseviciute, 2020*), suggesting an alliance is formed (Andersson et al., 2019b) and thought to be important (Nordgreen, Carlbring, Linna & Andersson, 2013; Pihlaja et al., 2018). Furthermore, those supporting and administering the service are likely to influence alliance (Berger, 2017), as does treatment being provided in a user-friendly manner that is not excessively technical, as this promotes greater adherence (Andersson, Carlbring, Berger, Almlöv & Cuijpers, 2009).

Whilst clients are generally positive about eTherapy (Andrews et al., 2018), with acceptability in some client groups being high, for example, those with OCD (Wootton, Titov, Dear, Spence & Kemp, 2011), others prefer to opt for a blended model (Andersson et al., 2019b) of eTherapy and face-to-face therapy; suggesting that some level of human interaction is beneficial.

Supported eTherapy is generally more acceptable (Andersson, 2018a), and therapist contact is believed to be positively linked to treatment adherence and reduction of client frustration regarding technological issues (Rozental, Boettcher, Andersson, Schmidt & Carlbring, 2015). Lack of therapist contact can lead to clients requesting more support (Donkin & Glozier, 2012; Knowles et al., 2015; Rennick-Egglestone et al., 2016; Rost et al., 2017) and improvement in clients with depression where human support was included (Gellatly et al., 2007) has been reported; consistent with earlier research studies on the effectiveness of digital health interventions (Andersson & Cuijpers, 2009, Christensen, Griffiths & Farrer, 2009; Mohr, Cuijpers & Lehman, 2011).

In the *model* (Cavanagh et al., 2011a, 2011b; Elison et al., 2014, 2017; Gellatly et al., 2018; Lidbetter & Bunnell, 2013; Luik et al., 2017), there is a good degree of human

interaction and contact between clients, eTCs and volunteers, with the service being delivered in the community or remotely, both of which are acceptable and familiar environments to clients.

Whilst eTherapy has been implemented with mixed success (Andersson et al., 2019b), a framework of support like that which is integral to the *model* (**Cavanagh et al., 2011a, 2011b; Elison et al., 2014, 2017; Gellatly et al., 2018; Lidbetter & Bunnell, 2013; Luik et al., 2017**) has recently been demonstrated to be acceptable by Richards et al. (2016), where clients identified having a 'supporter' to provide motivation, guidance, and feedback as being what that they liked most. Accessibility (Perera-Delcourt & Sharkey, 2019), convenience (being able to fit therapy into daily routine), anonymity (Beattie, Shaw, Kaur & Kessler, 2009), and flexibility of the approach (Richards et al., 2016 Ritterbrand et al., 2003) are also considered key factors in terms of acceptability.

Though eTherapy is an acceptable, accessible treatment (Brown, 2018), uptake is low with high drop-out rates (Waller & Gilbody, 2009; Rost et al., 2017). Treatment adherence and drop-out is thought to be linked to acceptability (Van Ballegooijen et al., 2014).

## 2.5 eTherapy: service delivery and peer support

eTherapy can be delivered using different programmes and delivery models, ranging from entirely self-guided, unsupported services through to guided/supported services where varying degrees of support is provided. The recent IAPT evaluation in practice of the Space from Depression programme (NICE, 2020b), where the programme was implemented differently in various services, reported mixed findings; indicating that delivery models are important in terms of outcomes and acceptability. However, studies examining eTherapy when provided as part of a stepped care approach are scarce (Andersson et al., 2019b). Support can be provided by a range of healthcare professionals from clinicians such as GPs, therapists, and psychologists through to non-clinicians, for example, PSWs.

Peer support has a long history in mental health services where individuals with lived experience have played a critical role in providing support to others similarly affected, informally and through self-help groups (Jackson, 2010) and is at the heart of the service delivery offer at Self Help Services (Lidbetter & Bunnell, 2013).

Different terms are used to describe peer support roles (see Table 8), with no formally accepted definition of a PSW (Repper & Carter, 2010).

## Table 8: Peer support role terms.

Peer volunteer, peer worker, peer befriender, patient expert, peer visitor, peer support specialist, peer educator, peer role model, peer coach, peer volunteer mentor, peer informant, lay person, peer trainer, peer visitor, peer counsellor, peer broker, peer supporter, consumer advisor, health coach, and peer buddy.

In this thesis, the term 'peer support' covers all terms associated with peer support and 'PSW' for all peer support roles.

Health Education England (2020, para. 1) state that PSWs are:

people who have lived experience of mental health challenges and choose to support others receiving services. They work towards the individual's wellbeing, giving hope and supporting recovery and their approach is built on shared experiences and empathy and is valued and supported by the NHS.

Peer support can be categorised on the nature of the interaction concerned (Bradstreet, 2006) and has 12 principles (Basset, Faulkner, Repper & Stamou, 2010) - see Table 9 - which risk

being compromised if peer support services are provided by statutory mental health services

(Basset et al., 2010).

Mutuality
Solidarity
Synergy
Sharing with safety and trust
Companionship
Hopefulness
Focus on strengths and potential
Equality and empowerment
Being yourself
Independence
Reduction of stigma
Respect and inclusiveness

Table 9: Principles of peer support (Basset et al., 2010).

Peer-supported delivery models may facilitate empowerment by giving PSWs a more active role in their recovery; addressing power imbalances often found in services and moving treatment from being controlled by experts to those with lived experience (Simon & Ludman, 2009; NHS Confederation Mental Health Network, 2013). Additionally, PSWs may act as a channel for relaying the common factors of a therapeutic relationship, instead of clinicians performing this function.

There is high quality evidence for a range of peer support interventions for depression (Bryan & Arkowitz, 2015; Dale, Caramalau, Lindenmeyer & Williams, 2008; Pfeiffer, Heisler, Piette, Rogers & Valenstein, 2011); the purposive use of self-disclosure in peer support can, in general, reframe perspectives, offer coping skills, challenge stigma, establish rapport, and convey empathy and understanding of personal struggle (Marino, Child & Campbell Krasinski, 2016).

Finding ways of providing peer or professional support has been suggested to be critical to seeing the full potential of eTherapy (Knowles, 2014), and research has shown that PSWs offer 'more authentic empathy' (Bailie & Tickle, 2015, p.48).

The PSW role is fulfilled by eTCs, resulting in non-clinical, peer support being fully integrated into the eTherapy *model* operating at Self Help Services; nationally recognised in the field of user and peer-led mental health service provision (Lidbetter & Bunnell, 2013). This is discussed as the *model* (Cavanagh et al., 2011a, 2011b; Elison et al., 2014, 2017; Gellatly et al., 2018; Lidbetter & Bunnell, 2013; Luik et al., 2017) was developed.

Employment of PSWs is key to recovery-focused mental health service delivery (Moll, Holmes, Geronimo & Sherman, 2009). The eTC role is a paid, non-clinical, PSW role which enables those with lived experience of anxiety and depression to gain valuable employment experience. eTCs have responsibility for co-ordination of the eTherapy service and are supported by a small team of volunteers (in the venue *model*). Their role is to provide motivational support, guidance, instruction, encouragement, feedback, and praise; all factors known to be associated with strong perceptions of the therapeutic alliance (Perera-Delcourt & Sharkey, 2019; Schneider, Hadjistavropoulos & Faller, 2016). eTCs also provide guidance regarding homework tasks, this being important since uncompleted homework has been reported to be associated with poorer outcomes in eTherapy treatment of GAD (Paxling et al., 2013). eTCs are appropriately self-disclosing, this being associated with better outcomes in eTherapy for depression in a study where therapists provided support (Holländare et al., 2016). As eTCs appropriately share their experiences of having lived with a mental health condition and of getting to a place where they feel able to enjoy a quality of life that is not adversely affected by anxiety or depression, they are often seen as role models; instilling hope in clients that recovery is possible - this being a phenomenon reported in the peer support literature (Mahlke et al., 2017; Solomon, 2004). Similarly, the sharing of experiences of coping with anxiety and depression by peer supporters, (in this case eTCs and eTherapy volunteers), has been identified as being something that can lead to clients feeling empowered to take control of their own self-care (Rogers et al., 2007; Solomon, 2004) and is a key component of the *model* as well as the wider charity's service offer.

A decade ago, Self Help Services successfully campaigned to have eTCs (an entirely new role, unique to the *model*, developed under the leadership of the thesis' author), recognised at national level as 'low intensity IAPT workers.' This raised the profile of the eTherapy PSW role and, specifically, that of the eTC. This meant IAPT underspend could be reallocated to eTherapy, enabling Self Help Services to expand its team of eTCs instead of funding clinical roles such as PWPs. This resulted in growth of the charity's eTherapy services and their profile.

Aside from the studies that form the portfolio of works, the integration of peer support with eTherapy is under-investigated. Only one study detailed the development of a peer-support protocol (Ray, Kemp, Hubbard & Cucciare, 2017) and, in general, studies focusing on eTherapy supporter type are few. Whilst Shandley et al. (2008) found eTherapy for panic disorder could be supported by suitably trained health professionals, the study failed to specify whether supporters should be clinically trained. Hollis et al. (2017) report a lack of research detailing levels of human support provided. Mohr et al. (2011) advocated incorporation of mental health clinicians or coaches into eTherapy, seeing this as integral to its success, and Robinson et al. (2010) recommend large scale trials on the clinical effectiveness and acceptability of technician-assisted (non-clinical) eTherapy.

Whilst peer support is a key to the *model* (Cavanagh et al., 2011a, 2011b; Elison et al., 2014, 2017; Gellatly et al., 2018; Lidbetter & Bunnell, 2013; Luik et al., 2017), by contrast, the IAPT manual (National Collaborating Centre for Mental Health, 2018) recommends that clinicians should support eTherapy stating:

digitally enabled therapies can achieve comparable outcomes to face-to-face therapy when the same therapy content is delivered in an online format that allows much of the learning to be achieved through patient self-study, reinforced, and supported by a suitably trained clinician (National Collaborating Centre for Mental Health, 2018, p.41).

This recent focus on clinicians has affected the PSW workforce, forcing providers to adopt clinical models; which is at odds with mental health policy's focus on expanding and diversifying peer support in the mental health workforce (Health Education England, 2017).

The portfolio of works contributes to eTherapy knowledge in respect of the integration of non-clinical, peer support and of the effective and acceptable delivery of eTherapy in real-world, community settings.

# 2.6 Section discussion

Innovative, emerging technologies, such as Artificial Intelligence (AI) and eTherapy, can be used to meaningfully engage clients in online interventions that, in turn, can result in positive health and social outcomes (Vasilica & Ormandy, 2017). eTherapy is effective in the treatment of anxiety and depression (Andrews, Cuijpers, Craske, McEvoy & Titov, 2010), and as effective when compared to face-to-face CBT (Andersson et al., 2014; Andersson, Titov, Dear, Rozental & Carlbring, 2019b; Carlbring, Andersson, Cuijpers, Riper & Hedman-Lagerlöf, 2018).

Programme type is less important in relation to effectiveness, however supported eTherapy appears to be more effective than unsupported eTherapy (Andersson & Cuijper, 2009; Andersson et al., 2019b; Baumeister et al., 2014). Many questions remain, however, regarding support in eTherapy (Andersson & Cuijpers, 2009) and how client improvement

may be impacted by the duration, frequency, and expertise of any human support provided (Cuijpers et al., 2009; Shandley et al., 2008). Furthermore, further research is warranted on the nature of assistance provided (Schneider, Hadjistavropoulos & Faller, 2016).

High drop-out rates are reported in eTherapy (Melville, Casey & Kavanagh, 2010; Rost et al., 2017; Waller & Gilbody, 2009), though positive outcomes are associated with greater treatment adherence (Van Ballegooijen et al., 2014) and, in this regard, acceptability and effectiveness are inextricably linked.

Numerous eTherapy terms and delivery model permeations have presented research challenges; moreover, most research to date has focused on programme effectiveness, as opposed to focusing on moderators and mediators of eTherapy (Andersson et al., 2019b), with research typically being quantitative and of a positivist approach – focussing on effect sizes and statistics instead of client experience and acceptability. There has also been a distinct absence in the literature of descriptions of service delivery models and, to a degree, the implementation of eTherapy within routine care settings (Andersson et al., 2019b). Scalability of eTherapy can, however, be achieved by replacing clinical supporters, and destigmatisation of mental health issues can be addressed by utilisation of peer support in eTherapy delivery models (Wilhelm et al., 2020), as per the *model* (Cavanagh et al., 2011a, 2011b; Elison et al., 2017; Gellatly et al., 2018; Lidbetter & Bunnell, 2013; Luik et al., 2017). Peer and non-clinical support are, however, under-used in eTherapy, despite the mental health workforce being in a declining state (British Medical Association, 2020).

Providers have been encouraged to publish on client experience and acceptance of services (Fleming et al., 2018). Doing so could result in the expansion of peer and non-clinical supported eTherapy services; allowing clinicians to concentrate on other areas of service provision. Further research is needed to identify whether clinician-supported eTherapy is more effective and acceptable than non-clinician supported eTherapy (**Gellatly et al., 2018**).

Overall, eTherapy has unfortunately not received the same amount of focus as other areas of IAPT, where face-to-face delivered low- and high-intensity interventions have been extensively researched. This is despite its undoubted potential to increase access and deliver positive outcomes. Given the scale of the problem of anxiety and depression, it is critical that eTherapy is routinely embedded as a mainstream, low-intensity treatment option.

The portfolio of works demonstrates the unique contribution made to eTherapy literature through demonstrating that an accessible, non-clinical, peer-supported *model* of eTherapy (**Cavanagh et al., 2011a, 2011b; Elison et al., 2014, 2017; Gellatly et al., 2018; Lidbetter & Bunnell, 2013; Luik et al., 2017**) is both effective and acceptable when delivered in a real-world setting as a low-intensity IAPT service for clients experiencing anxiety and depression.

#### 2.7 Section summary

This section has examined eTherapy in the management of anxiety and depression, including clinical effectiveness, acceptability, and service delivery models. The absence of inclusion of peer support in eTherapy service delivery models is detailed with the *model* highlighted as addressing this gap where peer support is both a fundamental and integral component, fulfilled by the eTC role.

This gap is addressed in section three, which offers a rationale for the published works, thesis aim and objectives, and details the unique contribution of the included articles to eTherapy knowledge that collectively describe, review, and evaluate the effectiveness and acceptability of a non-clinical, peer-supported *model* of eTherapy service delivery (**Cavanagh et al., 2011a, 2011b; Elison et al., 2014, 2017; Gellatly et al., 2018; Lidbetter & Bunnell, 2013; Luik et al., 2017**).

#### Section Three

This section provides the rationale for the published works, thesis aim and objectives, and details the unique contribution made to eTherapy knowledge achieved collectively through the included articles, as well as providing further detail on the *model* (**Cavanagh et al., 2011a, 2011b; Elison et al., 2014, 2017; Gellatly et al., 2018; Lidbetter & Bunnell, 2013; Luik et al., 2017**).

#### **3.1** Background and rationale for the published works

eTherapy was considered a new concept at the time when the *model* was developed and, whilst its effectiveness had been proven in the treatment of anxiety and depression (Andrews, Cuijpers, Craske, McEvoy & Titov, 2010), little was known about how to successfully implement eTherapy in services (Drozd, Vaskinn, Bergsund & Haga, 2016), and its acceptability was still largely unknown (Cavanagh et al., 2009).

Respondents of a national survey of 500 British Association for Behavioural and Cognitive Psychotherapies (BABCP) accredited CBT therapists in 2004, on the use of eTherapy, said that they would need to learn more about eTherapy before they would use it, and almost half had concerns about the potential increase in its use. Furthermore, respondents also reported a perceived lack of evidence for eTherapy (Whitfield & Williams, 2004), however it was not only CBT therapists who shared these views at the time, but others including NHS commissioners. Given this context, it was necessary to concentrate on quickly building the evidence base for the effectiveness of the *model* to secure a mainstream NHS commission for the service in one geographical area, beyond the initial pilot.

In the early to mid-2000s, providers were routinely required to submit client case vignettes, testimonials, and other monitoring reports to funders and commissioners as part of the terms and conditions of receiving funding. Hence, there was familiarity and adherence with this form of reporting. To ensure consistency with this established norm, the **Cavanagh et al.** (2011a) study was undertaken to describe and share knowledge of the *model*. **Cavanagh et al.** (2011a) also included a range of client case studies; frequently used as a method to undertake qualitative analysis (Stake, 2005; Yin, 2009, 2006) and, in this instance, to illustrate client acceptability as well as reporting on overall experience and onward journey – for example, securing of employment, volunteering opportunities, etc.; all important metrics for commissioners and funders.

Whilst Self Help Services was renowned for treading new ground, (e.g., being commissioned by the NHS to deliver user-led support groups as early as 2000), formal documentation of the unique approach was lacking. **Lidbetter and Bunnell (2013)** recorded this approach by providing a qualitative description of the charity's growth and development, highlighting the importance of lived experience of staff and volunteers in the charity's delivery of nonclinical, peer-supported services.

Whilst **Lidbetter and Bunnell (2013)** and **Cavanagh et al. (2011a)** helped raise the profile of the organisation and its services and, to an extent, qualitatively documented the *model*, it was clear that because of the then country-wide implementation of IAPT, that outcome reporting had shifted from the former qualitative, case-study/vignette methodology to that of a positivist approach with quantitative research favoured where client progress was measured in numbers. Cognisant of this, a quantitative analysis of the *model*, benchmarking outcomes against key IAPT outcomes (**Cavanagh et al., 2011b**), was undertaken. This demonstrated that the *model* was clinically effective for the first time when delivering BtB<sup>TM</sup> in the treatment of adults with depression or anxiety within a low-intensity IAPT service, delivered in the community where over a third of referrals were self-originated. This evidence enabled a mainstream NHS commission for the service and interest outside of Greater Manchester.

By 2012, commissioners were increasingly interested in providers developing services for those with dual diagnosis presentations, as it was expected that this would become one of the biggest issues that the country would face, with suggestions that as many as 75-85% of individuals with substance misuse issues would also experience clinical levels of mental ill health (Weaver et al., 2003). It was important to demonstrate that, in addition to being able to support those with symptoms of anxiety and depression, the *model* could also be used to support those with comorbid substance misuse issues. **Elison et al. (2014)** achieved this by demonstrating that the *model*, when used to deliver BFO, was clinically effective and appropriate for some individuals presenting with dual diagnosis issues, as well as depression and anxiety.

As eTherapy became established as a mainstream intervention, so too followed an expansion in the range and availability of eTherapy programmes (Bennion et al., 2017). Quantitative service evaluations (Elison et al., 2017; Luik et al., 2017) showed that the *model* could achieve clinically effective results using different programmes; demonstrating versatility and, critically, that programme type affected outcomes less than previously thought. Diversifying the *model* in this way also addressed client desire for greater programme choice, and facilitated tailoring of the service to client presentation, resulting in expansion of the *model's* scope to include clients with sleep difficulties (Luik et al., 2017). Elison et al. (2017) also provided important evidence for the charity in practice, in respect of programmes, in terms of determining which might be used going forward; an issue of importance since there was considerable variance in their cost as well as addressing comorbidity; an issue affecting many of the clients accessing the charity's services. Whilst the effectiveness of the remote *model* alongside the venue *model* via a hybrid configuration had been achieved through **Elison et al.** (2014, 2017) and **Luik et al.** (2017), the remote *model* had not been the exclusive subject of a quantitatively focused, service evaluation. This was achieved through **Gellatly et al.** (2018), which demonstrated that the remote *model* could meet IAPT key performance targets and could produce outcomes comparable to other eTherapy services operating in IAPT; enabling the evolved remote *model* to then be widely marketed to NHS commissioners across Greater Manchester and beyond. This was important in practice as the venue *model* was becoming increasing unpopular and costly to operate.

Aside from the Portfolio of Works that describe the *model* (Cavanagh et al., 2011a, 2011b; Elison et al., 2014, 2017; Gellatly et al., 2018; Lidbetter & Bunnell, 2013; Luik et al., 2017), only two papers have been published on peer-supported eTherapy (de Vares, 2007; Ray et al., 2017); both of which are limited in scope. The former described a protocol for an eTherapy service yet to be established, whilst research undertaken by Ray et al. (2017) was limited to the investigation of possible insertion of peer support into an existing clinical model.

By addressing gaps in the literature on the role of peer support in eTherapy service delivery and of the effectiveness and acceptability of non-clinical, peer-supported eTherapy in realworld settings, this portfolio of work examines the development, effectiveness, and acceptability of the *model* (Cavanagh et al., 2011a, 2011b; Elison et al., 2014, 2017; Gellatly et al., 2018; Lidbetter & Bunnell, 2013; Luik et al., 2017) in the management of anxiety and depression in adults, developed at Self Help Services, under the author's direction.

# 3.2 Thesis aim and objectives

The aim of this thesis is to describe the development and evaluation of a non-clinical, fully peer-supported eTherapy *model* (Cavanagh et al., 2011a, 2011b; Elison et al, 2014, 2017; Gellatly et al., 2018; Lidbetter & Bunnell, 2013; Luik et al., 2017) in the management of anxiety and depression in adults.

# The key objectives are to:

- 1 Examine the development of the *model*
- 2 Review and evaluate the clinical effectiveness of the *model*
- 3 Evaluate the acceptability of the *model*

# **3.3** Contribution of included papers

The candidate has made a unique contribution to each of the seven papers under scrutiny, and

as a whole – see Table 10.

Study	Study Objectives	Paper overview & contribution to the development of the model	Author contribution	Thesis Objectives
Paper 1:	To describe an	This paper	Conceptualisation	Examine the
Cavanagh, K.,	innovative, third	described, for the	of peer-supported	development
Seccombe, N.,	sector, non-	first time, the	eTherapy <i>model</i> .	of the <i>model</i>
Lidbetter, N., &	clinical, peer-	implementation of,	Responsible for	(Objective
Bunnell, D.	supported,	and service	operational	1).
(2011a).	eTherapy model	delivery model for,	development and	
Supported,	commissioned	the novel, peer-	implementation of	
service-user led,	within Greater	supported venue	the venue model.	
computerised	Manchester.	model when	Agreeing	
cognitive		operating as a low-	methodological	
behavioural		intensity IAPT	approach to take	
therapy (CCBT)		service for anxiety	with partners.	

Table 10:	Details of	f the pub	lished work	ks and the o	contribution	made.

self-help		and depression in	Involved with the	
clinics. Journal of		Greater	preparation,	
Public Mental		Manchester.	writing and	
Health, 10(4),		The paper	subsequent	
225-233.		demonstrated that	revisions of the	
		the model was able	manuscript	Evaluate the
		to meet a diversity	including decision	acceptability
		of need and that	to include case	of the model
		clients accessing	studies and	(Objective
		the service are	inclusion of	3).
		representative of	pictorial	
		the local population	description of	
		and of clients that	client pathway.	
		access IAPT	Responsible for	
		services instead of	ensuring overall	
		the service being	adherence to	
		supported by	service	
		clinicians (PWPs)	governance	
		as is typical of	regulations.	
		eTherapy services	Led on the	
		delivered through	internal	
		IAPT (Thew,	dissemination	
		2020).	strategy.	
		This paper		
		provided detail on		
		eTherapy client		
		experience by		
		examining the		
		experiences of four		
		clients via case		
		studies that		
		accessed the venue-		
		based model:		
		demonstrating its		
		acceptability.		
Paper 2:	To evidence the	This paper	Conceptualisation	Examine the
Cavanagh, K.,	implementation of	benchmarked the	of the peer-	development
Seccombe, N., &	a service user-led,	venue-based	supported	of the model
Lidbetter, N.	third sector	model's outcomes	eTherapy model.	(Objective
(2011b). The	eTherapy clinic	when used to	Responsible for	1)
Implementation	via a quantitative	deliver BtB <sup>™</sup> for	operational	
of Computerized	service	outcomes and	development and	
Cognitive	evaluation.	service uptake	implementation of	
Behavioural		including	the venue <i>model</i> .	
therapies in a	To report on	employment status,	Agreeing	
Service User-Led,	uptake and	pre- and post-	methodological	
Third Sector Self	outcomes for the	treatment, and the	approach to take	
Help	eTherapy	initial level of	with partners and	
clinic. Behaviour	programme	unsuitable referrals	the research	Review and
al and Cognitive	BtB <sup>™</sup> measured	compared to NHS	question.	evaluate the
Psychotherapy, 3	using the IAPT	IAPT	Responsible for	clinical
9(4), 427-442.	Minimum Data	demonstration sites,	ensuring overall	effectiveness
	Set (IAPT MDS)	reporting	adherence to	of the <i>model</i>
	and Key	comparable results	service	(Objective
	Performance	regarding key		2).

	Indicators in	outcome measures	governance	
1	relation to	and recovery rates.	regulations.	
	extending access	This paper	Provided advice	
1	to recovery	demonstrated for	and guidance to	
	(Department of	the first time, that	Data Lead in	
]	Health, 2008),	the peer-supported,	terms of collection	
	when delivered	eTherapy model	and analysis of	
,	via the <i>model</i> .	was effective and	data sample.	
		capable of meeting	Involved with the	
,	To test the	IAPT targets,	preparation,	
	generalisability of	without the	writing and	
	BtB <sup>TM</sup> in terms of	involvement of	subsequent	
]	previously proven	clinicians, - this	revisions of the	
	efficacy and	being a key step	manuscript.	
	effectiveness in	towards securing an	Led on the	
	NHS services, by	NHS commission	internal	
	studying pre- and	of the service.	dissemination	
	post-outcomes	Furthermore, the	strategy.	Evaluate the
	when delivered in	paper reported that		acceptability
1	the community	the <i>model</i> was able		of the <i>model</i>
	via the venue	to deliver three		(Objective
i i	model via a	times as many		3).
]	pragmatic study.	sessions compared		
		to the most engaged		
		IAP1		
		demonstration site;		
		accoptability and		
		future potential in		
		terms of client		
		throughput and		
		ability of the		
		service to		
		contribute towards		
		meeting of IAPT		
		prevalence targets.		
		The paper reported		
		on client		
		satisfaction using		
		the IAPT Patient		
		Experience		
		Questionnaire –		
		demonstrating high		
		satisfaction rates.		
		Inis paper		
		provided evidence		
		of the importance		
		on referral source,		
		and that cheffes that		
		likely to take up the		
		service and to		
		complete at least		
		two sessions, as		
		well as being		

		significantly more		
		Significantly more		
		the full e I herapy		
		programme. This		
		paper provided		
		important insight		
		into the relationship		
		between adherence,		
		and referral source		
		in the <i>model</i> .		
		The paper also		
		reported on		
		ethnicity, caseness.		
		chronic physical		
		health conditions		
		clinical		
		characteristics upon		
		storting the		
		starting the		
		етпегару		
		programme		
		(symptom		
		severity), gender,		
		medication, and age		
		and relationship		
		with referral		
		source, likelihood		
		of starting the		
		programme, as well		
		as completion rates		
		– giving new		
		information		
		regarding		
		accentability as		
		well as reporting		
		high rates of		
		angli faction		
Damar 2.	To describe the	This paper details 1	Comparison of the state	Exercises (1-
Paper 5:	To describe the	This paper detailed	Conceived the idea	Examine the
Lidbetter, N., &	growth of the	how those with	and led on the	development
Bunnell, D.	user-led charity,	lived experience of	design and write up	of the <i>model</i>
(2013). Self	Self Help	mental health	of the manuscript	(Objective
Help Services:	Services, founded	difficulties	including	1).
helping people to	by the thesis'	(specifically,	determining the	
help	author to	anxiety and	manuscript's focus,	
themselves. Ment	becoming a key	depression), can be	selection of case	
al Health and	provider of NHS	intrinsically and	studies and the	
Social	commissioned,	integrally involved	inclusion of the	
Inclusion, 17(2),	primary care	in the delivery of	'lessons learned'	
76-81.	mental health	NHS commissioned	narrative.	
	services.	primary care	Led on the external	
	including	mental health	and internal	
	eTherapy	services	dissemination	
	services in the	(specifically	strategy	
	North West of	eTherany services)	strattegy.	
	England	It detailed the		
	Eligiano.	abarity's ans-th		
1	1	charity s growth		1

		and expansion into other areas of service delivery, including differentiation of the eTherapy <i>model</i> into new locations, including a Category A prison; demonstrating the versatility of the <i>model</i> in terms of where it could be delivered i.e. from venues very different to the community centres where the <i>model</i> had originally operated from. The paper provided evidence that personal experiences are 'vital tools' in helping others work		Examine the development of the <i>model</i> ( <b>Objective</b> 1). Evaluate the acceptability of the <i>model</i> ( <b>Objective</b> 3)
		through their own difficulties in the eTherapy <i>model</i> and that the peer- support element is an integral component of the <i>model</i> .		3).
Paper 4: Elison, S., Ward, J., Davies, G., Lidbetter, N., Hulme, D., & Dagley, M. (2014). An outcomes study of eTherapy for dual diagnosis using Breaking Free Online. Advances in Dual	To examine outcomes for clients experiencing substance dependency and comorbid mild to moderate mental health problems accessing the dual diagnosis eTherapy programme	This paper provided evidence of the effectiveness and clinical effectiveness of the Breaking Free Online eTherapy programme for clients experiencing dual diagnosis difficulties, when accessed via a hybrid version of	Provided approval for the research team to access the database of anonymised clinical data from clients of Self Help Services' that had accessed the <i>model</i> . Responsible for ensuring overall adherence to service	Examine the development of the <i>model</i> ( <b>Objective</b> 1). Review and evaluate the clinical effectiveness of the <i>model</i> ( <b>Objective</b>
<i>Diagnosis</i> , 7(2), 52-62.	Breaking Free Online, via the <i>model</i> .	the <i>model</i> where face-to-face and telephone support are made available, according to client choice. This paper demonstrated the <i>model's</i> versatility	governance regulations. Provided comments on the manuscript ahead of publication. Led on the internal	2).
		in supporting a new	dissemination	
--------------------------------------	-------------------	------------------------------------	----------------------	----------------------
		client group for	strategy.	
		those first time –		
		those with dual		
		diagnosis issues i.e.		
		anxiety and/or		
		depression as well		
		as substance misuse		
		issues This was of		
		relevance because		
		of the comorbidity		
		of such issues		
		Therefore the		
		paper showed that		
		the <i>model</i> had a		
		wider reach than		
		was initially		
		demonstrated in		
		Cavanach et al		
		Cavanagn et al., 2011b		
Donor 5.	To ovplore the	This paper	Provided approval	Examina tha
Flicon S Word	foosibility and	novided further	for the research	davalonment
Liison, S., waiu,	outcomes of the	provided further	toom to account the	of the model
J., Williams, C., Espis C. Davies	model vie	evidence of the	detebase of	( <b>Objective</b> )
C Dugdala S	avamination of	three oThereny	anonymisod clinical	(Objective
G., Duguale, S., Smith V	LADT outcomes	nee e merapy	data from alignta of	1).
(2017)	for alignets	delivered through a	Calf Halp Services?	Davious and
(2017). Eassibility of a	ongoging with	bybrid version of	that had accounted	Neview allu
I tasionity of a	three different	the model where	that had accessed	evaluate the
based aThereny		face to face and	Deconomials for	offootivonooo
based, ernerapy	етпегару	talenhone and	Responsible for	effectiveness
	through a new and	telephone support	ensuring overall	
Service III Greater	nough a pre- and	were made	adherence to	(Objective
manchester.	post-test service	available,	service	<i>4</i> ).
repeated-	evaluation.	according to chent	governance	
measures and		choice.	regulations.	
between-groups		The significance of	Provided comments	
Life to the Full		this was that the	on the manuscript	
Life to the Full		paper demonstrated		
interactive,		that the <i>model</i>	publication.	
Sleepio <sup>1</sup> and			Led on the	
Breaking Free		positive outcomes	internal	
Unline at Sell		when used to	dissemination	
Help Services.		deliver 3 different	strategy.	
BMJ Open, 7(7).		elnerapy		
		programmes of		
		unifering formats		
		and components.		
		PTEVIOUSIY		
		enecuveness had		
		only been		
		established in		
		Cavanagn et al.,		
		<b>2011D</b> for BtB <sup>TM</sup>		
		and via Elison et		
		<b>al., 2014</b> for BFO.		

		This paper		
		therefore		
		demonstrated that		
		the <i>model</i> could		
		support client		
		choice regarding		
		eTherapy		
		programme		
		selection of		
		importance given		
		the emphasis on		
		collaboration as		
		well as the overall		
		ethos of Self Help		
		Services being that		
		of putting clients in		
		control of their		
		treatment, support		
		and ultimately,		
		their recovery.		<b>.</b>
Paper 6:	To evaluate the	This paper detailed	Contributed to the	Examine the
Luik, A. I.,	implementation of	the successful	design of the study,	development
Bostock, S.,	the eTherapy	implementation of	leading on the site's	of the <i>model</i>
Chisnall, L.,	programme,	the remote version	strategic delivery of	(Objective
Kyle, S. D.,	Sleepio <sup>™</sup> , when	of the <i>model</i> where	the project	1).
Lidbetter, N.,	delivered via the	clients accessing	including ensuring	
Baldwin, N., &	remote <i>model</i> for	the eTherapy	adherence to	
Espie, C. A.	clients	programme,	service governance	
(2017). Treating	experiencing	Sleepio <sup>™</sup> , are	regulations.	
depression and	anxiety and/or	supported entirely	Contributed to the	Review and
anxiety with	depression.	via the telephone	final draft of the	evaluate the
digital cognitive		by non-clinical,	manuscript.	clinical
behavioural		peer supporters.	Played a critical	effectiveness
therapy for		The paper	role in the	of the model
insomnia: a real		demonstrated that	dissemination of	(Objective
world NHS		the model achieves	the paper to	2).
evaluation using		a recovery rate	relevant service	
standardized		greater than the	users and	
outcome		IAPT target	professional	
measures.		recovery and	populations.	
Behavioural and		reliable recovery		
Cognitive		rate and a treatment		
Psychotherapy, 4		rate that is above		
5 (1), 91-96.		the IAPT average		
		for clients		
		experiencing		
		anxiety and/or		
		depression as well		
		as poor sleep,		
		therefore		
		demonstrating the		
		model's versatility		
		both in terms of its		
		composition (the		
			1	1

Paper 7: Gellatly, J., Chisnall, L., Seccombe, N., Ragan, K., Lidbetter, N., & Cavanagh, K. (2018). @ Home eTherapy service for people with common mental health problems: an evaluation. Behavioural and Cognitive Psychotherapy, 4 6 (1), 115-120.	To evaluate the implementation of an innovative @ home/remote eTherapy model by reporting service outcomes and comparing these with national IAPT service data.	remote <i>model</i> ) and of the client groups it can support. i.e. clients with anxiety and/or depression as well as sleep issues. This was of relevance because of the comorbidity of such issues and the bidirectional relationship between sleep and poor mental wellbeing, specifically anxiety, (Alvaro, Roberts & Harris, 2013). Therefore, the paper showed that the <i>model</i> had a wider reach than was initially demonstrated in <b>Cavanagh et al.,</b> 2011b and <b>Elison et al., 2014, 2017</b> . This paper reported on outcomes by providing data from the peer-supported @ home/remote <i>model</i> , comparing these against national IAPT outcomes and outlining the <i>model</i> where peer support and lived experience is central to this evolved model of eTherapy service delivery. Higher recovery rates are reported than the previous study undertaken on the venue <i>model</i>	Instigated the research project. Refined aims and objectives of the research ensuring compliance with service governance regulations. Contributed to the preparation, writing and revisions of the manuscript. Played a critical role in the dissemination of the paper to relevant service user and professional populations.	Examine the development of the <i>model</i> ( <b>Objective</b> <b>1</b> ). Review and evaluate the clinical effectiveness of the <i>model</i> ( <b>Objective</b> <b>2</b> ).
		Higher recovery rates are reported than the previous study undertaken on the venue <i>model</i> (Cavanagh et al., 2011b) which helped build the avidance base for	user and professional populations.	
		this, the remote version of the		

	model – needed to	
	secure the future re-	
	commissioning of	
	the service by the	
	NHS.	
	The paper reports	
	on the flexibility of	
	the model and the	
	possible	
	contribution that	
	this has in terms of	
	engagement and	
	outcomes and was	
	operationally	
	important since by	
	2018, many clients	
	were choosing not	
	to access the venue	
	version of the	
	<i>model</i> ; opting	
	instead for the	
	remote version	
	because of	
	convenience.	

## 3.4 The non-clinical, peer-supported eTherapy model

The non-clinical, peer-supported eTherapy *model* is depicted in Figure 1, having been developed in all its iterative parts via the portfolio of works (**Cavanagh et al., 2011a, 2011b; Elison et al., 2014, 2017; Gellatly et al., 2018; Lidbetter & Bunnell, 2013; Luik et al., 2017**).

Figure 1: The non-clinical, peer-supported, eTherapy model



#### Description of the model

The *model* originally operated from community venues with a suitable space (typically IT suites/community cafés, learning zones, schools, and hospitals), where up to four clients at any one time accessed the service via stand-alone PCs/laptops. In the venue *model*, the service is co-ordinated by an eTC (an individual typically with lived experience of a mental health condition who has previously accessed the *model*). eTCs are supported by, on average, three volunteers (individuals with lived experience of a mental health condition, typically anxiety and depression). Both roles are described in Section 2.5.

#### Referral and assessment

Clients are referred through a variety of means, including GP and self-referral (but must be registered with a GP), and attend an assessment undertaken by the eTC, during which risk and symptom severity is assessed to determine suitability for the service using the PHQ-9 and GAD-7. Confidentiality is discussed as is completion of sessional measures – this being a monitoring and evaluation requirement of the commissioning body. eTCs verify clients do not satisfy any exclusion criteria (see Table 11) and are able to meaningfully engage with programme content.

#### Table 11: eTherapy service exclusion criteria.

Presence of an organic brain disorder such as dementia

Client is in an acute phase of substance misuse

Client is in an acute phase of psychosis

Client has a moderate to severe learning difficulty

Client has a severe mental illness presentation (e.g., bipolar disorder, schizoaffective disorder, etc.)

Client < 18 years of age

#### eTherapy programmes available through the model

Clients are provided with a description of programmes available, which include BtB<sup>TM</sup>, Sleepio<sup>TM</sup> (Espie et al., 2012), BFO (Elison et al., 2013), LLTTF<sup>TM</sup> (Williams, n.d.), FearFighter<sup>TM</sup> (Marks et al., 2004; Schneider et al., 2005), and Silvercloud Health (Sharry et al., 2013; Richards et al., 2015). Programme availability is dependent on geographical location of the service provided and is collaboratively selected with clients who are then explained how the service works.

When the *model* was first developed, programme access was limited as only a few eTherapy programmes had been recommended by NICE (2002) and approved by NHS commissioners for local use. Today, programme selection is determined by the client's presenting problem and goals, and a much wider selection of programmes are available. The portfolio of works details the development of the evidence base for the *model's* effectiveness and of its acceptability in clients presenting with issues such as insomnia and dual diagnosis issues, using the disorder-specific programmes of Sleepio<sup>TM</sup> and BFO, respectively.

#### Client support, eTCs and volunteers

eTCs investigate client computer experience and specific needs via a short, supportive conversation to address any pre-existing anxiety about computer use, as well as to identify the presence of any disabilities that may necessitate adaptations. As eTCS have lived experience of a mental health condition, this experience is deployed to support clients on their road to recovery through offering a personal approach to support and via appropriate self-disclosure. Emphasis is also placed on the availability of eTherapy volunteers (individuals who also have lived experience of a mental health condition) for assistance, peer support and guidance during sessions. Additionally, clients are advised that they may contact the eTC in between sessions if, for example, they struggle with homework. The peer support element of the overall support offer, (visible through the eTCS and volunteers), conveys hope to clients

that it is possible to manage anxiety and depression to a point such that they do not continue to adversely affect daily functioning. At the end of the initial appointment, the client signs a service contract and watches an introductory video for the programme that it has been collaboratively agreed they will undertake, after which, the first session is organised. Thereafter, sessions typically take place on a weekly basis for six to eight weeks on average, though duration varies per programme.

#### Development of the remote model

The *model* was further developed to allow clients to access the service remotely (**Gellatly et al., 2018**); instigated as many clients expressed difficulties accessing the venue *model* due to work and other commitments. This, along with venue hire costs as well as the cost of IT hardware such as laptops etc. (which needed to be typically replenished every three years), gave rise to the development of the remote *model*. In some areas, a hybrid *model* operated, where eTCs assessed clients face-to-face (typically at Self Help Services' community venue); thereafter, clients were supported remotely by telephone by the eTC.

eTCs solely deliver the remote *model*; undertaking weekly support calls of approximately 15-20 minutes, during which programme progress reports (if available) along with goals are reviewed, as well as risk and safeguarding issues identified. The IAPT MDS is completed, and a general discussion takes place focussed on the module undertaken and any challenges encountered, along with enquiry into how clients plan to implement the intervention covered. If risk is identified at any point, a full risk assessment is carried out by the eTC and relevant actions taken, including liaison with the client's GP and/or referral to crisis services. This, and any other actions, are documented (for example, signposting to other resources), and the next support call date confirmed.

Upon completion of therapy, clients are reminded that they may re-access the programme as a self-help aid for a specific period (this varies per programme) and are discharged; with a copy of the discharge report provided to the client's GP.

### 3.5 Section summary

This section has provided the background to and rationale for the published works, detailed the thesis aim and objectives, and outlined the unique contribution of the published works to eTherapy literature. The *model*, which is the focus of the published works (**Cavanagh et al.**, **2011a**, **2011b**; **Elison et al.**, **2014**, **2017**; **Gellatly et al.**, **2018**; **Lidbetter & Bunnell**, **2013**; **Luik et al.**, **2017**), is described with all its iterations (venue, remote, and hybrid versions).

The next section will provide an overview of the papers that form the portfolio of works, including their strengths and limitations, followed by a critique by thesis objective.

#### Section Four

#### 4.1 **Overview of the publications**

An overview of the papers including their strengths and limitations, followed by a critique by thesis objective, is presented in this section.

All studies (**Cavanagh et al., 2011a, 2011b; Elison et al., 2014, 2017; Gellatly et al., 2018; Lidbetter & Bunnell, 2013; Luik et al., 2017**) describe aspects of the *model* and its development, including the original venue *model*, remote *model*, and the hybrid *model* (Objective 1).

Cavanagh et al. (2011b), Elison et al. (2014, 2017), Gellatly et al. (2018) and Luik et al. (2017) are all quantitative service evaluations where pre- and post-client outcome data was analysed. The IAPT MDS was collected in full (Cavanagh et al., 2011b; Gellatly et al., 2018) for the venue and remote *models* respectively, with outcomes benchmarked against national IAPT outcomes (Cavanagh et al., 2011b; Gellatly et al., 2018) to demonstrate effectiveness (objective 2).

Elison et al. (2014, 2017) and Luik et al. (2017) demonstrate the effectiveness of the *model* (objective 2) in achieving outcomes for clients with substance misuse issues and insomnia, in addition to anxiety and depression.

Finally, **Cavanagh et al. (2011a)** and **Lidbetter and Bunnell (2013)** are qualitative studies that, as well as describing the *model* (Objective 1), evaluate its acceptability (Objective 3).

#### 4.2 **Presentation and critique of the publications**

Paper 1: Cavanagh, K., Seccombe, N., Lidbetter, N., & Bunnell, D. (2011a). Supported, service-user led, computerised cognitive behavioural therapy (CCBT) self-help clinics. *Journal of Public Mental Health*, 10(4), 225-233.

This descriptive paper includes four qualitative case studies, and describes the implementation, operational delivery, management, and development of the venue *model* (Objective 1), its impact on the population served (Objective 3) and relevance in terms of UK mental health strategy, which centred on expanding availability and increasing accessibility of evidence-based interventions in the treatment of anxiety and depression.

**Cavanagh et al. (2011a)** describes the user-led nature of the charity to provide context to the *model* and details funding and commissioning arrangements, along with a description of venues it operated from (Objective 1). Information regarding referral source (to explain the *model*'s location in relation to other mental health services) and client pathway including the nature, functions, and duration of touchpoints with eTCs and volunteers and their role is provided (Objective 1). Outcome measures (IAPT MDS) are described, number of clients referred and information on programmes (and their content) available through the *model* at the time: BtB<sup>TM</sup>, LLFT<sup>TM</sup> and FearFighter<sup>TM</sup>. A summary of the service's key performance indicator, the IAPT recovery rate, is provided with suggestions as to why clients who have accessed the *model* do not meet the criterion for recovery, and what happens in terms of their potential onward pathway. Information regarding programme licenses and implications for the overall cost of delivering the *model* is detailed, with reference made to an accompanying, quantitative evaluation of the same service - **Cavanagh et al. (2011b**), where outcome data was benchmarked against national IAPT data to demonstrate the *model*'s effectiveness when used to deliver BtB<sup>TM</sup> in a real-world setting.

Four clients provided qualitative feedback on their experience of accessing the *model* as part of the charity's standard service monitoring and evaluation process, where it was routine to

seek views of clients to develop case studies and vignettes for inclusion in reports to funders and for inclusion in other relevant publications (Objective 3), as well as for continuous service improvement purposes. Two clients were male and two female, with half referred by their GP, one via a friend, and another through their psychologist. Client presentations were depression, anxiety, phobias, low mood, and health anxiety; all of which had been experienced for differing durations (range: five months to 30 years). Client age and marital status was provided for only half of the case studies, and medication status detailed in all but one of the case studies. The case studies provided insight into:

- How clients found out about the service
- Referral mechanism
- Conditions(s) that clients were seeking help with
- Accessibility issues
- Client views on the service and the eTherapy programme; what they liked, what could be improved (Objective 3)
- Client onward pathway

The paper reports that the *model* achieves a high throughput of clients with over a third selfreferring, with intake and outcome measures suggesting that clients accessing the service are representative of both the local population and IAPT clients accessing services for anxiety and depression, see **Cavanagh et al. (2011b)** for detail. The paper also reports that clients value the service highly, including programmes and support offered via eTCs and volunteers (Objective 3), and that the *model's* ability to meet a diversity of needs is demonstrated, detail also provided in **Cavanagh et al. (2011b)**.

The paper concludes that roll-out of the successful service *model* is highly recommended and is of interest to service users, providers, and commissioners wishing to develop similar

services. Future service and research developments are specified, including examination of the effectiveness of a later iteration - the remote *model* - reported in **Gellatly et al. (2018)**.

The lack of follow-up is a limitation of the study; however, it was not possible to track client progress beyond the point of discharge because of limited resources and the study being a service evaluation (and permission not having been sought at the outset to contact clients post-discharge). A further limitation is the failure to report on uptake by clients of other mental health services and on the utilisation of step-up, step-down, step-sideways, and step-out routes that connect Self Help Services and other TS providers to the NHS and other statutory organisations. Including this information would have given helpful insight as to where the *model* sat within the overall mental health system.

Findings may have been influenced because of the majority of the study's researchers being employed by Self Help Services and therefore being involved, albeit at arm's length, with the development and delivery of the *model*. However, as this was a service evaluation, a realist approach was applicable (Robson & McCartan, 2016), and it was impossible to fully exclude those authors employed by the organisation from involvement in the study. However, the individuals concerned did not directly contact clients selected as case studies, which mitigated against any Hawthorne effect (French, 1953) in terms of client response.

Finally, client outcomes reported via the case studies cannot be solely attributed to the *model* as improvement may have resulted due to other factors including new relationships, changes in personal circumstances, and other external support. Further exploration of these possible external factors via a grounded theory approach utilising structured interviews would have overcome this.

The study's strength is that it provides a robust overview of the *model*, including outlining the client pathway; enabling potential future replication by others (Objective 1), making a unique contribution to eTherapy literature in detailing a novel model of service delivery.

Furthermore, in including heterogeneous case studies, this gives rich insight into client acceptability of the service (Objective 3), though there is lack of consistency in terms of format and information included, which is recommended good practice in case study methodology (Simons, 2009). **Cavanagh et al. (2011a)** also supplies valuable information on referral source and throughput of the service; at the time, this was highly relevant given the focus on IAPT access and uptake (Department of Health, 2012). A decade on, research has shown that GPs are still largely unaware of eTherapy as an intervention (Breedvelt et al., 2019), and so the ability to self-refer remains as important.

In summary, **Cavanagh et al. (2011a)** was the first paper to describe the implementation of, and to provide a service operating procedure for, the venue *model* (Objective 1) when offered as a low-intensity IAPT service for clients experiencing anxiety and depression in a real-world, community setting. In doing so, the paper filled a gap in eTherapy research regarding community-delivered, peer-supported, non-clinical eTherapy models of service delivery.

## Paper 2: Cavanagh, K., Seccombe, N., & Lidbetter, N. (2011b). The Implementation of Computerized Cognitive Behavioural therapies in a Service User-Led, Third Sector Self Help clinic. *Behavioural and Cognitive Psychotherapy*, 39(4), 427-442.

This study evaluated the effectiveness of the venue *model* when delivering BtB<sup>TM</sup> for the first time in a real-world, community setting (Objective 2). It tested the generalisability of findings of an earlier large scale RCT (Proudfoot et al., 2004), and several pragmatic studies undertaken in the NHS (Cavanagh et al., 2006; Learmonth, Trosh, Rai, Sewell & Cavanagh, 2008) where BtB<sup>TM</sup> was demonstrated to be efficacious and effective. The study was a pragmatic, quantitative service evaluation where data (including pre- and post-clinical outcome measures and other metrics) was retrospectively analysed over a 16-month timeframe (November 2007 - March 2009). This study was required to demonstrate the

effectiveness of the *model* to NHS commissioners to secure mainstream funding for the service and complemented the **Cavanagh et al. (2011a)** qualitative study.

A total of 510 referrals were received during the period, of which 36.7% were self-referrals, 59.7% GP referrals, and 4.1% from other sources, e.g., the local NHS primary care mental health team. Focus was placed on referral source and demographics to enable comparison with IAPT demonstration sites (Clark et al., 2009), as the importance of self-referrals in widening access via the national roll-out of the IAPT scheme had been recognised.

Of the 510 initial referrals, 386 clients accessed the *model* and were screened for suitability by the eTC via an assessment lasting between 30 minutes to an hour, and 351 clients were deemed suitable to access the model. All completed a baseline assessment where demographic information was collected, as well as: PHQ-9 and GAD-7 measures; the 10-item Clinical Outcomes in Routine Evaluation Outcome Measure - CORE-OM (Connell & Barkham, 2007); the Work and Social Adjustment Scale - WSAS (Mundt et al., 2002); the IAPT Patient Experience Questionnaire - PEQ (to measure client satisfaction) - (Objective 3); and the Department of Health's Self-Report Questionnaire, which focuses on employment (Department of Health, 2008). CORE-OM (part of IAPT MDS at that time) was included because of its ability to measure symptoms in a broad manner, therefore acting as a general wellbeing measure whilst also allowing for discrimination between clinical and non-clinical populations (Connell & Barkham, 2007). GAD-7 and PHQ-9 were completed sessionally, whilst all other outcome measures were completed at baseline and upon completion of all eight sessions of BtB<sup>TM</sup>. The rationale for using IAPT measures was that these were the measures used by Clark et al. (2009), which reported on experience and outcomes of IAPT services offering low intensity services including clinician-supported eTherapy. In using the same outcome and satisfaction measures, a direct comparison of the *model* with recently established IAPT services could be made.

Chi squared analyses (Pearson's uncorrected) were undertaken to facilitate the exploration of any relationship between two independent variables, in this case referral source (GP or and self-referred groups) and completion of at least two episodes of treatment, uptake, and treatment completion (all eight sessions completed). The decision to choose the variable of completion of a minimum of two sessions by clients was taken as this is the IAPT definition of a completed course of treatment/treatment episode (Moller et al., 2019).

The Chi squared test is a robust statistical test (Cochran, 1952) that is easy to compute, yet provides detailed information. It requires large sample sizes, as per **Cavanagh et al. (2011b)**, as it is sensitive to sample size (McHugh, 2013). This test was also used to examine the relationship between ethnicity (making a comparison between white and BME groups) and referral source, programme uptake, completion of at least two sessions, and of all eight sessions.

When comparing the impact of referral source, self-referrals were more likely to translate into clients that went onto access the service (suggesting greater motivation to access eTherapy), as opposed to GP originated referrals, and were more likely to complete at least two sessions and all eight sessions. There was no difference between white and BME referrals in terms of likelihood to be GP or self-referred, and no difference between white and BME groups regarding uptake or completion of at least two or eight sessions.

Chi squared tests (Pearson's uncorrected) were used to explore the relationship between age and programme uptake, completion of at least two or eight sessions, with no difference found between these variables. Chi squared tests were used to explore the relationship between gender and likelihood of starting the programme, and of completing at least two or all eight sessions, with no difference found. No difference was found when examining the relationship between medication status (comparing those who reported to have taken psychotropic medication with those who had not). When comparing clients with chronic physical health conditions and those without, when examining the relationship between likelihood to start the programme and of completing at least two or all eight sessions, no differences were found between the groups.

Of those providing baseline data, 246 clients met the criteria for depression and 276 for anxiety caseness criteria (NHS Digital, 2017), with 228 meeting both criteria. Analysis found that clients who met caseness criteria at assessment were more likely to access the *model* than those who did not; however. no difference was found in respect of clients above and below caseness at assessment regarding completing at least two or all eight sessions. No differences between clients who started the programme or not were found at assessment in respect of measures of depression, anxiety, general wellbeing, or work and social adjustment, though those completing all eight sessions appeared to have lower symptom levels and general distress at intake. No differences in functioning or self-impairment were detected between those who failed to complete all eight sessions of the programme.

Outcome data was analysed on an intention-to-treat basis, with all clients opting to access the service undertaking a baseline assessment who had accessed at least two sessions (n=265). This approach is advantageous since it prevents any bias from occurring when incomplete data is related to outcome; of relevance to eTherapy where adherence and completion are reported to be related to outcomes (Van Ballegooijen et al., 2014). This method also provides a conservative estimate of treatment effect (Gupta, 2011).

The study reported large effect sizes for clients who had completed at least two sessions (n=265) in relation to depression and anxiety (d=0.8 and 0.9 respectively), indicating that the measure of the magnitude of the experiment effect, in this case, BtB<sup>TM</sup> when delivered via the *model*, was large. Medium effect sizes were found in relation to wellbeing; measured via the CORE-OM and WSAS. Furthermore, following treatment, 142 of 226 clients from the initial intake who had met caseness criteria at baseline for either anxiety or depression or both, no

longer met the caseness clinical cut-off for either depression (measured by PHQ-9) or anxiety (measured by GAD-7). The move from 226 clients meeting caseness at intake to 123 at their final measurement suggested that 50% of clients moved to recovery; demonstrating the *model's* clinical effectiveness when used to deliver BtB<sup>TM</sup> (Objective 2).

Acceptability and client satisfaction were measured using the PEQ at assessment and end of treatment. Of those that had completed the questionnaire, 89% were happy with the overall experience of using the service, and 93% with the type of treatment received (Objective 3).

Employment data was incomplete, with pre- and post-data available for only 40.4% of clients who had accessed two or more sessions, and with no change in employment status recorded at the final measurement point; though some clients had moved into employment from welfare benefits, whilst others had secured volunteering roles or moved from part- to full-time employment.

Data benchmarked against data reported from IAPT demonstration sites (Clark et al., 2009) revealed more self-referrals and fewer unsuitable referrals, as well as similar proportions of referrals in terms of gender split and over 65s. The finding that self-referrals may result in greater treatment adherence was of importance since many IAPT services report high dropout rates at step 2 (Chan & Adams, 2014), and adherence is known to be associated with outcome and acceptability (Van Ballegooijen et al., 2014). Similarly, the *model's* higher completion rate of 52% (compared to 37% in Doncaster and 24% in Newham) was viewed favourably by commissioners.

The study found that the *model* delivered three times as many treatment episodes as IAPT demonstration site with the highest level of engagement, with self-referrals being more likely than GP originated referrals to engage and complete the entire programme, and seemingly not impacting on the number of unsuitable referrals. This indicated that clients are likely to do better when accessing a treatment that they have chosen rather than one that has been imposed.

This finding was of interest amongst commissioners with access targets to meet and contrasted with earlier research that indicated that GP referred clients were likely to do better (Mataix-Cols, Cameron, Gega, Kenwright & Marks, 2006). Furthermore, changes in effect size on key IAPT outcome measures and recovery rates for those completing at least two treatment sessions were comparable with IAPT demonstration sites (Objective 2), showing that clinical outcomes achieved by the *model* were comparable to those reported in other low-intensity services (Clark et al., 2009), though clients accessing IAPT demonstration sites had slightly higher symptom severity.

BME clients were just as likely to start, finish, and complete BtB<sup>™</sup> via the *model*. Given BME access to IAPT services is a key issue that many have sought to address (Beck, Naz, Brooks & Jankowska, 2019), this was a key finding; demonstrating the utility of the *model* and its role in serving population need (Objectives 1 and 3). Whilst BME referrals were slightly under-represented compared to the population in Manchester, with higher levels of baseline distress, no difference was found regarding access and outcomes.

Client satisfaction rates (Objective 3) were high in clients accessing the *model* (measured by the PEQ), however the views of those who disengaged were not sought; this being an area previously highlighted as necessitating further research (Kaltenthaler et al., 2008), and a limitation of the study. Other limitations included an under-representation of older clients aged over 65, incomplete data collection for employment outcomes, and lack of follow-up data. **Cavanagh et al. (2011b)** is a practice-based study without a control, which is problematic in terms of the internal validity of the study as outcomes could be explained by other variables or may have arisen due to methodological errors (Patino & Ferreira, 2018). This is mitigated by there being evidence of low rates of spontaneous remission in those seeking help with chronic anxiety and mild-moderate depression (Clark et al., 2009; Posternak & Miller, 2001).

Another limitation is that of there being incomplete data; managed by using Last Observation Carried Forward (LOCF) methods. However, such analyses have been criticised for their accuracy (Lachin, 2016), though are believed to represent a conservative approach to responding to the issue of missing data. Data completion issues arose due to eTCs and volunteers failing to be stringent regarding data collection; addressed via the introduction of eTC case management supervision.

Though there are these limitations, **Cavanagh et al. (2011b)** has several strengths. Firstly, it has a large sample size (n=510), rendering the research more efficient with greater reliability of results (Faber & Fonseca, 2014), demonstrating the effectiveness of the *model* (Objective 2). Furthermore, it uses objective data as a strength (Guetterman, 2019), enabling conclusions to be drawn about the sample in terms of intake variables and outcomes. This study provided unique insight into this under-researched area of eTherapy.

In summary, **Cavanagh et al. (2011b)** demonstrated, for the first time, that the *model* provided a statistically and clinically effective low-intensity IAPT service, achieving positive outcomes across a wide range of clients including those from BME communities, accessing the programme, BtB<sup>TM</sup>.

# Paper three: Lidbetter, N., & Bunnell, D. (2013). Self Help Services: helping people to help themselves. *Mental Health and Social Inclusion*, 17(2), 76-81.

This paper provides a reflexive, qualitative, subjective overview of the charity, Self Help Services, written by the thesis' author with input from the charity's then Communications Officer. Peer supported and user-led services in the treatment of anxiety and depression were scarce at the time and, societally, it was not accepted that those with lived experience of mental health conditions could develop, lead, and deliver primary mental health services, particularly NHS-funded services. The author's unique contribution includes establishing and founding a charity that developed and delivered primary care mental health services commissioned by the NHS for the treatment of anxiety and depression (Gamsu, 2011), and identifying the need for and developing, the *model*.

**Lidbetter and Bunnell (2013)** provides a detailed overview of Self Help Services, its ethos, and services (which included peer-led, drop-in groups, courses, 1:1 psychological therapy services) and eTherapy services, delivered through the *model* (Objective 1). The paper details the development and history of the charity from its inception, whilst giving tips for others interested in developing services using a similar approach, concluding with providing insight into the charity's future potential service provision.

The paper reports some of the learning and challenges experienced in the charity's history, including detailing how the expansion and securing of new services by competitive tender, which entailed the transfer of staff to Self Help Services, had been at odds with the 'grow your own' culture of the organisation and led to roles being filled by staff without lived experience of a mental health difficulty, which had implications for the overall dynamics of the charity, given its user-led ethos.

The paper includes two case studies: one detailing the journey of the charity's then Performance and Informatics manager from an attendee and service user of the charity's dropin support groups, to becoming an eTC and then Manager of the overall charity's eTherapy services; the other provides an overview of the experience of an eTherapy client who accessed BtB<sup>TM</sup> via the venue *model*. This case study details the client's satisfaction in terms of the *model's* simplicity and accessibility, highlighting the practical benefit of eTherapy including enabling her to secure employment (Objective 3).

The limitations of **Lidbetter and Bunnell (2013)** are that it is written by the founder of the charity and, therefore, may have been subject to researcher personal bias, including influencing the selection of case studies (Anderson, 2010). Furthermore, there are a small

number of case studies; more case studies would have provided further insight into client experience and would have strengthened the study (Yin, 2009).

Social desirability bias may have taken place, including under-reporting of socially undesirable attitudes and over-reporting of desirable attitudes by case study participants. Paulhus (1984) proposes that social desirability has two components: impression management, where people purposively strive to fit in or please the audience concerned; and self-deception, often unconscious and based on the motivation to maintain a positive self-concept. Both can lead to inaccurate self-reports and erroneous conclusions being made (Latkin, Edwards, Davey-Rothwell & Tobin, 2017).

The strength of **Lidbetter and Bunnell (2013)** include that data collected was based on human experience and, therefore, provides a rich, detailed, subjective perspective of the *model*, supporting transferability to other settings (Anderson, 2010). Furthermore, it provides vital insight and information on challenges encountered within the charity; offering learning that is of value to other organisations wishing to initiate similar services, including the development of a novel model of eTherapy service delivery (Objective 1).

## Paper 4: Elison, S., Ward, J., Davies, G., Lidbetter, N., Hulme, D., & Dagley, M. (2014). An outcomes study of eTherapy for dual diagnosis using Breaking Free Online. *Advances in Dual Diagnosis*, 7(2), 52-62.

This pragmatic, quantitative service evaluation where data (baseline and post-treatment clinical outcomes) was analysed, evaluated clinical outcomes (Objective 2) for the first time in clients experiencing dual diagnosis issues (substance misuse and mental health difficulties, in this case, anxiety and/or depression), accessing BFO via either the venue or remote version of the *model*, or a combination of each (Objective 1).

The study was undertaken to demonstrate the effectiveness of the *model* in supporting those with dual diagnosis issues to NHS commissioners, and to augment the small dual diagnosis service delivered at the time, which consisted a community-based, structured dual diagnosis course.

A range of clinical outcome measures were taken: WSAS, GAD-7, PHQ-9, alcohol use via the Alcohol Use Disorders Identification Test - AUDIT (Saunders et al., 1993) -, and drug use via the Drug Use Disorders Identification Test - DUDIT (Berman et al., 2004). The IAPT Phobia Scale (IAPT, 2011), a three-item scale containing single item measures for specific phobia, panic disorder, and social anxiety disorder was also used. Collectively, these measures allowed for analysis of client progress in terms of both substance misuse and anxiety and/or depression.

A total of 74 clients were referred to the service, with follow-up data available for 47. Having determined that the data did not follow a normal distribution using Shapiro-Wilk tests, non-parametric Mann-Whitney U-tests revealed differences between those clients for whom only baseline data was available and those with follow-up data; the former presented with more severe depression and impaired social functioning.

Of clients with follow-up data, 62% were male, 96% were white-British, 29% had not received treatment previously for their presenting problem, and 33% had previously been in treatment for substance use. Alcohol was the main substance of dependence for 78% of clients, 16% were equally split between being dependent on either cannabis, cocaine sulphate, crack cocaine, or heroin, and 6% equally split between being dependent on amphetamines, buprenorphine, or synthetic cannabis.

Periods of engagement varied considerably and was dependent on client perception of need, ranging from 15-154 days, with mean length being 54 days; similar to that taken by clients accessing other programmes typically delivered over a six-week period.

Irrespective of time taken to access the *model*, at discharge, clients completed the same psychometric measures. Statistical regression analysis found the number of days between baseline and discharge assessments did not act as a predictor of change in outcome scores.

Analysis of outcome measures, via the Wilcoxon-signed rank test for those with follow-up data, found statistically significant changes including decreases in scores for social impairment, depression, anxiety (all p <0.0001), drug use (p= 0.002), and alcohol dependence (p=0.028), as well as a reduction in scores for the one-item measure for social anxiety (p=0.027). No statistically significant changes were found in the phobia item measure in respect of panic disorder and specific phobia.

Furthermore, large effect sizes were found for reductions in social impairment (r=0.68), depression (r=0.59), and anxiety (r=0.63); medium effect sizes for reductions in alcohol use (0.45), drug use, and social anxiety (r=0.32).

Clinically significant changes were found in clients with follow-up data for those reaching clinical cut-offs for mild, moderate, and severe depression and anxiety; demonstrating that the overall clients meeting clinical cut-offs, in terms of symptom severity, reduced post-treatment. Specifically, clients reaching the clinical cut-off for depression at baseline was 89% and 89% for anxiety, with a total of 74% having moderate to severe depression and 72% moderate to severe anxiety. Post-treatment, 70% remained above the clinical cut-off for depression and 72% above the clinical cut-off for anxiety, with 49% reporting moderate to severe depression and 34% reporting moderate to severe anxiety. Clients that did not reach clinical cut-off for depression increased from 5-14% pre-post treatment assessment, and from 4-14% for anxiety.

Elison et al. (2014) demonstrated that the *model* could, when delivering BFO, generate statistically significant reductions in social impairment, alcohol and drug use, and social anxiety with reductions in depression, anxiety, and social impairment being particularly

robust, as demonstrated through large effect sizes and reductions in alcohol and drug use relatively robust, demonstrated through medium effect sizes. Furthermore, **Elison et al.** (2014) demonstrated that the *model*, when used to deliver BFO, could achieve clinically significant improvements in clients with dual diagnosis presentations in respect of reductions in anxiety and depression symptom severity.

The limitations of Elison et al. (2014) include its small sample size (47) and attrition rate; both factors which may impact on the reliability of the statistical outcomes reported (Faber & Fonesca, 2014). Quantitative research is accepted as a paradigm that supports researchers in demonstrating the generalisability and reliability of an intervention (Henn, Weinstein & Foard, 2006), however this is dependent on there being a sufficient sample size. In eTherapy research, small sample sizes are common, and it is recommended that research is conducted with larger sample sizes (Andersson & Cuijpers, 2009; Cuijpers et al., 2009). As the study was undertaken in a real-world setting, however, sample size was determined solely by the number of clients opting to access the service. In terms of attrition, the rate was reported as being 36%; in keeping with eTherapy literature (Eysenbach, 2005), but more favourable than rates reported in other studies (Chistensen et al., 2004a, b; Favolden et al., 2005). This is particularly relevant since this study was not subject to the same environment as that seen in RCTs; being conducted innovatively in a real-world setting. Furthermore, it is likely that the drop-out rate may have been reflective of the level of severity and impairment experienced by those clients accessing the service; it has been reported that higher social impairment and low functioning may affect retention and engagement (Barrett et al., 2008).

A strength of **Elison et al. (2014)** is that it reports on engagement length; seldom addressed in eTherapy research as is similarly the case with studies lacking information on reasons for drop-out (van Ballegooijen, 2014). In giving this information, helpful insight is provided into how engagement length may impact on treatment outcomes. Previously, BFO had only been positively piloted as a programme to support those with substance misuse issues (Elison et al., 2013). However, **Elison et al. (2014)** found both versions of the *model* (venue and remote), or a combination of each, could achieve clinically significant improvements in clients with dual diagnosis presentations in respect of reductions in anxiety and depression symptom severity (Objective 2). Therefore, this could be used to support clients with dual diagnosis presentations (Objective 1), though it is recognised that the majority of clients had alcohol dependency issues (78%) compared to drug dependency issues (22%). This finding extended the reach of the venue, remote, or a combination of both, *models* to an additional client group and, in doing so, made a unique contribution to eTherapy literature.

Paper 5: Elison, S., Ward, J., Williams, C., Espie, C., Davies, G., Dugdale, S., ... Smith, K. (2017). Feasibility of a UK community-based, eTherapy mental health service in Greater Manchester: repeated-measures and between-groups study of 'Living Life to the Full Interactive,' 'Sleepio<sup>TM</sup>' and 'Breaking Free Online' at 'Self Help Services.' *BMJ Open*, 7(7).

This feasibility study ascertained the viability and outcomes of the venue, remote, or combination *model* when used to deliver three different eTherapy programmes: LLTTF<sup>TM</sup>, BFO and Sleepio<sup>TM</sup>; an insomnia for clients experiencing a range of mental health issues including depression anxiety and/or stress, sleep disruption, or problems with alcohol or drugs. The study was a service evaluation that examined the *model's* potential to provide a helpful addition to traditional IAPT service offerings; building on previous effectiveness studies undertaken (**Cavanagh et al., 2011b; Elison et al., 2014**).

Of 1,786 clients completing a baseline assessment, data was collected from 1,068 clients (59.8%) who had completed one of three eTherapy programmes between 2011 and 2015 accessed via venue, remote, or a combination of the *models*.

Clients were assessed by an eTC on access (via phone or face-to-face), completing a range of standardised psychometric assessments - PHQ-9, GAD-7 and WSAS - to identify main area of difficulty, and were triaged to one of three groups; this allowed them access to the most appropriate programme available.

The breakdown of clients per programme was: 81% - LLTTF<sup>™</sup> interactive; 8% Sleepio<sup>™</sup>; and 11% BFO.

Data analysis undertaken using the Shapiro-Wilks found pre- and post-data for PHQ-9, GAD-7 and WSAS was non-normally distributed (all = p<0.05).

Kruskal-Wallis analyses of variance (ANOVA) was deployed to examine baseline differences between the three eTherapy groups in respect of IAPT MDS. Analysis of Covariance (ANCOVA) was used to examine if programme assignment was predictive of the extent of the change in functioning of clients when age and gender were controlled for.

Within-group, repeated-measures Wilcoxon signed-rank ANOVA tests were utilised to examine pre- and post-changes for the same psychometric measures within each of the groups, with Pearson's effect sizes calculated from test statistics using ANOVA and ANCOVA tests, as well as sample size.

A total of 63.6% of the overall sample were female (compared to 40.2% in the group accessing BFO). The mean age of the sample was 37.38 years. Regression analysis of length of time of engagement in treatment found that, whilst engagement varied (4-288 days), it was not associated with the degree of change in symptom severity of depression, anxiety, or social functioning from baseline to post-treatment in LLTTF<sup>TM</sup> or Sleepio<sup>TM</sup>. However, for clients accessing BFO, the number of days of engagement was associated with the degree of change of scores for depression, anxiety, and social functioning from baseline to post-treatment the second s

Comparison of the three groups, using Kruskal-Wallis ANOVAs on baseline psychometric scores, found some statistically significant differences: the BFO group had significantly higher baseline scores for depression and anxiety than Sleepio<sup>TM</sup> and LLTTF<sup>TM</sup> groups. When comparing degrees of psychometric score pre-post change between the three groups whilst controlling for age and gender, ANCOVA revealed no statistically significant differences. Furthermore, comparison of baseline to post-treatment, using Wilcoxon signed-rank tests, found statistically significant reductions in all psychometric measure scores for anxiety, depression, and social functioning for all three groups (p for all groups = <0.0001), with moderate to large effect sizes.

**Elison et al.** (2017) also reports on the percentages of clients reaching clinical thresholds for mild, moderate, and severe depression and anxiety at baseline and at post-treatment, via Chi-squared analysis, finding that in all the groups, the percentages of clients reaching thresholds for clinically significant depression and anxiety after treatment from baseline significantly reduced. Moreover, the percentages of clients in the moderate and severe categories post-treatment reduced but increased in the minimal and mild categories.

Statistically significant in-group reductions in outcome scores for anxiety, depression, and social functioning were found for all programmes accessed through the *model*. The degree of symptom reduction and of reduction in scores were also found to be comparable across all programmes, with group assignment not being predictive of the degree of change in respect of depression, anxiety, or social functioning between baseline and post-treatment assessment. Although equivalent outcomes were reported, there were statistically significant differences between the groups at baseline assessment in terms of symptom severity with clients that accessed BFO having higher baseline scores for anxiety and depression, and higher scores for anxiety post-treatment. This can be explained by those accessing BFO having more complex issues as it is known that mental health can be impaired by substance use (Buckley, 2006).

The higher rate of anxiety seen post-treatment in clients accessing BFO may be explained by the phenomenon seen in clients in the early stage of substance misuse recovery, where a temporary exacerbation of mental health symptoms can arise (Lubelczyk, 2015; Powell & Taylor, 1992) because of the anti-depressant effects of any substances that were previously taken, no longer being present; this being particularly so with opiates (De Arcos, 2008).

The study's limitations are that there is no data available in respect of those clients who disengaged following the baseline assessment, or why clients did not require treatment. Furthermore, the varying sample size of the groups is problematic as ideally the sample size would have been similar for all groups, thereby allowing optimal performance of equivalence testing between groups (Rusticus & Lovato, 2014). The lack of follow-up data and absence of a control group are also limitations, though recent research has shown that the evaluation of eTherapy as an intervention through RCTs may be subject to methodological limitations because of the personalisation of the approach, which may give rise to enhanced clinical effectiveness (Krebs, Prochaska & Rossi, 2010; Lustria et al., 2013; Noar, Benac & Harris, 2007).

A strength of **Elison et al. (2017)** is its large sample size and use of the IAPT MDS; allowing for future benchmarking against IAPT services. It also reports both statistically and clinically significant findings with reductions in all three groups in the percentages of clients reaching threshold scores for clinically significant anxiety and depression, post-treatment. This is a strength as published research often focusses on the statistical significance whilst failing to determine the clinical importance of results (Nordahl-Hansen, Øien, Volkmar, Shic & Cicchetti, 2018). The study also sheds light on attrition, detailing that 38.5% of clients had disengaged/dropped out following baseline assessment, 34.9% had been discharged following baseline assessment because treatment was not required, and 35% had been referred to another more appropriate service, for example, a high-intensity IAPT service.

**Elison et al. (2017)** makes a unique contribution to eTherapy literature by continuing to build the evidence base for the clinical effectiveness of the remote, venue, or combination of both *models* when used to support those with anxiety and depression (Objective 2) accessing a range of eTherapy programmes.

Paper 6: Luik, A. I., Bostock, S., Chisnall, L., Kyle, S. D., Lidbetter, N., Baldwin, N., & Espie, C. A. (2017). Treating depression and anxiety with digital cognitive behavioural therapy for insomnia: a real world NHS evaluation using standardized outcome measures. *Behavioural and Cognitive Psychotherapy*, 45(1), 91-96

This study evaluated changes in depression and anxiety in clients with poor sleep, in combination with having symptoms of depression and anxiety. Clients accessed Sleepio<sup>TM</sup> (a six-session programme) via the remote *model* (Objective 1) between March 2014 and May 2015, with the initial assessment offered either face-to-face or by phone.

The study was a pragmatic, real-world service evaluation of eTherapy within an IAPT programme, where outcomes for clients that had accessed the service were evaluated post-treatment.

Six telephone support calls of between 20-30 minutes long were provided, during which, clients were explained techniques (and suggestions on implementation) with the eTC also determining treatment progress. No clinical input was provided, and clients did not access any other IAPT intervention at the time.

Levels of anxiety and depression were independently measured during each of the calls via GAD-7 and PHQ-9. Insomnia was evaluated at baseline and during the final assessment via the Insomnia Severity Index - ISI (Bastien, Vallières, & Morin, 2001). Clients that did not improve after accessing Sleepio<sup>™</sup> through the *model* were referred to other support services.

Additional client data (gender, age, use of sleep medication) was collected and obtained within Sleepio<sup>TM</sup> itself.

A total of 120 clients received an access code for Sleepio<sup>™</sup>; of these 98, accessed the service and baseline and post-treatment assessment data was taken. Paired and unpaired T-tests and linear mixed models were used to analyse the data.

The average age of clients was 44.4 years, with 66% female and 14% reported to have taken sleep medication. At baseline, average scores for PHQ-9 were 11, 8.6 for GAD-7, and 18.5 on the ISI. 72% of clients had moderate to severe depressive symptoms (PHQ-9 >9) or moderate to severe anxiety symptoms (GAD-7 >7), and 89% experienced clinical insomnia (ISI >14). Of the 98 clients, 72 (73%) completed treatment, 15 clients completed four to six sessions and 11 dropped out before session four.

Mean PHQ-9 reduced from 10.1 to 4.4 (p<0.001), and mean GAD-7 score reduced from 7.7 to 3.7 (p<0.001) in clients who completed post-treatment measures and the end of treatment. To account for any missing data, a linear mixed model (n=98, unstructured) was used; the difference before and after treatment remained statistically significant (PHQ-9 t (74) =13.7, p < 0.001; GAD-7 t (84) = 9.4 p < 0.001) with gender, age, or use of sleep medication not reducing these associations.

Baseline depression and anxiety rates were higher in the non-completer group than completers (i.e., those completing all six sessions) for both depressive and anxiety symptoms.

Of the 71 clients who scored above PHQ-9 and GAD-7 threshold for caseness at baseline, 68% moved to recovery when the LOCF was used. Analysis of only complete cases found the recovery rate to be 58% and 65% of the total population scoring below case-level after the treatment period. Additionally, 59% of clients achieved IAPT reliable recovery; this figure decreased to 52% when complete cases were analysed.

Insomnia symptoms significantly decreased (n=71) from an average of 18.3 to 6.6 (p<0.001) as measured by the ISI. Furthermore, changes in insomnia symptoms were statistically significantly related to changes seen on the GAD-7 (r=0.41, p<0.001) and the PHQ-9 (r=0.32, p = 0.006).

Those completing Sleepio<sup>TM</sup> achieved an IAPT recovery rate of 68%, with a reliable recovery rate of 59%; both being above the IAPT target of  $\geq$  50% and 43% respectively (NHS Digital, 2020).

Clients who failed to complete treatment had higher baseline levels for anxiety and depression, which could be interpreted that eTherapy is unsuitable for clients with elevated levels of symptom severity, though this is contradicted by Karyotaki et al., 2018 who report clients with severe depression benefit from eTherapy.

The limitation of **Luik et al. (2017)** is its lack of a control arm, though this was not possible due to the research taking place in a real-world setting.

The strengths of the study include that it provided a satisfactory level of detail on Sleepio<sup>TM</sup>, as well as giving information on the treatment pathway and care touchpoints between client and the eTC (Objective 1); assisting with replication, which is something considered to be rare in real-world research (Robson & McCartan, 2016). Furthermore, **Luik et al. (2017)** uses IAPT measures, enabling recovery rates to be benchmarked against national IAPT data (Objective 2). The study also had high completion rates (73%) compared to the average seen in IAPT at the time of 48% (NHS Digital, 2020), supporting the viability of the *model* for clients experiencing poor sleep as well as depression and anxiety.

Luik et al. (2017) contributes to the eTherapy literature by demonstrating that Sleepio<sup>TM</sup> could be successfully delivered via the remote *model*, with treatment completion rates above the IAPT average and recovery rates higher than IAPT self-help treatment that directly target anxiety and depression. Furthermore, Luik et al. (2017) demonstrates that symptoms of

anxiety and depression in clients presenting with these conditions, as well as having problems with sleep, could be alleviated when Sleepio<sup>TM</sup> is accessed via the *model* (this being consistent with the findings of Christensen et al., 2016); demonstrating its effectiveness and versatility (Objectives 1 and 2). Substantial decreases in insomnia symptoms were also reported and correlations found between changes in insomnia and changes in depression and anxiety; suggesting that better sleep might act as a mediator of outcome for depression and anxiety, however further research is required to definitively confirm any associations.

# Paper 7: Gellatly, J., Chisnall, L., Seccombe, N., Ragan, K., Lidbetter, N., & Cavanagh, K. (2018). @ Home eTherapy service for people with common mental health problems: an evaluation. *Behavioural and Cognitive Psychotherapy*, 46(1), 115-120.

This study is a pragmatic, retrospective service evaluation that evaluated the implementation of the remote *model* (referred to in the paper as an @ home model) (Objective 1). Furthermore, clinical outcomes achieved were benchmarked against national IAPT service data (Objective 2).

Clients accessing the *model* consented for their anonymised data to be used in routine evaluations, including for this study, and were provided with an assessment as well as between six to 12 20-minute telephone support calls (depending on eTherapy programme accessed). Support calls were made by an eTC who was not formally qualified and had taken part in a two-week induction programme, supported by monthly supervision.

Five programmes were available - Sleepio<sup>TM</sup>, BFO, LLTTF<sup>TM</sup>, Silvercloud Health (Space from Depression), and BtB<sup>TM</sup> - though not all were available across all the sites where the *model* operated and, therefore, analysis took account of all programmes rather than individual programme outcomes. Clients commenced the most appropriate package available;

determined by their pre-treatment outcome measures and following a collaborative discussion with the eTC.

At every session, PHQ-9, GAD-7 and WSAS were taken. Clients above the clinical cut-off for PHQ-9 and GAD-7 at assessment met caseness. First and last scores were used in the calculation for recovery, reliable improvement, and reliable recovery.

Data were analysed using SPSS (v22), with descriptive statistics presented and inferential statistics used to assess clinical and statistical significance of change on IAPT key measures, using data from the last attended appointment used.

A total of 2,054 clients were referred to the service over 30 months; 66% attended an assessment with 53.4% attending at least two treatment appointments; this being equivalent to completing an IAPT course of treatment (Health and Social Care Information Centre [HSCIC], 2016).

Statistically significant improvements for completers across all programmes combined were found on all outcome measures including PHQ-9, GAD-7, and WSAS, with large pre- to post-treatment effect sizes for changes in symptom severity of depression (Cohen's d= 0.98) and anxiety (d=1.07), with a medium effect size found for functioning (d = 0.53).

At intake, 91.4% of clients met caseness levels for anxiety and depression or both and, of these clients, 61.6% recovered at their final appointment, 56.7% met the criteria for reliable recovery, and 66.2% met criteria for reliable improvement (from start to end of treatment), with only 4.4% having reliably deteriorated.

Most completers were female (57%), aged 45 years or under (79.9%), with more than half being under 36 years of age (57%) and from a white ethnic group (91.2%). A total of 60.6% accessed LLTTF<sup>TM</sup>, 19.1% Silvercloud Health, 4.4% BtB<sup>TM</sup>, 3.6% Sleepio<sup>TM</sup>, and 0.1% BFO.

106

An independent samples T-test was conducted to compare moving to recovery by clients from caseness, with appointment attendance. A statistically significant difference was found between number of sessions attended for those clients that moved towards recovery (mean = 6.04 sessions, SD = 1.8) and those that still met caseness (mean = 4.62 sessions, SD = 2.0), with a large magnitude in the differences of the means (mean difference = 1.42, 95% CI: 1.11 to 1.73), (d=0.937).

Analysed data was also used to benchmark outcomes achieved through the remote *model* against IAPT data (HSCIC, 2016), with 96.3% of clients that finished treatment waiting less than six weeks to enter the service compared to 91.4% within IAPT services nationally (HSCIC, 2016). When comparing caseness at intake against national IAPT, similar numbers were found (91.4% vs 88.6%). Similarly, an equivalent number of referrals completed a programme through the remote *model* (53.4%), when compared to national IAPT service data for clients referred to guided eTherapy (52%). However, on average, almost twice the number of eTherapy appointments were attended (5.6) by clients accessing the remote *model* in this study, compared with that reported in national IAPT services (2.9). This was a positive finding as drop-out rates are known to be an issue in low intensity IAPT services (Chan & Adams, 2014).

Recovery and reliable recovery rates in the study (61.6% and 56.7%) surpassed those in national IAPT services (52.2% and 42.8%), whilst reliable improvement rates were broadly comparable (66.2% and 60.8%). Furthermore, effect sizes determined for PHQ-9 and GAD-7 were found to be equivalent to high performance benchmarks for national services, as detailed in the work undertaken by Delgadillo et al. (2014) where the evaluation of evidence-based intervention measurements in routine practice were explored.

The limitations of **Gellatly et al. (2018)** are that reliable improvement and reliable recovery figures are compared against data from all IAPT referrals, as eTherapy data was not available

from NHS Digital, 2015. Furthermore, the absence of a control (due to the study being a service evaluation) and the lack of follow-up data or information on why clients dropped out of treatment, are additional limitations.

The strength of **Gellatly et al. (2018)** is its large study sample size and its comparison of outcomes for all programmes combined, which gave rise to being able to demonstrate statistically and clinically significant outcomes, which, in turn, demonstrated for the first time the clinical effectiveness of the remote *model*; making a unique contribution to the eTherapy literature. Furthermore, **Gellatly et al. (2018)** provided an appropriate level of detail about the eTC role; emphasising the role of peer support. The study also benchmarked outcomes against national IAPT service data and, when possible, this included comparing data with data from other low intensity eTherapy IAPT services, making for an almost like-for-like comparison.

#### 4.3 Critique of the presented publications by thesis objective

#### **<u>Objective 1:</u>** Examine the development of the *model*.

Based on previous experience of developing peer-supported, user-led primary care mental health services, the author led the development and delivery of the *model*. External, credible experts, renowned in the field of eTherapy research, were brought on board with access to research resources and expertise in journal article submissions, and which allowed for an independent evaluation of the *model* to be undertaken. This was necessary as eTherapy was still relatively new and, where available, services operated from settings such as GP surgeries, run by clinicians.

Lidbetter and Bunnell (2013) and Cavanagh et al. (2011a) provide qualitative insight into the venue and remote *model* and the 'grow your own' culture of the charity where volunteers
with lived experience of mental ill-health are supported into paid employment to become eTCs. In contrast to research that merely proposed a potential delivery model (De Vares, 2007), or minimally investigated the topic in respect of integrating lived experience into the design of eTherapy services (Ray et al., 2017), or simply focussed on adding lived experience to an existing clinical support offer (Tomasino et al., 2017), **Lidbetter and Bunnell (2013)** and **Cavanagh et al. (2011a)** describe a fully functioning, peer-supported eTherapy *model*.

Furthermore, the portfolio of papers describes the *model* when delivered by a user-led, TSO based in the community; again, addressing a gap in the literature concerning eTherapy delivered in real-world settings; an area requiring further research (Glasgow, Phillips & Sanchez, 2014; Mohr, Cheung, Schueller, Hendricks & Duan, 2013a; Pham, Wiljer & Cafazzo, 2016; So et al., 2013).

Elison et al. (2014, 2017) and Luik et al. (2017) detail the expansion of the venue and remote *model* (as well as a combination of each) to being able to support different client groups, including those affected by difficulties such as substance misuse issues (Elison et al., 2014, 2017) and insomnia (Luik et al., 2017), in addition to anxiety and depression.

A strength of **Lidbetter and Bunnell (2013)** is that it addresses scalability in providing an overview of both venue and remote *model*, whilst detailing operational and strategic challenges, making a unique contribution to eTherapy literature. A limitation, however, is that it lacks the detail of **Cavanagh et al. (2011a)**, which provides information regarding client-eTC care-pathway touch points. Both studies fail to describe the amount of human support provided with either version of the *model*, though **Cavanagh et al. (2011b)** addresses this by providing information on the venue *model*; giving duration of the initial screening appointment and information on the roles of eTCs and volunteers - explaining that support can range from technical assistance through to supporting clients to get the best out of programmes accessed.

Elison et al. (2014) provides information on the duration and structure of follow-up support, overall length of engagement - known to be important to outcomes (Baumeister et al., 2014) - as well as detailing the method of support (e.g., face-to-face or telephone). Luik et al. (2017) also gives an overview of the role of the eTC, the length of the support calls and the nature of the content covered during support interactions; addressing a gap in knowledge in eTherapy literature that is still regarded as under-investigated (Lattie et al., 2019). To enable replication, a comprehensive description of the *model* and of eTherapy programmes would be required. Across the portfolio of works, whilst the *model* is described at overview level, intricate, specific, and particularised detail consistent with a Service Operating Procedure (SOP), for example, was not included. Lack of fine detail regarding eTherapy delivery models and implementation continues to be a common feature of eTherapy research (Hollis et al., 2017), though clearly is a matter of operational significance given it is known that the amount of human support varies according to the type of clinical issue being addressed (Newman et al., 2011), and that implementation of eTherapy continues to be problematic (Thew, 2020).

The limitation of the methods used are that the papers describe the *model* instead of going into a deeper analysis, and lack depth. The overall purpose, however, was to provide an overview of the *model* instead of giving this level of detail that an observational study, for example, would have provided (Pope & Mays, 2006).

#### **Objective 2:** Review and examine the clinical effectiveness of the *model*.

The efficacy of eTherapy in the treatment of depression has been demonstrated in a range of RCTs, yet knowledge of how eTherapy works in routine care is scarce (Hedman et al., 2014b). Five papers in the portfolio of works demonstrated the effectiveness of the *model* in a real-world setting when used to deliver a range of eTherapy programmes for those with anxiety and depression, via quantitative retrospective service evaluation (**Cavanagh et al..2011b**; **Elison et al., 2014, 2017; Luik et al., 2017; Gellatly et al., 2018**). To ensure credibility in

the evaluations, a multi-disciplinary team with a range of academic expertise were brought together to ensure that the evaluations and their findings were as robust as possible. Experts, including statisticians, were brought in for robust statistics support, for example, with credibility in the field of quantitative eTherapy research.

Such quantitative evaluations allowed for the exploration of the implementation of eTherapy and assisted with result interpretation (Oakley, Strange, Bonnell, Allen & Stephenson, 2006), but did not generate generalisable data (Moule et al., 2016). In contrast, the effectiveness of eTherapy in the treatment of anxiety and depression has been demonstrated via meta-analyses of RCTs (Andrews, Cuijpers, Craske, McEvoy & Titov, 2010) and systematic reviews, which report the intervention as being an established treatment for depression, SAD, and panic disorder (Hedman, Ljótsson & Lindefors, 2012). However, it was not possible to conduct studies of these types due to being constrained to undertaking service evaluations; this being a requirement of commissioners.

Furthermore, the efficacy of the *model* was not examined because it was not possible to have a control group (i.e., clients that did not access eTherapy via the *model*) as it would have been unethical to refuse some clients treatment as all had contacted the charity for eTherapy support with anxiety and/or depression. It could, therefore, be argued that some of the observed improvement may have occurred without any intervention having taken place, e.g., natural recovery. The addition of a control group to the studies would have increased the internal validity of the research design, however this was not possible because of funding constraints and due to the studies being service evaluations. Thus, the lack of a control group represents a limitation, however as the studies were carried out in a real-world setting - something that has been called for within the field of eTherapy research (Karyotaki et al., 2017) -, this is a strength as effectiveness research is considered more relevant for providers and commissioners, and to everyday practice (Treweek & Zwarenstein, 2009).

111

Meta-analyses examine RCTs that typically assess the efficacy and effectiveness of eTherapy by comparing outcomes against a control group or comparison group (for example, face-toface CBT), via calculation of the standardised mean difference (SMD). SMD was calculated in **Cavanagh et al. (2011b)**, **Elison et al. (2014, 2017)**, **Gellatly et al. (2018)** and **Luik et al.** (2017); therefore, a robust and validated method was used to determine effectiveness. Furthermore, **Cavanagh et al. (2011b)**, **Elison et al. (2017)** and **Gellatly et al. (2018)** all had large sample sizes with sufficient power, allowing for meaningful detection of difference and the reporting of statistically significant results, which because of their large sample sizes, are likely to result in them being less exposed to bias (Biau, Kernéis & Porcher, 2008).

**Cavanagh et al. (2011b)** reported large effect sizes for clients completing at least two sessions in respect of depression (d = 0.8) and anxiety (d=0.9). **Gellatly et al. (2018)** reported large effect sizes when comparing pre- and post-treatment data for changes in level of severity of depression using the PHQ-9 (Cohen's d = 0.98) and anxiety using the GAD-7 (d=1.07), where anything over 0.8 is considered a large effect size (Cohen, 2013). **Elison et al. (2017**) reported moderate to large effect sizes for social functioning, anxiety, and depression.

Outcome measures used in the studies were purposefully selected so that the same measures used in IAPT services were used in this quantitative research on the *model* to measure changes in anxiety and depression symptom severity (measured by GAD-7 and PHQ-9) and functioning (measured by the WSAS), and facilitated initial comparison of the *model* with IAPT services (**Cavanagh et al., 2011b**). Following the subsequent commissioning of the *model* as an IAPT service, it then became a requirement of commissioners that the full IAPT MDS was used and hence why in **Luik et al. (2017)** and **Gellatly et al. (2018)** outcome data from the *model* could be directly benchmarked against IAPT service data.

Whilst an intervention may prove effective in terms of its ability to impact on symptom severity reduction (measured by clinical outcome tools such as the GAD-7 and PHQ-9), if

clients are dropping out of the service, this impacts on the effectiveness of the intervention. Attrition is a key issue in eTherapy service delivery and has been widely reported in the literature as needing further attention (Donkin, Christensen & Naismith, 2011; Eysenbach, 2005; Khadjesari et al., 2011; van Balleegooijen et al., 2014). It is thought to be associated with poorer outcomes for clients and is, therefore, important to address (Delgadillo et al., 2014). Attrition was an issue in **Cavanagh et al. (2011b)**, **Elison et al. (2014, 2017)**, **Luik et al. (2017)** and **Gellatly et al. (2018)**, where relatively high numbers of clients failed to complete eTherapy when accessed via the *model*. Whilst this could be a limitation and potentially a negative aspect of the *model*, it could be argued that clients discontinued accessing the service because they had resolved their difficulties and therefore no longer needed support (**Elison et al., 2014**).

The long-term effectiveness of the *model* has not been established yet (though work in this area is ongoing), and is a limitation of **Cavanagh et al. (2011b**), **Elison et al. (2017**) and **Gellatly et al. (2018**) in that long-term, follow-up monitoring did not take place because outcome measures were only taken pre- and post-clients accessing the *model*, in contrast to other studies where measures were taken at a greater length of time post-treatment (Andrews et al., 2018).

**Gellatly et al. (2018)** found better recovery outcomes seemed to be associated with clients that attended more treatment appointments, and **Elison et al. (2017)** found length of time of engagement was associated with the degree of change of symptom severity of depression for clients accessing BFO via the *model*; suggesting that engagement is an important consideration and has a role in terms of effectiveness (Donkin et al., 2011).

## **Objective 3:** Evaluate the acceptability of the model.

Three studies evaluated the acceptability using case studies (Cavanagh et al., 2011a; Lidbetter & Bunnell, 2013) and via the IAPT PEQ (Cavanagh et al., 2011b); a client satisfaction survey focusing on length of time clients wait for an appointment, overall client experience of using the service, and client satisfaction with treatment type received; administered at baseline and treatment end. In **Cavanagh et al. (2011b**), 44.9% of clients completed pre- and post-PEQs, with 90% satisfied with their overall experience of the service and 91% satisfied with the treatment received. Using the PEQ measure ensured that client satisfaction was measured via a standardised and well-known tool widely used in IAPT services and, therefore, enabling benchmarking of satisfaction at a later stage if required. Additionally, **Cavanagh et al. (2011b**) reports a high completion rate compared to that seen in the IAPT demonstration sites. Given adherence is likely to be linked to acceptability (Van Ballegooijen et al., 2014), inclusion of this metric was a useful additional measure. Similarly, **Gellatly et al. (2018)** reports twice as many appointments attended compared to national IAPT reported rates; of interest, since engagement is likely to be linked with acceptability (Cavanagh et al., 2018).

The five case studies detailed in **Cavanagh et al. (2011a)** and **Lidbetter and Bunnell (2013)** provide additional helpful insight into client experience, as well as the then eTherapy manager's story of moving from client to service manager. Furthermore, given the heterogenous nature of the case studies, insight is gained into the model's acceptability by providing insight from a range of clients.

Other methods of assessing acceptability in eTherapy research include use of questionnaires (Burke et al., 2018), which do not permit in-depth analysis of patient experience, through to deeper qualitative methodologies such as focus group and interviews, as used by Holst et al. (2017). Due to the pragmatic nature of the studies, however, methods used to assess acceptability were necessarily restricted to those already used in service, e.g., case study testimonials and the PEQ.

Client acceptability in eTherapy is known to be an area requiring further attention (Rost et al., 2017; Wilhelm et al., 2020), along with attrition (Kaltenthaler et al., 2008; Richards & Richardson, 2012; van Ballegooijen et al., 2014; Waller & Gilbody, 2009), though across the body of works, attrition rates were not high. Further understanding of issues underpinning client acceptability could have been gained if the research had also focused on clients failing to complete the full course of eTherapy delivered through the *model*, and reasons for disengagement (Marks et al., 2007); this being a limitation of **Cavanagh et al. (2011b**), **Elison et al. (2014, 2017), Luik et al. (2017) and Gellatly et al. (2018)**. Within a service evaluation, however, this was not possible for pragmatic reasons.

#### 4.4 Section summary

In this section, the papers comprising the portfolio of works are presented, individually critiqued, and then examined by thesis objective.

Collectively, the body of works makes a unique contribution to eTherapy knowledge by detailing a peer-supported, non-clinical *model* of eTherapy, developed in a user-led organisation, delivered in real-world settings, that is effective and acceptable in the treatment of anxiety and depression in adults. This was specifically achieved through **Cavanagh et al.** (2011a, b) and Lidbetter and Bunnell (2013), where acceptability of the venue *model* was demonstrated in clients with anxiety and depression. Cavanagh et al. (2011b), Elison et al. (2014), Gellatly et al. (2018) and Luik et al. (2017) showed that the *model* was able to deliver statistically and clinically significant improvements in clients experiencing anxiety and depression, with Cavanagh et al. (2011b) demonstrating the pragmatic effectiveness of the venue *model* and Gellatly et al. (2018) the remote *model*. Furthermore, Elison et al. (2014) demonstrated the effectiveness of the *model* when used to deliver BFO for clients experiencing anxiety and depression as well as drug or alcohol dependency issues, and Luik

et al. (2017) demonstrated the effectiveness of the model when used to deliver Sleepio<sup>TM</sup> in clients experiencing poor sleep and anxiety and depression.

The next section will examine ethical considerations, methodology and limitations, the pragmatic paradigm, and the author's reflections on the journey taken in creating the body of works.

#### Section Five

In this section, ethical considerations, methodology and limitations, the pragmatic paradigm, and a personal reflection of the thesis, are provided.

#### 5.1 Ethical considerations

Whist undertaking this body of work, several ethical considerations were of relevance, including: the avoidance of harm whilst undertaking real-world research (non-maleficence); coercion of clients to participate; informed consent; having respect for autonomy; protecting and preserving confidentiality (including the safe storage of participant data); and commitment to doing good - beneficence (Beauchamp & Childress, 2008). All are covered by the ethical principles that health researchers use to protect clients from harm, and are distilled down to four rights of subjects considering participation in research, as follows (International Council of Nurses, 2003):

- The right not to be harmed
- The right of full disclosure
- The right of self-determination/to take part or to withdraw at any time
- The right of privacy, anonymity, and confidentiality

The author was mindful of possible ethical issues regarding involvement in the research, whilst at the same time having a commitment to leading the organisation and overseeing its research strategy. This could have potentially given rise to researcher or experimenter bias, which occurs when an individual conducting a research study intentionally or unintentionally influences the results, for example, by asking leading questions, or through poor research design (Galdas, 2017). Bias itself can be defined as:

"any influence that provides a distortion in the results of a study" (Polit & Beck, 2014)

A further consideration is that of clients agreeing to participate in, or respond to, research in a way that they believe would be desired by the researcher; all of which can undermine research findings (Perrier, Etchegary, Palarchio & Snelgrove, 2009). It is accepted that bias is hard to eliminate and exists in all research and across research designs (Smith & Noble, 2014). In respect of this body of works, it is recognised that the author brought her own experiences and beliefs to the work from the outset; specifically, that the *model* could achieve successful outcomes. As a result, researcher bias may have played a role in the selection of the case studies in Lidbetter and Bunnell (2013) and Cavanagh et al. (2011a), as it is known that case study research methodologically facilitates bias toward verification where the researcher's views are confirmed (Flyvbjerg, 2006). The effects of researcher bias, however, were mitigated by ensuring that clients were only contacted by staff not directly involved in the research, and that it was made clear that both positive and negative feedback was welcomed. Research bias was also mitigated by bringing in a team of external, respected academics to facilitate independent evaluation of the *model*. It is, however, recognised that there was a conflict-of-interest present (Yanos & Ziedonis, 2006), arising from the opposing agendas inherent to the roles of researcher and Chief Officer and Founder. In the latter role, the author's focus was on growing the organisation and securing commissions for peerdelivered services; fuelled by a fervent desire to promote the efficacy of user-led mental health services. As researcher, the author's goal was to undertake unbiased research to further understanding of the *model* and to test its efficacy; both of which required a neutral stance. Yanos and Ziedonis (2006) state:

"It has been stated that the field of psychological therapy would stagnate without the involvement of researchers who have direct clinical experience with the health conditions and service systems being studied" (p.249).

Furthermore, the dual role of counsellor-researcher role (whilst not an exact replica of the

relationship of the author-researcher relationship, has similarities), has been reported by Fleet, Burton, Reeves and DasGupta (2018) as being worth the struggle, even though conducting effective and ethical research can be hard.

A further consideration encountered concerned clients completing outcome measures; this aspect of the service explained to clients as being necessary to determine if clients believe that the intervention has been able to effect change, however organisationally, these scales were also used for funding and research purposes (Kewley & McBride, 2013). This issue was addressed in the studies that form the portfolio of works by ensuring that clients were informed, prior to accessing the *model*, that they were required to fill in several questionnaires before, during, and on completion of therapy, and that their questionnaire answers could be published anonymously in internal or external reports, including for the purposes of service evaluations. This was a critical issue to address as those involved in research that are also involved in delivering the service that is the subject of the research, must maintain a balance when working with clients; whilst the research protocol needs to be explained to the client, this cannot be at the expense of impacting on the session/interaction itself (Castonguay et al., 2010).

Respect for autonomy was achieved across the body of works by ensuring that the *model* was delivered in a collaborative, non-hierarchical manner without a 'them and us' relationship; something often seen in services where power is typically held with one partner, usually the healthcare professional (Seale, 2016). Additionally, as clients were supported by volunteers and eTCs (many of whom had themselves accessed the *model* and other services provided by Self Help Services), this helped address power imbalances, with clients being made aware from the outset (through service promotional materials and through disclosure), that those delivering the service had experience of living with a mental health difficulty.

At all times, clients' rights to be self-governing was respected (Bond & Dryden, 2012) by ensuring that clients were fully aware of the right not to be involved in service evaluations, including requesting their anonymised data be excluded from future data analysis. The right for clients to be self-governing is in synergy with the ethos of Self Help Services that was, and still is, that those experiencing anxiety and depression when given appropriate guidance and direction, are in the best position to be able to help themselves. This point was emphasised by the charity's strapline, which at the time was: *'helping people to help themselves.*' This was further demonstrated in the writing up of client case studies, where clients were given the opportunity to read and comment on the definitive version of the case study and request information be removed, as per ethical case study research (Widdowson, 2011).

Clients were also made aware of how to make a complaint about the service and to whom the service provider was accountable. In addition, clients were advised as to what types of information would be disclosed about them, with all information provided in advance of clients accessing the *model*.

It was essential clients were able to make an informed choice regarding their participation in the research. Some may have operated from a belief system where they felt that, if they refused to participate, this may, in some form, negatively affect their care/accessing of the service; a point highlighted by Cleary, Hunt, Robertson and Escott (2009). This issue was overcome by including information about research in the initial client agreement form, which was discussed at assessment to ensure that clients understood their rights in respect of the research. If a client accessed the *model* and then subsequently decided to opt out of participating in any research (including having their data analysed and/or included in future research and service evaluations), this was recorded on the client's record on the service's client management system, and the data excluded from any dataset analyses.

The body of works that form this thesis were service evaluations as defined by the Health Research Authority (2017), therefore approval from an NHS research ethics committee was not required. Additionally, NHS commissioners were routinely informed in advance of the service evaluations being undertaken and did not communicate any further requirements in relation to the studies. Clients accessing the *model* provided advanced consent for their anonymised data to be used by the organisation for routine evaluations (including board reports, service evaluations, publicity, and promotional purposes), and were asked if they were happy to share their experience of accessing the service by way of contributing to an anonymised case study and/or client testimonial.

As Self Help Services was a small organisation at the time when the studies comprising the portfolio of works were undertaken, the organisation did not have a formal research policy in place but followed ethical procedures in line with those of the NHS. The organisation now has a Research and Evaluation Policy and a Privacy Policy (Self Help Services, 2021), which cover procedures regarding ethics, client participation and research.

# 5.2 Methodology and limitations

All studies were carried out in a real-world setting; important because eTherapy research is not typically undertaken in such environments (Adelman et al., 2014; Karyotaki et al., 2017). Whilst real-world evidence (RWE) is often assigned lower credibility, it is practical in nature, reflecting actual practice being research that focusses on analysis of vast quantities of data that has already been gathered, such as service evaluations (Kim, Lee & Kim, 2018). Realworld studies are likely to provide a more realistic view of outcomes compared to those derived from highly controlled research studies, for example RCTs (**Elison et al., 2014**), because participants are representative of client populations seen in services and do not have to meet strict inclusion and exclusion criteria. Therefore, real-world studies can give information on utilisation patterns and health outcomes, and when combined with RCTs, provide a fuller view of the pros and cons of an intervention. They can also be used to determine whether results from RCTs are generalisable to real-world client groups and different settings and, therefore, have wider generalisability (Blonde, Khunti, Harris, Meizinger & Skolnik, 2018). As most of the eTherapy programmes delivered through the model had already had their efficacy demonstrated via RCTs, further evaluation of their effectiveness was not required. Instead, the objective was to evaluate whether the model, when used to deliver eTherapy programmes in a real-world setting, was clinically effective and acceptable in practice. Whilst RCTs are considered the 'gold standard' methodology when examining and evaluating interventions (Hariton & Locascio, 2018), leading psychological therapy professionals, particularly those from counselling backgrounds, are critical of RCTs; although they believe that the method is likely to continue to be the cornerstone of commissioning policy for some time (Cooper & Reeves, 2012). However, as eTherapy is considered a complex, disruptive digital innovation comprising multiple tailorable and modifiable components, RCTs may not be as applicable. A disruptive innovation is one that interferes or disrupts traditional services - in this case, face-to-face delivered psychological therapies, in a way that is not expected - and which leads to services being offered at lower prices or to different client groups, which can then give rise to affordable healthcare (Glabman, 2009).

The Medical Research Council (MRC) proposes that multiple methodologies should be used (including those which examine implementation of interventions in healthcare) when researching complex healthcare interventions such as eTherapy. They have a framework for the development and evaluation of complex healthcare interventions (Craig et al., 2008), which includes piloting, describing, and evaluating, which is consistent with the approach taken when examining, evaluating, and reviewing the *model*. Furthermore, regarding the acceptability of new interventions, the MRC suggests that this can be examined via feasibility

studies as per **Elison et al. (2017)**, to investigate delivery models and client compliance; the latter being issues of importance when considering effectiveness in a broader sense (Craig et al., 2008; Moore et al., 2015).

In the portfolio of works, studies focusing on the effectiveness of the *model* - **Cavanagh et al. (2011b)**, **Elison et al. (2014, 2017)**, **Luik et al. (2017)** and **Gellatly et al. (2018)** - did not compare the intervention with a control group; therefore, there is a possibility that outcomes arose due to other factors, such as clients getting help elsewhere. Control group type is known to significantly mediate effect sizes (Grist & Cavanagh, 2013) and is important to consider; however, this was not possible to implement due to the studies being service evaluations. Service evaluation methodology was chosen because the studies (HRA, 2017):

- Were designed and solely conducted to determine and define current care
- Answered the question of whether the service met the IAPT standard
- Had no randomisation
- Involved an intervention that was already in use and where clients had already made a choice to access the intervention before the service evaluation
- Included analysis of existing data; specifically, pre- and post-data in the quantitative studies

Whilst the *model's* short-term, post-treatment effectiveness was demonstrated, the lack of focus on follow-up, due to service capacity constraints was a common issue and a limitation across many of the studies (**Cavanagh et al., 2011a; 2011b; Elison et al., 2017; Gellatly et al., 2018**), though is an issue commonly reported in eTherapy research (Andersson et al., 2018b).

A further limitation is the lack of focus on clients who dropped out; an issue across the portfolio of works except for **Lidbetter & Bunnell, 2013**. Given relatively high drop out rate has been identified as being a key issue in eTherapy (Schmidt, Forand & Strunk, 2019), it

would have been useful to have looked at this issue further, however due to the studies being service evaluations, this was not possible. Additionally, the issue of drop out is further complicated by there being a lack of an agreed reporting standard regarding completion (Etzelmueller et al., 2020) with much variation existing in terms of what constitutes a completion in eTherapy.

In IAPT, treatment completion/what constitutes a treatment episode, is defined as being receipt of two or more treatment contacts/sessions (Moller et al., 2019) for any intervention including eTherapy. Whilst **Gellatly et al., 2018,** (which focussed on the remote *model*), reported a treatment completion rate of 53.4% based in accord with the IAPT definition of completion, **Cavanagh et al., 2011b**, (which focussed solely on the venue *model*), reported 53% of clients completed all eight sessions of eTherapy. Given the lack of consistent definition of completion across the portfolio of works, it has not been possible to date to determine whether completion rate differs depending on the version of the *model* accessed, however when compared to the overall eTherapy drop out rate reported by Richards and Richardson, 2012 of 57%, the rates reported in **Gellatly et al., 2018** and **Cavanagh et al., 2011b** respectively of 46.6% and 47% compare favourably.

Overall, emphasis on the peer-support element of the *model* was less than would have been optimal because of greater emphasis being placed on programmes and on the reporting of quantitative outcomes. Bias in health research is a key issue to be aware of and, in the case of conflicts of interest, can lead to inferior quality studies (Odierna, Forsyth, White & Bero, 2013); though, in the studies presented in the portfolio, this was hard to completely eliminate. However, declaration/conflicts of interest were acknowledged by relevant authors to ensure transparency in respect of their positions in various eTherapy programme software companies (**Cavanagh et al., 2011a, 2011b; Elison et al., 2014, 2017; Luik et al., 2017**) in keeping with the US (United States) Institute of Medicine (IOM) report's recommendations (Field & Lo, 2009).

Finally, a key area to reflect on is how acceptability and effectiveness were measured. Across the body of works, clinical effectiveness was determined using a positivist, quantitative approach when determining whether symptom severity reduced post treatment (Cavanagh et al., 2011b, Elison et al., 2014, 2017; Luik et al., 2017; Gellatly et al., 2018). Benchmarking the *model* against IAPT outcomes was the most applicable standard to compare the *model's* effectiveness against, however the downside to this approach is that a drop in symptom severity may not necessarily mean that clients have recovered from depression/anxiety to the extent that they are no longer adversely affected. Whilst improvement can be examined by numerical analysis to determine changes in depression or anxiety, this is only one way of proving an intervention's effectiveness. Evidence-based healthcare is more encompassing, involving the systematic collection, synthesis, and application of all evidence available, including the views of clients/data subjects and others aside from experts, as well as secondary sources such as opinion pieces and other articles. Rycroft-Malone et al. (2004) describe four main types of evidence that can be used: research, professional knowledge/clinical experience, patient experience and preferences, and local data and information. A more holistic evidence-based approach involving routinely seeking clients' views via focus groups, interviews and examining client-reported, subjective symptoms using tools such as the Psychological Outcome Profiles tool - PSYCHLOPS (Ashworth et al., 2004), which facilitates a client-centred definition of therapy outcome that is client generated - would have been advantageous.

# **5.3 Pragmatic paradigm**

The pragmatic research paradigm has its foundations in the philosophy of pragmatism (Maxcy, 2003) and encompasses a wide range of methods, based on the notion that researchers should adopt the methodological approach that works best for the research

problem that is being investigated (Kaushik & Walsh, 2019). The pragmatist philosophy is underpinned by the belief that knowledge is socially constructed, and that reality is what works. Furthermore, meaning is not something that can be separated from human experience and need, and is dependent on the context (Dillon et al., 2000). The pragmatist paradigm is one that aligns itself to understanding and addressing problems in the real world (Kaushik & Walsh, 2019) and was used throughout the body of works. This meant the author was not committed to any system of research philosophy, and instead was free to choose whichever approach supported an optimal understanding of the research problem (Bryman, 2008; Mertens, 2005 Patton, 1990).

Consequently, five of the papers in the portfolio of works used a quantitative approach; in keeping with the positivist epistemology approach taken by NICE in its recommendation for psychological therapies (Mollon, 2008; Guy, 2012). Two papers were qualitatively focused. Although a quantitative approach is key when benchmarking data against national outcome data such as IAPT, NICE has attracted criticism for being too numbers-driven and for taking a positivist epistemology in its recommendation for psychological therapies (Mollon, 2008; Guy, 2012), with some suggesting that an interpretive paradigm could assist instead with shaping service-based cultures and changing evaluation; giving rise to an overall improvement in the quality of CBT research (Williams, 2015).

Whilst RCTs provide evidence of what works for the average client (McLeod, 2011), this

form of large-scale study does not support identification of subjective issues that, in turn, may also impact and predict outcomes. Positivist paradigms have reigned in eTherapy research at the expense of qualitative research on client experience, acceptability, and real-world delivery models in a wide range of client groups, beyond those typically the subject of research. However, more needs to be known about including what aids and hinders client engagement needed, along with research into the reasons for overall adherence, take-up, dropout, and completion (Rost et al., 2017). The *model* itself developed directly as a result of client feedback; transforming over time from being available only via the initial community venuebased model (**Cavanagh et al., 2011b**) to accessibly by clients from the comfort of their homes via the remote model (**Gellatly et al., 2018**). This development happened solely because clients said that they didn't want to have to travel to the service, finding this problematic and a barrier. Had the model failed to evolve in this manner, it is likely that eventually clients would have voted with their feet, drop-out rates would have increased, and completion rates would have been adversely affected.

Furthermore, the views of those with lived experience (of critical importance since this provides a qualitative contribution to research, shedding light on issues such as acceptability, accessibility, etc.) are less likely to be the focus of research (Glasby, 2006).

Whilst the body of works includes two qualitative studies (**Lidbetter & Bunnell, 2013**; **Cavanagh et al., 2011a**) that have case studies that shed some light on acceptability, it is recognised that much more in-depth qualitative research is needed to understand a range of issues, including:

- the mode of delivery of support that clients prefer
- the optimum amount of support required
- who is best placed to provide the support?
- what qualities the supporter should have
- whether is it necessary for the supporter to have received professional training

**Gellatly et al. (2018)** commented on whether supporters should have professional training, and Cuijpers et al. (2009) stated that all the above were areas that required further focus. A phenomenological approach could be taken in the future to address these issues, as this form of research is concerned with the lived experience of clients and their subjective understandings of their experience. Adopting this type of approach would move beyond

research studies that simply report that clients find the approach acceptable and, instead, to those that provide insight as to why the service is acceptable and the identification of moderators of engagement and outcomes (Grist & Cavanagh, 2013).

The pragmatic paradigm taken across this body of works supports both interpretive and positivist research approaches; supporting the notion that each has their value when investigating the efficacy and acceptability of the *model*. Several studies have called for the pragmatic effectiveness of eTherapy to be further investigated (Adelman et al., 2014; Karyotaki et al., 2017), and there is a growing body of evidence to suggest that real-world studies, as well as RCTs, are needed when evaluating complex interventions such as multi-component interventions where programme and service delivery model are important (Craig, Dieppe & Macintyre, 2008).

## 5.4 Reflections

As a researcher, it is important to be aware of what is influencing your external and internal responses, as well as knowing what has impacted on the relationship with the research topic and the research subjects; this being the skill of reflexivity (Etherington, 2004).

It is recognised that the studies that form the body of works were driven by the author's desire to firstly prove that a user-led TSO could deliver high quality eTherapy services via the *model* that could achieve comparable outcomes to face-to-face delivered IAPT services, and to be able to demonstrate this to key stakeholders, including commissioners.

The author's firm belief (born out of personal experience of living with anxiety) that it is not necessary for mental health services to always be delivered by professionals, and that those with lived experience of a mental health difficulty are key to effective mental health service delivery when such individuals are provided with appropriate training and support, may have had an influencing effect that could have affected the studies and the overall direction of travel of the research undertaken. These beliefs and desires were outwardly expressed by the author in conversations with staff and volunteers and through external and internal service promotional materials and may well have been similarly communicated unintentionally by those delivering the service. This may have affected outcomes and results through the concept of insider bias, though was mitigated by bringing in a team of external academics eminent in the field of eTherapy research to facilitate independent evaluation of the *model*. Given the ethos of the charity and its user-led status is something that is overtly celebrated and recognised as a key strength, this could have played a role in terms of bias. Being aware of the existence of this ideology, culture, and the politics (for example, at the time, it was not a widely held view that user-led organisations could deliver clinically effective services), is an essential element of reflexivity, adding validity and rigour by providing information about the context of the location of the data (Etherington, 2004).

As leader of the organisation, the author's role included developing and executing the organisation's strategy and developing the *model* with the goal of getting it commissioned by the NHS on a mainstream basis and accepted as a viable, effective, low-intensity IAPT service. Other key activities undertaken as part of the role were to assist with making decisions regarding selecting the type of research study to undertake, identifying appropriate research partners, assisting with developing the research question(s), agreeing the optimum methodological approach with partners, identifying when research should be carried out and on what to focus, determining data sets, revising manuscripts, and formulating dissemination strategies. Over the years, the author built up research skills, albeit in a non-traditional way without the appropriate language, as a mental health strategist and organisational leader where it was necessary to continuously juggle competing strategic and operational demands, whilst building the evidence base for the *model*.

129

Looking back, whilst the goal was achieved in getting the *model* successfully mainstreamed across large parts of Greater Manchester (for approaching a decade), forming a key component of low intensity IAPT services in the region, it is without doubt that if there were the opportunity to embark on this path again, things may have been done differently. For example, choosing to work with more independent experts akin to the research on eTherapy undertaken via the independent REAACT trial (Gilbody et al., 2015), and the more recent independent research carried out by Lou et al. (2020). At times it was challenging striking a balance between meeting both research and organisational objectives, versus balancing this with sometimes slightly differing needs or priorities of research partners. For example, the key goal for some of the research partners was to further demonstrate that their company's eTherapy programme was effective, however the author's goal was to show that the model was effective in a real-world setting. Gilbody et al. (2015) found that commercially developed cCBT programmes conferred no benefit over a free-to-use product, and one must be cognisant of possible competing agendas, particularly when working with corporates - in this case, software companies - as this may lead to research bias. Across the portfolio of works, this was in part addressed by ensuring that author affiliations and conflicts of interest were routinely declared.

To date, no studies have been undertaken that provide comprehensive operational delivery detail including SOPs for the *model* because of commercial sensitivities, as disclosing such information could have potentially jeopardised the charity's commercial interests at the time. Whilst this is so, this meant that exact replication of the *model* would be difficult, which in turn could be said to affect the verification of the findings by others (Richards et al., 2003).

Finally, allegiance bias, defined as being something that occurs when the results of a study are affected by the researcher's theoretical or treatment preferences (Luborsky, Singer & Luborsky, 1975) in outcomes studies, is something that must be considered. If this bias exists, it could put the validity of outcomes studies into dispute, however, was reduced across the body of works by involving a variety of people in the studies and where research collaborations between investigators with differing alliances and complementary areas of expertise existed (Leykin & DeRubeis, 2009).

#### 5.5 Section summary

In this section, methodology, limitations, personal reflections, the pragmatic paradigm, and ethical considerations have been examined.

Specifically, researcher bias and the competing agenda between researcher and founder and leader have been discussed, along with the issue of commercial interest. The author's own views on issues such as peer support and their personal investment in the *model* are discussed in relation to the impact this may have had on the studies that form the portfolio of works in respect of allegiance bias.

Respect for client autonomy and their right to self-govern are addressed with recognition that these principles were very much supported by the organisation's ethos and belief that those with lived experience are best placed to support themselves and others, and that a 'them and us' hierarchical service structure is not necessary.

Regarding methodology, service evaluations are discussed along with the value of conducting research in real-world settings (generalisability and ability to provide more realistic outcomes), since this was the approach (and setting) used in most studies that form the portfolio of works. This approach is contrasted with that of RCTs, and an argument is put forward as to why eTherapy as a complex, disruptive innovation is not particularly suited to being researched using this methodology.

The limitations of the methodological stance taken across the portfolio of works is critiqued, including the lack of follow-up data and control group, as well as the absence of focus on participants failing to complete the full eTherapy programme when delivered through the

*model.* Furthermore, the reliance on numerical data for the reporting on outcomes is discussed, with recognition that evidence-based healthcare is wider than this and therefore use of other outcome measures, particularly patient-reported ones, may have been preferable.

The pragmatic research paradigm is explored, along with a rationale provided as to why it was chosen (freedom for the author to select the most appropriate methodological approaches for the studies that form the portfolio of works) and which resulted in the portfolio comprising five quantitative papers and two papers that are qualitative.

The next section summarises the thesis' key outcomes, whilst also providing recommendations for practice and future research and a conclusion to the thesis.

#### **Section Six**

In this concluding section, thesis objectives are revisited and recommendations for practice and future research are provided.

#### 6.1 Key thesis outcomes

The thesis objectives were to:

- 1. Examine the development of the model
- 2. Review and evaluate the clinical effectiveness of the model
- 3. Evaluate the acceptability of the *model*

#### Outcomes

The outcomes in terms of meeting the above objectives have been demonstrated through the thesis and achieved through the portfolio of published works.

The development of the *model* has been examined (Objective 1) across the body of works (**Cavanagh et al., 2011a, 2011b; Elison et al., 2014, 2017; Gellatly et al., 2018; Lidbetter & Bunnell, 2013; Luik et al., 2017**). This has included its initial creation and development into further iterations of the remote and hybrid *model*, capable of providing support for clients experiencing anxiety and depression, as well as insomnia and dual diagnosis issues is described.

The effectiveness of the *model* has been reviewed and evaluated (Objective 2) in **Cavanagh** et al. (2011b), Elison et al. (2014, 2017), Gellatly et al. (2018) and Luik et al. (2017), with the finding that it is effective in the treatment of anxiety and depression, and for clients also experiencing sleep (Luik et al., 2017) and dual diagnosis issues (Elison et al., 2014, 2017).

The *model's* acceptability has been evaluated (Objective 3) via **Cavanagh et al. (2011a, b)** and **Lidbetter and Bunnell (2013)** and demonstrated to be acceptable, though further research is needed to understand factors that moderate acceptability as well as further examination of those who drop-out. This is particularly important given the link between treatment adherence, acceptability, and effectiveness.

Through the development and publication of the seven focused papers that are presented in this thesis, a unique contribution to eTherapy knowledge has been established through describing the development of a non-clinical peer-supported *model* and demonstrating that, when this is delivered in a real-world setting, it is effective and acceptable in the management of anxiety and depression in adults.

#### 6.2 Recommendations for future research and practice

As a result of the findings from the included articles that form the portfolio of published works, the following recommendations for future research are proposed:

- Examination of client step-up, step-down, step-sideways, and step-out utilisation rates, and how the *model* connects with the NHS, TSO, and statutory providers.
- Research focusing on clients that fail to complete therapy via the *model* and reasons for drop-out; an issue known to impact on the ability to predict the longer-term impact of eTherapy (Waller & Gilbody, 2009). Specifically, it would be helpful to compare drop out rates for different versions of the *model* and to further understand why clients drop out of eTherapy and whether drop out can be predicted so as to inform any future developments to the *model* that may be implemented aimed at addressing engagement and client retention.
- Examination of the optimal nature of client-eTC care pathway touchpoints, duration, nature, and mode of interaction to identify the key ingredients of an effective eTherapy

*model* (Hollis et al., 2017), including what support involves (Hadjistavropoulos, Schneider, Klassen, Dear & Titov, 2018) and its impact on adherence and therefore effectiveness and acceptability.

- Research focusing on the longer-term follow-up of clients post-discharge, in respect of clinical and employment outcomes in those accessing the *model* when delivered as part of an IAPT, step 2 service.
- The peer-supported *model* should be directly compared with the same intervention supported by a PWP, through an RCT. Research of this nature would provide insight into the impact that the workforce may have on outcomes and is the subject of a current study led by the thesis' author.
- Research into issues that may affect the future scaling up implementation of eTherapy into existing mental health services, to gain a better understanding of how to deliver and integrate eTherapy within mental health and other services (Thew, 2020).

Other recommendations for research arising from this thesis include:

- Investigating the role that the therapeutic relationship plays (Margison et al., 2000) on eTherapy outcomes achieved through the *model*; of interest given it has been reported that a different type of therapeutic relationship develops when those with lived experience deliver services instead of that seen in traditional staff-client relationships (Sweeney et al., 2014).
- Research into client perception of eTherapy using scales such as the Perceptions of Computerized Therapy Questionnaire-Patient Version - PCTQ-P (Carper, McHugh, Murray & Barlow, 2014) - as this would add further understanding of acceptability.
- As technology and AI develops, eTherapy is likely to become more personalised and intuitive (Wright, 2019); in part by incorporating more explicit inclusion of common factors of traditionally delivered therapy (Peck, 2010) and less focus being placed on

therapy modality. In this regard, an evaluation study is currently underway, led by the thesis' author comparing treatment outcomes in clients experiencing anxiety accessing different therapy modalities including clinical hypnotherapy, accessed face-to-face, by telephone and online.

- As blended models become more widespread, results from a largescale stakeholder survey conducted in eight European countries found acceptability was greater when a blend of face-to-face treatment was integrated with eTherapy (Topooco et al., 2017). Therefore, research into outcomes achieved by models that offer a blended approach would too be beneficial.
- Finally, it may also be advantageous to undertake feasibility studies regarding the piloting eTherapy services at high-intensity level for clients experiencing greater symptom severity.

As a result of the findings of the portfolio of works, the following recommendations for practice are being made:

- That the *model*, when used to deliver eTherapy programmes (such as LLTF Children and Young People CYP), should be piloted in the community.
- That guidelines should be published to increase fidelity to eTherapy delivery models that incorporate client views and experiences (Roddis, Liversedge & Ryder, 2019).
- To develop a standardised training programme for eTherapy supporters (Hadjistavropoulos et al., 2018; Marks & Cavanagh, 2009), as it may not be a matter of who delivers the service but instead the quality of training that eTherapy supporters receive (Thew, 2020).

## 6.3 Conclusions

This thesis has described the development and evaluation of a non-clinical, fully peersupported, eTherapy *model*.

The ground-breaking peer-supported *model* pioneered at Self Help Services has been conceptualised, developed, and designed under the author's leadership, for adults experiencing anxiety and depression and additional difficulties with sleep and dual diagnosis issues, when offered as a low-intensity IAPT service. Furthermore, in all its iterations (venue, remote, and hybrid versions), the *model* has been demonstrated to be both effective and acceptable; capable of achieving outcomes equivalent to those reported by IAPT services.

The author instigated and co-instigated, as well as conceptualising research on the *model* via the studies that form the portfolio of works; leading and directing the strategic development, delivery, and evaluation of the *model*. This has included assisting with refining the aims and objectives of the research studies, ensuring compliance with service governance regulations, and playing a critical role in the dissemination of papers to service user and professional populations.

Typically, eTherapy research in regard to the specification, nature, and effectiveness when studied in real-world settings is scarce (Adelman et al., 2014; Karyotaki et al., 2017); however, by contrast, the research represented via the studies that form the portfolio of works, directed by the author, makes a unique contribution to eTherapy literature by providing evidence of the clinical effectiveness and acceptability of a peer-supported *model* of eTherapy when delivered in a real-world setting.

Though roll out of the *model* was originally recommended a decade ago (**Cavanagh et al., 2011a**), peer-supported eTherapy has yet to take its rightful place in the primary care mental health landscape, such that it is widely adopted and incorporated into mainstream practice; a

137

phenomenon commonplace with disruptive (Barnett et al., 2011) and all innovations (Rogers, 2003), despite its potential being clear (Gretton & Honeyman, 2016).

Given demand for mental health services is expected to rise because of the COVID-19 pandemic (Holmes et al., 2020), never has there been a time when the *model* is more needed, yet its future expansion could be at risk given introduction of guidance that states that the intervention should be supported by clinicians (National Collaborating Centre for Mental Health, 2018). COVID-19 has undoubtedly prompted a radical increase in the delivery of online psychotherapeutic sessions, which contrasts greatly with the former infrequent use of online mental health interventions (Feijt et al., 2020), including eTherapy. It seems unlikely therefore that eTherapy as an intervention will ever return to the former pre-pandemic lower usage levels and as such is likely to form an ever-larger component of the mental health offer available globally.

Finally, it is essential that the potential of those with lived experience of mental health difficulties, such as anxiety and depression, in supporting others is realised in the mental health workforce. Specifically, peer support must be integrally woven into mainstream mental health services such as eTherapy, in the way the *model* facilitates and, in doing so, helping people to help themselves.

## References

- Adelman, C., Panza, K., Bartley, C., Bontempo, A., & Bloch, M. (2014). A meta-analysis of computerized cognitive-behavioral therapy for the treatment of DSM-5 anxiety disorders. *The Journal of Clinical Psychiatry*, 75(7), E695-e704.
- Aggleton, P., & Chalmers, H. (2000). *Nursing models and nursing practice* (pp. 6-7). Hampshire: Palgrave.
- Allen, P., Bartlett, W., Perotin, V., Zamora, B., & Turner, S. (2011). New forms of provider in the English National Health Service. *Annals of Public and Cooperative Economics*, 82(1), 77-95.
- Allen, P., Turner, S., Bartlett, W., Perotin, V., Matchaya, G., & Zamora, B. (2012). Provider diversity in the English NHS: a study of recent developments in four local health economies. *Journal of Health Services Research & Policy*, 17(1\_suppl), 23-30.
- Alvaro, P. K., Roberts, R. M., & Harris, J. K. (2013). A systematic review assessing bidirectionality between sleep disturbances, anxiety, and depression. *Sleep*, 36(7), 1059-1068.
- American Psychiatric Association. (2013). Diagnostic and statistical manual of mental disorders. *BMC Medicine*, 17, 133-137.
- Anderson, C. (2010). Presenting and evaluating qualitative research. *American Journal of Pharmaceutical Education*, 74(8).
- Anderson, P., Jacobs, C., & Rothbaum, B. O. (2004). Computer-supported cognitive behavioral treatment of anxiety disorders. *Journal of Clinical Psychology*, 60(3), 253-267.
- Andersson, G., Bergström, J., Holländare, F., Carlbring, P. E. R., Kaldo, V., & Ekselius, L. (2005). Internet-based self-help for depression: randomised controlled trial. *The British Journal of Psychiatry*, 187(5), 456-461.
- Andersson, G., Carlbring, P., Berger, T., Almlöv, J., & Cuijpers, P. (2009). What makes internet therapy work? *Cognitive Behaviour Therapy*, *38*(S1), 55-60.
- Andersson, G., Carlbring, P., Furmark, T., & SOFIE Research Group. (2012). Therapist experience and knowledge acquisition in internet-delivered CBT for social anxiety disorder: a randomized controlled trial. *PloS One*, 7(5), e37411.
- Andersson, G., & Cuijpers, P. (2009). Internet-based and other computerized psychological treatments for adult depression: a meta-analysis. *Cognitive Behaviour Therapy*, 38(4), 196-205.

- Andersson, G., Cuijpers, P., Carlbring, P., Riper, H., & Hedman, E. (2014). Guided Internet-based vs. face-to-face cognitive behavior therapy for psychiatric and somatic disorders: a systematic review and meta-analysis. *World Psychiatry*, 13(3), 288-295.
- Andersson, G. (2018a). Internet interventions: past, present and future. *Internet Interventions*, 12, 181-188.
- Andersson, G., Rozental, A., Shafran, R., & Carlbring, P. (2018b). Long-term effects of internet-supported cognitive behaviour therapy. *Expert Review of Neurotherapeutics*, 18(1), 21-28.
- Andersson, G., Carlbring, P., Titov, N., & Lindefors, N. (2019a). Internet interventions for adults with anxiety and mood disorders: A narrative umbrella review of recent metaanalyses. *The Canadian Journal of Psychiatry*, 64(7), 465-470.
- Andersson, G., Titov, N., Dear, B. F., Rozental, A., & Carlbring, P. (2019b). Internetdelivered psychological treatments: from innovation to implementation. *World Psychiatry*, 18(1), 20-28.
- Andrews, G., Cuijpers, P., Craske, M. G., McEvoy, P., & Titov, N. (2010). Computer therapy for the anxiety and depressive disorders is effective, acceptable and practical health care: a meta-analysis. *PloS One*, 5(10), e13196.
- Andrews, G., Basu, A., Cuijpers, P., Craske, M. G., McEvoy, P., English, C. L., & Newby, J. M. (2018). Computer therapy for the anxiety and depression disorders is effective, acceptable and practical health care: an updated meta-analysis. *Journal of Anxiety Disorders*, 55, 70-78.
- Anxiety UK. (2020). Nutrition and Anxiety: A self-help guide. Retrieved from https://www.anxietyuk.org.uk/products/anxiety-condition/anxiety/nutrition-andanxiety-a-self-help-guide/
- Apolinário-Hagen, J., Kemper, J., & Stürmer, C. (2017). Public acceptability of e-mental health treatment services for psychological problems: a scoping review. *Journal of Medical Internet Research Mental Health*, 4(2), e10.
- Ardito, R. B., & Rabellino, D. (2011). Therapeutic alliance and outcome of psychotherapy: historical excursus, measurements, and prospects for research. *Frontiers in Psychology*, 2, 270.
- Arnberg, F. K., Linton, S. J., Hultcrantz, M., Heintz, E., & Jonsson, U. (2014). Internetdelivered psychological treatments for mood and anxiety disorders: a systematic review of their efficacy, safety, and cost-effectiveness. *PloS One*, (5), e98118. doi:10.1371/journal.pone. 0098118

- Ashworth, M., Shepherd, M., Christey, J., Matthews, V., Wright, K., Parmentier, H., ... & Godfrey, E. (2004). A client-generated psychometric instrument: the development of 'PSYCHLOPS.' *Counselling and Psychotherapy Research*, 4(2), 27-31.
- Bailie, H. A., & Tickle, A. (2015). Effects of employment as a peer support worker on personal recovery: A review of qualitative evidence. *Mental Health Review Journal*, 20(1), 48–64.
- Baird, B., Cream, J., & Weaks, L. (2018). *Commissioner Perspectives on Working with the Voluntary, Community and Social Enterprise Sector*. London: King's Fund.
- Baldwin, D. S., Anderson, I. M., Nutt, D. J., Allgulander, C., Bandelow, B., den Boer, J. A.,
  ... & Malizia, A. (2014). Evidence-based pharmacological treatment of anxiety
  disorders, post-traumatic stress disorder and obsessive-compulsive disorder: a
  revision of the 2005 guidelines from the British Association for
  Psychopharmacology. *Journal of Psychopharmacology*, 28(5), 403-439.
- Barak, A., Klein, B., & Proudfoot, J. G. (2009). Defining internet-supported therapeutic interventions. *Annals of Behavioral Medicine*, *38*(1), 4-17.
- Barnett, J., Vasileiou, K., Djemil, F., Brooks, L., & Young, T. (2011). Understanding innovators' experiences of barriers and facilitators in implementation and diffusion of healthcare service innovations: a qualitative study. *BMC Health Services Research*, 11(1), 342.
- Barrett, M. S., Chua, W. J., Crits-Christoph, P., Gibbons, M. B., & Thompson, D. (2008). Early withdrawal from mental health treatment: Implications for psychotherapy practice. *Psychotherapy: Theory, Research, Practice, Training*, 45(2), 247.
- Barth, J., Munder, T., Gerger, H., Nüesch, E., Trelle, S., Znoj, H., ... & Cuijpers, P. (2013). Comparative efficacy of seven psychotherapeutic interventions for patients with depression: a network meta-analysis. *Focus*, 14(2), 229-243.
- Basset, T., Faulkner, A., Repper, J., & Stamou, E. (2010). *Lived Experience Leading the Way: Peer Support in Mental Health.* London: Together UK.
- Bastien, C. H., Vallières, A., & Morin, C. M. (2001). Validation of the Insomnia Severity Index as an outcome measure for insomnia research. *Sleep Medicine*, 2(4), 297–307. https://doi.org/10.1016/s1389-9457(00)00065-4
- Batterham, P. J., Calear, A. L., Sunderland, M., Kay-Lambkin, F., Farrer, L. M., & Gulliver, A. (2019). A brief intervention to increase uptake and adherence of an online program for depression and anxiety: Protocol for the Enhancing Engagement with Psychosocial Interventions (EEPI) Randomized Controlled Trial. *Contemporary Clinical Trials*, 78, 107-115.

- Baumeister, H., Reichler, L., Munzinger, M., & Lin, J. (2014). The impact of guidance on Internet-based mental health interventions - A systematic review. *Internet Interventions*, 1(4), 205-215.
- Baxter, A. J., Scott, K. M., Vos, T., & Whiteford, H. A. (2013). Global prevalence of anxiety disorders: a systematic review and meta-regression. *Psychological Medicine*, 43(5), 897-910.
- Baxter, A. J., Scott, K. M., Ferrari, A. J., Norman, R. E., Vos, T., & Whiteford, H. A. (2014). Challenging the myth of an "epidemic" of common mental disorders: trends in the global prevalence of anxiety and depression between 1990 and 2010. *Depression and Anxiety*, *31*(6), 506-516.
- Beattie, A., Shaw, A., Kaur, S., & Kessler, D. (2009). Primary-care patients' expectations and experiences of online cognitive behavioural therapy for depression: a qualitative study. *Health Expectations*, 12(1), 45-59.
- Beauchamp, T.L., & Childress, J. F. (2008). Principles of Biomedical Ethics (6<sup>th</sup> ed.). New York: Oxford University Press.
- Beck, A., Naz, S., Brooks, M., & Jankowska, M. (2019). Improving Access to
   Psychological Therapies (IAPT). *Black, Asian and Minority Ethnic Service User Positive Practice Guide. 2019.* Retrieved from
   https://www.scribd.com/document/458163875/IAPT-BAME-PPG-2019-pdf
- Bee, P. E., Lovell, K., Lidbetter, N., Easton, K., & Gask, L. (2010). You can't get anything perfect: "User perspectives on the delivery of cognitive behavioural therapy by telephone." *Social Science and Medicine*, 71(7), 1308-1315.
- Bennett-Levy, J., Richards, D. A., Farrand, P., Christensen, H., Griffiths, K. M., Kavanagh,
  D. J., ... & Proudfoot, J. (2010). Low intensity CBT interventions: a revolution in mental health care. *Oxford Guide to Low Intensity CBT Interventions*, *3*, 18.
- Bennion, M. R., Hardy, G., Moore, R. K., & Millings, A. (2017). E-therapies in England for stress, anxiety or depression: what is being used in the NHS? A survey of mental health services. *BMJ Open*, 7(1).
- Ben-Zeev, D., Schueller, S. M., Begale, M., Duffecy, J., Kane, J. M., & Mohr, D. C. (2015). Strategies for mHealth research: lessons from 3 mobile intervention studies. Administration and Policy in Mental Health and Mental Health Services Research, 42(2), 157-167.
- Beresford, P., Nettle, M., & Perring, R. (2010). Towards a social model of madness and distress. *Exploring what service users say. York: Joseph Rowntree Foundation*, 27-41.

- Berger, T., Caspar, F., Richardson, R., Kneubühler, B., Sutter, D., & Andersson, G. (2011). Internet-based treatment of social phobia: a randomized controlled trial comparing unguided with two types of guided self-help. *Behaviour Research and Therapy*, 49(3), 158-169.
- Berger, T. (2017). The therapeutic alliance in internet interventions: a narrative review and suggestions for future research. *Psychotherapy Research*, *27*(5), 511-524.
- Berman, A. H., Bergman, H., Palmstierna, T., & Schlyter, F. (2005). Evaluation of the Drug Use Disorders Identification Test (DUDIT) in criminal justice and detoxification settings and in a Swedish population sample. *European Addiction Research*, 11(1), 22-31.
- Biau, D. J., Kernéis, S., & Porcher, R. (2008). Statistics in brief: the importance of sample size in the planning and interpretation of medical research. *Clinical Orthopaedics* and Related Research, 466(9), 2282-2288.
- Binnie, J. (2015). Do you want therapy with that? A critical account of working within IAPT. *Mental Health Review Journal*. 20(2), 79-83. doi:10.1108/MHRJ-11-2014-0044
- Blonde, L., Khunti, K., Harris, S. B., Meizinger, C., & Skolnik, N. S. (2018). Interpretation and impact of real-world clinical data for the practicing clinician. *Advances in Therapy*, 35(11), 1763-1774.
- Boettcher, J., Rozental, A., Andersson, G., & Carlbring, P. (2014). Side effects in Internetbased interventions for Social Anxiety Disorder. *Internet Interventions*, *1*(1), 3-11.
- Boland, R. J., & Keller, M. B. (2008). Antidepressants. Psychiatry, 2123-2160.
- Bond, T., & Dryden, W. (2012). Ethical principles for CBT practitioners. In W. Dryden & R. Branch (Eds.). *The CBT Handbook Therapy*. London: Sage.
- Borgueta, A. M., Purvis, C. K., & Newman, M. G. (2018). Navigating the ethics of Internetguided self-help interventions. *Clinical Psychology: Science and Practice*, 25(2), e12235.
- Bower, P., Gilbody, S. (2005). Stepped care in psychological therapies: access, effectiveness and efficiency. Narrative Literature Review. *The British Journal of Psychiatry*, 186, 11-17.
- Boyles, J. & McKinnon Fathi, N. (2019). 'At what cost? The impact of IAPT on third sector psychological therapy provision,' in C. Jackson & R. Rizq (Eds.), *The Industrialisation of Care*, PCCS Books Ltd, pp. 232-275.
- Brabyn, S., Araya, R., Barkham, M., Bower, P., Cooper, C., Duarte, A., ... & Mattock, R.(2016). The second Randomised Evaluation of the Effectiveness, cost-effectiveness

and Acceptability of Computerised Therapy (REEACT-2) trial: does the provision of telephone support enhance the effectiveness of computer-delivered cognitive behaviour therapy? A randomised controlled trial. *Health Technology Assessment*, 1-64.

- Bradstreet, S. (2006). Harnessing the 'lived experience': formalising peer support approaches to promote recovery. *The Mental Health Review*, *11*(2), 33.
- Breedvelt, J. J., Zamperoni, V., Kessler, D., Riper, H., Kleiboer, A. M., Elliott, I., ... & Bockting, C. L. (2019). GPs' attitudes towards digital technologies for depression: an online survey in primary care. *British Journal of General Practice*, 69(680), e164-e170.
- Brettle, A. (2012). Psychological therapies for depression. A summary of the evidence. Retrieved from https://www.bacp.co.uk/media/1962/bacp-psychological-therapiesfor-depression-summary-of-evidence-2012.pdf
- British Association for Counselling and Psychotherapy (2020). Good Practice across the Counselling Professions 004. What works in counselling and psychotherapy relationships. Retrieved from https://www.bacp.co.uk/media/9667/bacp-whatworks-counselling-psychotherapy-relationships-gpacp004-sep20.pdf
- British Medical Association (2020). Measuring progress: Commitments to support and expand the mental health workforce in England. Retrieved from https://www.bma.org.uk/media/2405/bma-measuring-progress-of-commitments-formental-health-workforce-jan-2020.pdf
- Brown, G. W., & Harris, T. (2012). Social origins of depression: A study of psychiatric disorder in women. London: Routledge.
- Brown, J. S. (2018). Increasing access to psychological treatments for adults by improving uptake and equity: rationale and lessons from the UK. *International Journal of Mental Health Systems*, *12*(1), 67.
- Bryan, A. E., & Arkowitz, H. (2015). Meta-analysis of the effects of peer-administered psychosocial interventions on symptoms of depression. *American Journal of Community Psychology*, 55(3-4), 455-471.
- Bryman, A. (2008). "Of methods and methodology," *Qualitative Research in Organizations and Management: An International Journal*, 3(2), 159-168.
- Buckley, P. F. (2006). Prevalence and consequences of the dual diagnosis of substance abuse and severe mental illness. *The Journal of Clinical Psychiatry*, 67, 5-9.
- Burke, J., Richards, D., & Timulak, L. (2018). Helpful and hindering events in internetdelivered cognitive behavioural treatment for generalized anxiety. *Behavioural and Cognitive Psychotherapy*, 47(3), 386-399.
- Buszewicz M.J., Chew-Graham, C. (2011). Improving the detection and management of anxiety disorders in primary care. *British Journal of General Practice*. 61(589): 489–490.
- Butler, A. C., Chapman, J. E., Forman, E. M., & Beck, A. T. (2006). The empirical status of cognitive-behavioral therapy: a review of meta-analyses. *Clinical Psychology Review*, 26(1), 17-31.
- Bystedt, S., Rozental, A., Andersson, G., Boettcher, J., & Carlbring, P. (2014). Clinicians' perspectives on negative effects of psychological treatments. *Cognitive Behaviour Therapy*, 43(4), 319-331.
- Campos, D., Bretón-López, J., Botella, C., Mira, A., Castilla, D., Mor, S., ... & Quero, S. (2019). Efficacy of an internet-based exposure treatment for flying phobia (NO-FEAR Airlines) with and without therapist guidance: a randomized controlled trial. *BMC Psychiatry*, 19(1), 86.
- Cape, J., & Parham, A. (1998). Relationship between practice counselling and referral to outpatient psychiatry and clinical psychology. *British Journal of General Practice*, 48(433), 1477-1480.
- Carey, G. E., & Braunack-Mayer, A. J. (2009). Exploring the effects of government funding on community-based organizations: 'top-down' or 'bottom-up' approaches to health promotion?. *Global Health Promotion*, 16(3), 45-52.
- Carlbring, P., Andersson, G., Cuijpers, P., Riper, H., & Hedman-Lagerlöf, E. (2018). Internet-based vs. face-to-face cognitive behavior therapy for psychiatric and somatic disorders: an updated systematic review and meta-analysis. *Cognitive Behaviour Therapy*, 47(1), 1-18.
- Carper, M. M., McHugh, R. K., & Barlow, D. H. (2013). The dissemination of computerbased psychological treatment: a preliminary analysis of patient and clinician perceptions. Administration and Policy in Mental Health and Mental Health Services Research, 40(2), 87-95.
- Carper, M. M., McHugh, R. K., Murray, H. W., & Barlow, D. H. (2014). Psychometric analysis of the perceptions of computerized therapy questionnaire-patient version (PCTQ-P). Administration and Policy in Mental Health and Mental Health Services Research, 41(1), 104-113.

- Casement, P. (2009). Beyond words the role of psychoanalysis. *The Psychologist*, 22(5), 404-405.
- Castonguay, L. G., Nelson, D. L., Boutselis, M. A., Chiswick, N. R., Damer, D. D.,
  Hemmelstein, N. A., ... & Spayd, C. (2010). Psychotherapists, researchers, or both?
  A qualitative analysis of psychotherapists' experiences in a practice research
  network. *Psychotherapy: Theory, Research, Practice, Training*, 47(3), 345.
- Cavanagh, K., Shapiro, D. A., Van Den Berg, S., Swain, S., Barkham, M., & Proudfoot, J. (2006). The effectiveness of computerized cognitive behavioural therapy in routine care. *British Journal of Clinical Psychology*, 45(4), 499-514.
- Cavanagh, K., Shapiro, D. A., Van Den Berg, S., Swain, S., Barkham, M., & Proudfoot, J. (2009). The acceptability of computer-aided cognitive behavioural therapy: a pragmatic study. *Cognitive Behaviour Therapy*, 38(4), 235-246.
- Cavanagh, K., Seccombe, N., Lidbetter, N., & Bunnell, D. (2011a). Supported, service-user led, computerised cognitive behavioural therapy (CCBT) self-help clinics. *Journal of Public Mental Health*, 10(4), 225-233.
- Cavanagh, K., Seccombe, N., & Lidbetter, N. (2011b). The implementation of computerized cognitive behavioural therapies in a service user-led, third sector self help clinic. *Behavioural and Cognitive Psychotherapy*, *39*(4), 427-442.
- Cavanagh, K., & Millings, A. (2013). (Inter) personal computing: the role of the therapeutic relationship in e-mental health. *Journal of Contemporary Psychotherapy*, 43(4), 197-206.
- Cavanagh, K., Belnap, B. H., Rothenberger, S. D., Abebe, K. Z., & Rollman, B. L. (2018). My care manager, my computer therapy and me: the relationship triangle in computerized cognitive behavioural therapy. *Internet Interventions*, 11, 11-19.
- Center for Technology and Behavioral Health. (2020). Synthesizing current research on digital health technologies for substance use disorders and co-occurring conditions. Retrieved from https://www.c4tbh.org/program-

reviews/?f=true&category=category-mental-health&commercially\_available=mix

- Cerne, A., Rifel, J., Rotar-Pavlic, D., Svab, I., Selic, P., & Kersnik, J. (2013). Quality of life in patients with depression, panic syndrome, other anxiety syndrome, alcoholism and chronic somatic diseases: a longitudinal study in Slovenian primary care patients. *Wiener Klinische Wochenschrift*, 125(1-2), 1-7.
- Chan, S. W., & Adams, M. (2014). Service use, drop-out rate and clinical outcomes: a comparison between high and low intensity treatments in an IAPT service. *Behavioural and Cognitive Psychotherapy*, 42(6), 747-759.

- Chew, C., & Osborne, S. P. (2009). Exploring strategic positioning in the UK charitable sector: emerging evidence from charitable organisations that provide public service. *British Journal of Management*, 20, 90-105.
- Christensen, H., Griffiths, K. M., & Jorm, A. F. (2004). Delivering interventions for depression by using the internet: randomised controlled trial. *BMJ*, *328*(7434), 265.
- Christensen, H., Griffiths, K. M., & Farrer, L. (2009). Adherence in internet interventions for anxiety and depression: systematic review. *Journal of Medical Internet Research*, 11(2), e13.
- Christensen, H., Batterham, P. J., Gosling, J. A., Ritterband, L. M., Griffiths, K. M., Thorndike, F. P., ... & Mackinnon, A. J. (2016). Effectiveness of an online insomnia program (SHUTi) for prevention of depressive episodes (the GoodNight Study): a randomised controlled trial. *The Lancet Psychiatry*, 3(4), 333-341.
- Churchill, R., Davies, P., Caldwell, D., Moore, T. H., Jones, H., Lewis, G., & Hunot, V. (2010). Interpersonal, cognitive analytic and other integrative therapies versus treatment as usual for depression. *The Cochrane Database of Systematic Reviews*, (9).
- Ciuca, A. M., Berger, T., Crişan, L. G., & Miclea, M. (2016). Internet-based treatment for Romanian adults with panic disorder: protocol of a randomized controlled trial comparing a Skype-guided with an unguided self-help intervention (the PAXPD study). *BMC Psychiatry*, 16(1), 6.
- Clark, D. M., Layard, R., Smithies, R., Richards, D. A., Suckling, R., & Wright, B. (2009). Improving access to psychological therapy: Initial evaluation of two UK demonstration sites. *Behaviour Research and Therapy*, 47(11), 910-920.
- Clark, D. M. (2011). Implementing NICE guidelines for the psychological treatment of depression and anxiety disorders: the IAPT experience. *International Review of Psychiatry*, 23(4), 318-327.
- Clark, D. M. (2018). Realising the mass public benefit of evidence-based psychological therapies: the IAPT program. *Annual Review of Clinical Psychology*, 14.
- Cleary, M., Hunt, G. E., Robertson, M., & Escott, P. (2009). Mental health research through clinical innovation or quality improvement–a reflection on the ethical aspects. *Journal of Ethics in Mental Health*, *4*(2), 1-3.
- Cludius, B., Schröder, J., & Moritz, S. (2018). Expectancy Effects in Self-Help Depression Treatment: First Evidence that the Rationale Given for an Online Study Impacts the Outcome. *Behavioural and Cognitive Psychotherapy*, 46(2), 195-208.

- Clum, G. A., & Watkins, P. L. (2008). Self-help therapies: Retrospect and prospect. In P. L.Watkins & G. A. Clum (Eds.), *Handbook of Self-help Therapies* (p. 419–436). New York: Lawrence Erlbaum Associates Publishers.
- Cochran, W. G. (1952). The χ2 test of goodness of fit. *The Annals of Mathematical statistics*, 315-345.
- Cohen, J. (2013). *Statistical Power Analysis for the Behavioral Sciences*. London: Academic Press.
- Comer, J. S., Blanco, C., Hasin, D. S., Liu, S. M., Grant, B. F., Turner, J. B., & Olfson, M. (2010). Health-related quality of life across the anxiety disorders: results from the national epidemiologic survey on alcohol and related conditions (NESARC). *The Journal of Clinical Psychiatry*, 72(1), 43-50.
- Connell, J. & Barkham, M. (2007). *CORE-10 User Manual, Version 1.1* Bath: CORE System Trust and CORE Information Management Systems Ltd.
- Cooney, P., Jackman, C., Coyle, D., & O'Reilly, G. (2017). Computerised cognitive– behavioural therapy for adults with intellectual disability: randomised controlled trial. *The British Journal of Psychiatry*, *211*(2), 95-102.
- Cooper, M., & Reeves, A. (2012). The role of randomised controlled trials in developing an evidence-base for counselling and psychotherapy. *Counselling and Psychotherapy Research*, 12(4), 303-307.
- Cox, R. C., & Olatunji, B. O. (2020). Sleep in the anxiety-related disorders: A meta-analysis of subjective and objective research. *Sleep Medicine Reviews*, 51, 101282.
- Craig, P., Dieppe, P., Macintyre, S., Michie, S., Nazareth, I., & Petticrew, M. (2008). Developing and evaluating complex interventions: the new Medical Research Council guidance. *BMJ*, 337.
- Craske, M., & Stein, M. (2016). Anxiety. The Lancet, 388(10063), 3048-3059.
- Cuijpers, P., Van Straten, A., & Warmerdam, L. (2007). Behavioral activation treatments of depression: A meta-analysis. *Clinical Psychology Review*, 27(3), 318-326.
- Cuijpers, P., Marks, I. M., van Straten, A., Cavanagh, K., Gega, L., & Andersson, G. (2009). Computer-aided psychotherapy for anxiety disorders: A meta-analytic review. *Cognitive Behaviour Therapy*, 38(2), 66-82.
- Cuijpers, P., Cristea, I. A., Karyotaki, E., Reijnders, M., & Huibers, M. J. (2016). How effective are cognitive behavior therapies for major depression and anxiety disorders? A meta-analytic update of the evidence. *World Psychiatry*, 15(3), 245-258.

- Cuijpers, P., Karyotaki, E., Reijnders, M., & Huibers, M. J. (2018). Who benefits from psychotherapies for adult depression? A meta-analytic update of the evidence. *Cognitive Behaviour Therapy*, 47(2), 91-106.
- Curcio, C., & Corboy, D. (2020). Stigma and anxiety disorders: A systematic review. *Stigma and Health*, *5*(2), 125.
- Dale, J., Caramlau, I. O., Lindenmeyer, A., & Williams, S. M. (2008). Peer support telephone calls for improving health. *Cochrane Database of Systematic Reviews*, (4).
- De Arcos, F. A., Verdejo-García, A., Ceverino, A., Montañez-Pareja, M., López-Juárez, E., Sánchez-Barrera, M., ... & Pérez-García, M. (2008). Dysregulation of emotional response in current and abstinent heroin users: negative heightening and positive blunting. *Psychopharmacology*, 198(2), 159-166.
- Delgadillo, J., McMillan, D., Lucock, M., Leach, C., Ali, S., & Gilbody, S. (2014). Early changes, attrition, and dose–response in low intensity psychological interventions. *British Journal of Clinical Psychology*, 53(1), 114-130.
- Department of Health (2008). *Improving Access to Psychological Therapies (IAPT) Outcomes Toolkit 2008/9*. London: Department of Health.
- Department of Health. (2012). *IAPT three-year report: The first million patients*. Retrieved from

https://www.mhinnovation.net/sites/default/files/downloads/innovation/reports/Thre e-year-report.pdf

- Department of Health, NHS England, Public Health England (2016). Joint review of partnerships and investment in voluntary, community and social enterprise organisations in the health and care sector: final report produced in partnership by representatives of the VCSE sector and the Department of Health, NHS England and Public Health England. Retrieved from https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachm ent\_data/file/524243/VCSE\_Investment\_Review\_A.pdf
- de Vares, A. (2007). Volunteer-aided Computerised Cognitive Behavioural Therapy (cCBT) Project. *Housing, Care and Support*, 10(1), 6.
- Dillon, Deborah R., David G. O'Brien, and Elizabeth E. Heilman. 2000. Literacy research in the next millennium: From paradigms to pragmatism and practicality. *Reading Research Quarterly*, 35, 10-26
- Donker, T., Bennett, K., Bennett, A., Mackinnon, A., van Straten, A., Cuijpers, P., ... & Griffiths, K. M. (2013). Internet-delivered interpersonal psychotherapy versus internet-delivered cognitive behavioral therapy for adults with depressive symptoms:

randomized controlled non-inferiority trial. *Journal of Medical Internet Research*, *15*(5), e82.

- Donkin, L., Christensen, H., Naismith, S. L., Neal, B., Hickie, I. B., & Glozier, N. (2011). A systematic review of the impact of adherence on the effectiveness of e-therapies. *Journal of Medical Internet Research*, *13*(3), e52.
- Donkin, L., & Glozier, N. (2012). Motivators and Motivations to Persist With Online Psychological Interventions: A Qualitative Study of Treatment Completors. *Journal* of Medical Internet Research, 14(3), 262-273.
- Drozd, F., Vaskinn, L., Bergsund, H. B., Haga, S. M., Slinning, K., & Bjørkli, C. A. (2016).
  The implementation of internet interventions for depression: a scoping review. *Journal of Medical Internet Research*, 18(9), e236.
- Ebert, D. D., Heber, E., Berking, M., Riper, H., Cuijpers, P., Funk, B., & Lehr, D. (2016).
  Self-guided internet-based and mobile-based stress management for employees:
  results of a randomised controlled trial. *Occupational and Environmental Medicine*, 73(5), 315-323.
- Ekers, D., Richards, D., & Gilbody, S. (2008). A meta-analysis of randomized trials of behavioural treatment of depression. *Psychological Medicine*, 38(5), 611-623.
- Elison, S., Humphreys, L., Ward, J., & Davies, G. (2013). A pilot outcomes evaluation for computer-assisted therapy for substance misuse: an evaluation of Breaking Free Online. *Journal of Substance Use*, 19(4), 1-6.
- Elison, S., Ward, J., Davies, G., Lidbetter, N., Hulme, D., & Dagley, M. (2014). An outcomes study of eTherapy for dual diagnosis using Breaking Free Online. Advances in Dual Diagnosis, 7(2), 52-62.
- Elison, S., Ward, J., Williams, C., Espie, C., Davies, G., Dugdale, S., ... & Smith, K. (2017).
  Feasibility of a UK community-based, eTherapy mental health service in Greater
  Manchester: repeated-measures and between-groups study of 'Living Life to the Full
  Interactive', 'Sleepio<sup>™</sup>' and 'Breaking Free Online' at 'Self Help Services.' *BMJ Open*, 7(7), e016392.
- Espie, C. A., Kyle, S. D., Williams, C., Ong, J. C., Douglas, N. J., Hames, P., & Brown, J. S. (2012). A randomized, placebo-controlled trial of online cognitive behavioral therapy for chronic insomnia disorder delivered via an automated media-rich web application. *Sleep*, 35(6), 769-781.
- Etherington, K. (2004). Research methods: Reflexivities-roots, meanings, dilemmas. *Counselling and Psychotherapy Research*, *4*(2), 46-47.

- Etzelmueller, A., Vis, C., Karyotaki, E., Baumeister, H., Titov, N., Berking, M., ... & Ebert, D. D. (2020). Effects of internet-based cognitive behavioral therapy in routine care for adults in treatment for depression and anxiety: Systematic review and meta-analysis. *Journal of Medical Internet Research*, 22(8), e18100.
- Evans, J., Macrory, I., & Randall, C. (2016). *Measuring national wellbeing: Life in the UK*. London: Office for National Statistics.
- Eysenbach, G. (2005). The law of attrition. Journal of Medical Internet Research, 7(1), e11.
- Faber, J., & Fonseca, L. M. (2014). How sample size influences research outcomes. *Dental Press Journal of Orthodontics*, 19(4), 27-29.
- Fairburn, C. G., & Patel, V. (2017). The impact of digital technology on psychological treatments and their dissemination. *Behaviour Research and Therapy*, 88, 19-25.
- Farvolden, P., Denisoff, E., Selby, P., Bagby, R. M., & Rudy, L. (2005). Usage and longitudinal effectiveness of a Web-based self-help cognitive behavioral therapy program for panic disorder. *Journal of Medical Internet Research*, 7(1), e7.
- Feijt, M., de Kort, Y., Bongers, I., Bierbooms, J., Westerink, J., & IJsselsteijn, W. (2020). Mental health care goes online: Practitioners' experiences of providing mental health care during the COVID-19 pandemic. *Cyberpsychology, Behavior, and Social Networking*, 23(12), 860-864.
- Fernández, A., Rubio-Valera, M., Bellón, J. A., Pinto-Meza, A., Luciano, J. V., Mendive, J. M., ... & Serrano-Blanco, A. (2012). Recognition of anxiety disorders by the general practitioner: results from the DASMAP Study. *General Hospital Psychiatry*, 34(3), 227-233.
- Ferrari, A. J., Somerville, A. J., Baxter, A. J., Norman, R., Patten, S. B., Vos, T., & Whiteford, H. A. (2013a). Global variation in the prevalence and incidence of major depressive disorder: a systematic review of the epidemiological literature. *Psychological Medicine*, 43(3), 471-481.
- Field, M. J., & Lo, B. (Eds.). (2009). Conflict of Interest in Medical Research, Education, and Practice. Washington, D.C: National Academies Press.
- Fineberg, N. A., Haddad, P. M., Carpenter, L., Gannon, B., Sharpe, R., Young, A. H., ... & Sahakian, B. J. (2013). The size, burden and cost of disorders of the brain in the UK. *Journal of Psychopharmacology*, 27(9), 761-770.
- Fleet, D., Burton, A., Reeves, A., & DasGupta, M. P. (2016). A case for taking the dual role of counsellor-researcher in qualitative research. *Qualitative Research in Psychology*, 13(4), 328-346.

- Fleming, T., Dixon, R., Frampton, C., & Merry, S. (2012). A pragmatic randomized controlled trial of computerized CBT (SPARX) for symptoms of depression among adolescents excluded from mainstream education. *Behavioural and Cognitive Psychotherapy*, 40(5), 529-541.
- Fleming, T., Bavin, L., Lucassen, M., Stasiak, K., Hopkins, S., & Merry, S. (2018). Beyond the trial: systematic review of real-world uptake and engagement with digital selfhelp interventions for depression, low mood, or anxiety. *Journal of Medical Internet Research*, 20(6), e199.
- Flyvbjerg, B. (2006). Five misunderstandings about case-study research. *Qualitative Inquiry*, *12*(2), 219-245.
- Foley, T., & Woollard, J. (2019). The Digital Future of Mental Healthcare and its Workforce. Retrieved from https://topol.hee.nhs.uk/wp-content/uploads/HEE-Topol-Review-Mental-health-paper.pdf
- Folker, A. P., Mathiasen, K., Lauridsen, S. M., Stenderup, E., Dozeman, E., & Folker, M. P. (2018). Implementing internet-delivered cognitive behavior therapy for common mental health disorders: A comparative case study of implementation challenges perceived by therapists and managers in five European internet services. *Internet Interventions*, *11*, 60-70.
- French, J. R. (1953). Experiments in field settings. *Research Methods in the Behavioral Sciences*, 1953, 98-135.
- Gagnon, M., & Oliffe, J. L. (2015). Male depression and suicide: What NPs should know. *The Nurse Practitioner*, 40(11), 50-55.
- Galdas, P. (2017). Revisiting bias in qualitative research: Reflections on its relationship with funding and impact. *International Journal of Qualitative Methods*, 16(1), 1609406917748992. doi:10.1177/1609406917748992
- Gamsu, M. (2011). "Tell us what the problem is and we'll try to help": Towards more effective commissioning of local voluntary sector organisations. Retrieved from https://archive.vsnw.org.uk/static/files/publications/Tell\_us\_what\_the\_problem\_is.p df
- Gellatly, J., Bower, P., Hennessy, S., Richards, D., Gilbody, S., & Lovell, K. (2007). What makes self-help interventions effective in the management of depressive symptoms? Meta-analysis and meta-regression. *Psychological Medicine*, 37(9), 1217-1228. doi:10.1017/s0033291707000062
- Gellatly, J., Bower, P., McMillan, D., Roberts, C., Byford, S., Bee, P., ... Lovell, K. (2014). Obsessive Compulsive Treatment Efficacy Trial (OCTET) comparing the clinical

and cost effectiveness of self-managed therapies: study protocol for a randomised controlled trial. *Trials*, 15(1), 278.

- Gellatly, J., Chisnall, L., Seccombe, N., Ragan, K., Lidbetter, N., & Cavanagh, K. (2018).
  @ Home eTherapy service for people with common mental health problems: an evaluation. *Behavioural and Cognitive Psychotherapy*, 46(1), 115-120.
- Gilbody, S., Littlewood, E., Hewitt, C., Brierley, G., Tharmanathan, P., Araya, R., ... & Kessler, D. (2015). Computerised cognitive behaviour therapy (cCBT) as treatment for depression in primary care (REEACT trial): large scale pragmatic randomised controlled trial. *BMJ*, 351.
- Glabman, M. (2009). Disruptive innovations that will change your life in health care. *Managed Care*, *18*, 12-21.
- Glasby, J., & Beresford, P. (2006). Commentary and issues: Who knows best? Evidencebased practice and the service user contribution. *Critical Social Policy*, 26(1), 268-284.
- Glasgow, R. E., Phillips, S. M., & Sanchez, M. A. (2014). Implementation science approaches for integrating eHealth research into practice and policy. *International Journal of Medical Informatics*, 83(7), e1-e11.
- Gretton, C., & Honeyman, M. (2016). The digital revolution: eight technologies that will change health and care. The King's Fund. Retrieved from https://www.kingsfund.org.uk/publications/digital-revolution
- Grist, R., & Cavanagh, K. (2013). Computerised cognitive behavioural therapy for common mental health disorders, what works, for whom under what circumstances? A systematic review and meta-analysis. *Journal of Contemporary Psychotherapy*, 43(4), 243-251.
- Grobler, G. (2013). Major depressive disorder: The South African Society of Psychiatrists (SASOP) Treatment Guidelines for Psychiatric Disorders. South African Journal of Psychiatry, 19(3), 157-163.
- Guetterman, T. C. (2019). Basics of statistics for primary care research. *Family Medicine and Community Health*, 7(2).
- Gulland, A. (2017). Items dispensed in England rise sharply in 10 years. *British Medical Journal (Online)*, 357.
- Gupta, S. K. (2011). Intention-to-treat concept: a review. *Perspectives in Clinical Research*, 2(3), 109.

- Gutkin, T. (2012). Ecological Psychology: Replacing the medical model paradigm for school-based psychological and psychoeducational services. *Journal of Educational and Psychological Consultation*, 22(1-2), 1-20.
- Guy, A., Loewenthal, D., Thomas, R., & Stephenson, S. (2012). Scrutinising NICE: The impact of the National Institute for Health and Clinical Excellence Guidelines on the provision of counselling and psychotherapy in primary care in the UK. *Psychodynamic Practice*, 18(1), 25-50.
- Hackman, C., Balhara, Y. P. S., Clayman, K., Nemec, P., Notley, C., Pike, K., ... & Swarbrick, M. (2019). Perspectives on the International Classification of Diseases, 11th Revision (ICD-11); an international qualitative study to Understand and improve mental health Diagnosis using expertise by Experience: INCLUDE Study. *The Lancet Psychiatry*.
- Hadjistavropoulos, H. D., Schneider, L. H., Klassen, K., Dear, B. F., & Titov, N. (2018).
  Development and evaluation of a scale assessing therapist fidelity to guidelines for delivering therapist-assisted Internet-delivered cognitive behaviour therapy. *Cognitive Behaviour Therapy*, 47(6), 447-461.
- Hammen, C. (2005). Stress and depression. Annual Review of Clinical Psychology, 1, 293-319.
- Hariton, E., & Locascio, J. J. (2018). Randomised controlled trials—the gold standard for effectiveness research. BJOG: An International Journal of Obstetrics and Gynaecology, 125(13), 1716.
- Haroz, E. E., Ritchey, M., Bass, J. K., Kohrt, B. A., Augustinavicius, J., Michalopoulos, L.,
  ... & Bolton, P. (2017). How is depression experienced around the world? A systematic review of qualitative literature. *Social Science & Medicine*, *183*, 151-162.
- Health and Social Care Information Centre (HSCIC) (2016). Psychological Therapies: Annual Report on the Use of IAPT Services. England, 2015-16. Retrieved from https://digital.nhs.uk/data-and-information/publications/statistical/psychologicaltherapies-annual-reports-on-the-use-of-iapt-services/annual-report-2015-16
- Health Education England. (2017). Stepping forward to 2020/21: The mental health workforce plan for England. Retrieved from https://www.hee.nhs.uk/ourwork/mental-health
- Health Education England (2020). Peer support workers. Retrieved from https://www.hee.nhs.uk/our-work/mental-health/new-roles-mental-health/peersupport-workers

Health Research Authority. (2017). *Defining Research Table*. Retrieved from http://www.hradecisiontools.org.uk/research/docs/DefiningResearchTable\_Oct2017-1.pdf

- Hedman, E., Ljótsson, B., & Lindefors, N. (2012). Cognitive behavior therapy via the Internet: a systematic review of applications, clinical efficacy and cost– effectiveness. *Expert Review of Pharmacoeconomics & Outcomes Research*, 12(6), 745-764.
- Hedman, E., Ljótsson, B., Kaldo, V., Hesser, H., El Alaoui, S., Kraepelien, M., ... & Lindefors, N. (2014). Effectiveness of Internet-based cognitive behaviour therapy for depression in routine psychiatric care. *Journal of Affective Disorders*, 155, 49-58.
- Hedman, E. El Alaoui, S. Lindefors, N. Andersson, E. Rück, C. Ghaderi, A. et al. (2014).
  Clinical effectiveness and cost-effectiveness of Internet- vs. group-based cognitive behavior therapy for social anxiety disorder: 4-year follow-up of a randomized trial. *Behaviour Research and Therapy*, 59, 20–29.
- Hegeman, J. M., Kok, R. M., Van der Mast, R. C., & Giltay, E. J. (2012). Phenomenology of depression in older compared with younger adults: meta-analysis. The British Journal of Psychiatry, 200(4), 275-281.
- Henn, M., Weinstein, M., & Foard, N. (2005). A short introduction to social research. London: Sage.
- Hobbs, M. J., Mahoney, A. E., & Andrews, G. (2018). Integrating iCBT for generalized anxiety disorder into routine clinical care: treatment effects across the adult lifespan. *Journal of Anxiety Disorders*, 51, 47-54.
- Hofman, J., Pollitt, A., Broeks, M., Stewart, K., & van Stolk, C. (2016). Review of Computerised Cognitive Behavioural Therapies: Products and Outcomes for People with Mental Health Needs. *Rand Health Quarterly*, 6(4).
- Hofmann, S. G., & Hinton, D. E. (2014). Cross-cultural aspects of anxiety disorders. *Current Psychiatry Reports*, *16*(6), 450.
- Holländare, F., Gustafsson, S. A., Berglind, M., Grape, F., Carlbring, P., Andersson, G., ...
  & Tillfors, M. (2016). Therapist behaviours in internet-based cognitive behaviour therapy (ICBT) for depressive symptoms. *Internet Interventions*, *3*, 1-7.
- Hollis, C., Falconer, C. J., Martin, J. L., Whittington, C., Stockton, S., Glazebrook, C., & Davies, E. B. (2017). Annual Research Review: Digital health interventions for children and young people with mental health problems–a systematic and metareview. *Journal of Child Psychology and Psychiatry*, 58(4), 474-503.

- Holmes, E. A., O'Connor, R. C., Perry, V. H., Tracey, I., Wessely, S., Arseneault, L., ... & Ford, T. (2020). Multidisciplinary research priorities for the COVID-19 pandemic: a call for action for mental health science. *The Lancet Psychiatry*, 7(6), 547-560. doi:10.1016/S2215-0366(20)30168-1
- Holst, A., Nejati, S., Björkelund, C., Eriksson, M. C., Hange, D., Kivi, M., ... & Petersson,
  E. L. (2017). Patients' experiences of a computerised self-help program for treating depression–a qualitative study of Internet mediated cognitive behavioural therapy in primary care. *Scandinavian Journal of Primary Health Care*, 35(1), 46-53.
- Horvath, A. O., Del Re, A. C., Flückiger, C., & Symonds, D. (2011). Alliance in individual psychotherapy. *Psychotherapy*, *48*(1), 9.
- IAPT. (2011). The IAPT Data Handbook Guidance on recording and monitoring outcomes to support local evidence-based practice. Version 2. March 2011. Retrieved from https://webarchive.nationalarchives.gov.uk/20160302160058/http://www.iapt.nhs.uk /silo/files/iapt-data-handbook-v2.pdf
- International Council of Nurses (2003). *Ethical Guidelines for Nursing Research*. Geneva: ICN.
- Jackson, C. (2010) Critical Friends, *Openmind* 161, January/February, 14-15.
- Jeffery, A. (2006). Moving away from the medical model. *Veterinary Nursing Journal*, 21(9), 13-16.
- Johnstone, L., Boyle, M., Cromby, J., Dillon, J., Harper, D., Kinderman, P., Longden, E., Pilgrim, D., Read, J. (2018). *The Power Threat Meaning Framework Towards the identification of patterns in emotional distress, unusual experiences and troubled behavior, as an alternative to functional psychiatric diagnosis*. Leicester: The British Psychological Society.
- Jonassaint, C., Belnap, B., Huang, Y., Karp, J., Abebe, K., & Rollman, B. (2019). Racial Differences in the Effectiveness of Internet-Delivered Mental Health Care. *Journal* of General Internal Medicine, 35(2), 490-497.
- Joormann, J., & Stanton, C. H. (2016). Examining emotion regulation in depression: A review and future directions. *Behaviour Research and Therapy*, 86, 35-49.
- Kaltenthaler, E., Shackley, P., Stevens, K., Beverley, C., Parry, G., & Chilcott, J. (2002). A systematic review and economic evaluation of computerised cognitive behaviour therapy for depression and anxiety. In *Database of Abstracts of Reviews of Effects (DARE): Quality-assessed Reviews [Internet]*. York: Centre for Reviews and Dissemination (UK).

- Kaltenthaler, E., Sutcliffe, P., Parry, G., Beverley, C., Rees, A., & Ferriter, M. (2008). The acceptability to patients of computerized cognitive behaviour therapy for depression: a systematic review. *Psychological Medicine*, *38*(11), 1521-1530.
- Kampmann, I. L., Emmelkamp, P. M., & Morina, N. (2016). Meta-analysis of technologyassisted interventions for social anxiety disorder. *Journal of Anxiety Disorders*, 42, 71-84.
- Kara, S., Yazici, K. M., Güleç, C., & Ünsal, I. (2000). Mixed anxiety–depressive disorder and major depressive disorder: comparison of the severity of illness and biological variables. *Psychiatry Research*, 94(1), 59-66.
- Karyotaki, E., Riper, H., Twisk, J., Hoogendoorn, A., Kleiboer, A., Mira, A., ... & Andersson, G. (2017). Efficacy of self-guided internet-based cognitive behavioral therapy in the treatment of depressive symptoms: a meta-analysis of individual participant data. *JAMA Psychiatry*, 74(4), 351-359.
- Karyotaki, E., Ebert, D., Donkin, L., Riper, H., Twisk, J., Burger, S., . . . Buntrock, C. (2018). Do guided internet-based interventions result in clinically relevant changes for patients with depression? An individual participant data meta-analysis. *Clinical Psychology Review*, 63, 80-92.
- Kaushik, V., & Walsh, C. A. (2019). Pragmatism as a research paradigm and its implications for social work research. *Social Sciences*, 8(9), 255.
- Kellett, S., & Matthews, L. (2008). Reflecting on the clinical, organisational and methodological implications of the Doncaster IAPT site: a commentary on Richards and Suckling. In *Clinical Psychology Forum - New Series* (Vol. 181, p. 28). Leicester: British Psychological Society.
- Kendall, T., Pilling, S., Glover, N., & Taylor, C. (2011). Guidelines in mental health– National and international perspectives. *International Review of Psychiatry*, 23(4), 314-317. doi:10.3109/09540261.2011.607431
- Kenning, C., Blakemore, A., Bower, P., Safari, M., Cuijpers, P., Brown, J. S., ... & Waheed, W. (2019). Preventing depression in the community by voluntary sector providers (PERSUADE): intervention development and protocol for a parallel randomised controlled feasibility trial. *BMJ Open*, 9(10), e023791.
- Kessler, R. C., Berglund, P., Demler, O., Jin, R., Merikangas, K. R., & Walters, E. E. (2005). Lifetime prevalence and age-of-onset distributions of DSM-IV disorders in the National Comorbidity Survey Replication. *Archives of General Psychiatry*, 62(6), 593-602.

- Kessler, R. C., Ruscio, A. M., Shear, K., & Wittchen, H. U. (2009). Epidemiology of anxiety disorders. In M. B. Stein, & T. Steckler (Eds.), *Behavioral Neurobiology of Anxiety and its Treatment* (pp. 21-35). Berlin: Springer.
- Kessler, R. C., & Bromet, E. J. (2013). The epidemiology of depression across cultures. *Annual Review of Public Health*, 34, 119-138.
- Kewley, E., & McBride, D. L. (2013). Ethical dilemmas that arise when mental health professionals use client date for research purposes. Retrieved from https://opus.uleth.ca/handle/10133/5281
- Khadjesari, Z., Murray, E., Kalaitzaki, E., White, I. R., McCambridge, J., Thompson, S. G.,
  ... & Godfrey, C. (2011). Impact and costs of incentives to reduce attrition in online
  trials: two randomized controlled trials. *Journal of Medical Internet Research*, 13(1),
  e26.
- Kim, H. S., Lee, S., & Kim, J. H. (2018). Real-world evidence versus randomized controlled trial: clinical research based on electronic medical records. *Journal of Korean Medical Science*, 33(34).
- Kleiboer, A., Donker, T., Seekles, W., van Straten, A., Riper, H., & Cuijpers, P. (2015). A randomized controlled trial on the role of support in Internet-based problem solving therapy for depression and anxiety. *Behaviour Research and Therapy*, 72, 63-71.
- Klein, R. G., & Pine, D. S. (2001). Anxiety disorders. In M. Rutter, E. Taylor, & M. Hersov (Eds.), *Child and Adolescent Psychiatry* (3rd ed.). New York: Blackwell Scientific.
- Klein, D. N. (2008). Classification of depressive disorders in the DSM-V: Proposal for a two-dimension system. *Journal of Abnormal Psychology*, 117(3), 552
- Klein, B., Austin, D., Pier, C., Kiropoulos, L., Shandley, K., Mitchell, J., ... & Ciechomski, L. (2009). Internet-based treatment for panic disorder: Does frequency of therapist contact make a difference?. *Cognitive Behaviour Therapy*, 38(2), 100-113.
- Klein, J. P., Gerlinger, G., Knaevelsrud, C., Bohus, M., Meisenzahl, E., Kersting, A., ... & Härter, M. (2016). Internet-based interventions in the treatment of mental disorders: overview, quality criteria, perspectives. *Der Nervenarzt*, 87(11), 1185-1193.
- Knight, T., & Thomas, P. (2019). Anxiety and depression in the age of austerity: Public health's problems with IAPT. *Perspectives in Public Health*, 139(3), 128-130.
- Knol, M. J., Twisk, J. W., Beekman, A. T., Heine, R. J., Snoek, F. J., & Pouwer, F. (2006). Depression as a risk factor for the onset of type 2 diabetes mellitus. A metaanalysis. *Diabetologia*, 49(5), 837.

- Knowles, S. E., Toms, G., Sanders, C., Bee, P., Lovell, K., Rennick-Egglestone, S., ... & Gilbody, S. (2014). Qualitative meta-synthesis of user experience of computerised therapy for depression and anxiety. *PLoS One*, 9(1).
- Knowles, S., Lovell, K., Bower, P., Gilbody, S., Littlewood, E., & Lester, H. (2015). Patient experience of computerised therapy for depression in primary care. *BMJ Open*, 5(11), E008581
- Krebs, P., Prochaska, J. O., & Rossi, J. S. (2010). A meta-analysis of computer-tailored interventions for health behavior change. *Preventive Medicine*, *51*(3-4), 214-221.
- Kroenke, K., & Spitzer, R. L., Williams, J. B. (2001). The PHQ-9: validity of a brief depression severity measure. *Journal of General Internal Medicine*, 16(9), 606-613.
- Kroenke, K., Spitzer, R. L., Williams, J. B., Monahan, P. O., & Löwe, B. (2007). Anxiety disorders in primary care: prevalence, impairment, comorbidity, and detection. *Annals of Internal Medicine*, 146(5), 317-325.
- Kuehner, C. (2017). Why is depression more common among women than among men?. *The Lancet Psychiatry*, 4(2), 146-158.
- Lachin, J. M. (2016). Fallacies of last observation carried forward analyses. *Clinical Trials*, *13*(2), 161-168.
- Latkin, C. A., Edwards, C., Davey-Rothwell, M. A., & Tobin, K. E. (2017). The relationship between social desirability bias and self-reports of health, substance use, and social network factors among urban substance users in Baltimore, Maryland. *Addictive Behaviors*, 73, 133-136.
- Lattie, E. G., Adkins, E. C., Winquist, N., Stiles-Shields, C., Wafford, Q. E., & Graham, A. K. (2019). Digital mental health interventions for depression, anxiety, and enhancement of psychological well-being among college students: systematic review. *Journal of Medical Internet Research*, 21(7), e12869.
- Laurie, N., & Bondi, L. (Eds.). (2006). Working the Spaces of Neoliberalism: activism, professionalisation and incorporation. Oxford: Wiley-Blackwell.
- Learmonth, D., Trosh, J., Rai, S., Sewell, J., & Cavanagh, K. (2008). The role of computeraided psychotherapy within an NHS CBT specialist service. *Counselling and Psychotherapy Research*, 8(2), 117-123.
- Leichsenring, F., & Rabung, S. (2008). Effectiveness of long-term psychodynamic psychotherapy: A meta-analysis. *JAMA*, *300*(13), 1551-1565.
- Leykin, Y., & DeRubeis, R. J. (2009). Allegiance in psychotherapy outcome research: Separating association from bias. *Clinical Psychology: Science and Practice*, 16, 54–65.

Lidbetter, N., (2003). The stigmatisation of anxiety disorders. In A. Crisp (Ed.). *Every Family in the Land* (pp. 52-53). London: Royal Society of Medicine Press Ltd.

- Lidbetter, N., & Bunnell, D. (2013). Self Help Services: helping people to help themselves. *Mental Health and Social Inclusion*, *17*(2), 76-81.
- Lidbetter, N., & O'Neill, C. (2010). Anxiety: falling between primary and secondary care. *Mental Health Today (Brighton, England)*, 17.

Lidbetter, N. (2020a). Understanding Anxiety: An Anxiety UK self help guide. Manchester: Anxiety UK. Retrieved from https://www.anxietyuk.org.uk/products/anxiety-condition/anxiety/newunderstanding-anxiety-booklet-instant-download/

- Lidbetter, N. (2020b). The Caregivers Guide to Anxiety. Manchester: Anxiety UK. Retrieved from https://www.anxietyuk.org.uk/products/anxietycondition/anxiety/the-caregivers-guide-to-anxiety-instant-download/
- Lincoln, Y. S. (1985). *Lincoln and Guba's evaluative criteria. Qualitative research guidelines project: Robert Wood Johnson Foundation*. Retrieved from http://www.qualres.org/HomeLinc-3684.html
- Liu, X., & Jiang, K. (2016). Why is diagnosing MDD challenging?. Shanghai Archives of Psychiatry, 28(6), 343.
- Lorenzo-Luaces, L., Johns, E., & Keefe, J. R. (2018). The generalizability of randomized controlled trials of self-guided internet-based cognitive behavioral therapy for depressive symptoms: systematic review and meta-regression analysis. *Journal of Medical Internet Research*, 20(11), e10113.
- Lovell, K., & Bee, P. (2008). Implementing the NICE OCD/BDD guidelines. *Psychology and Psychotherapy: Theory, Research and Practice*, 81(4), 365-376.
- Lovell, K., Bower, P., Richards, D., Barkham, M., Sibbald, B., Roberts, C., ... & Hennessy,
  S. (2008). Developing guided self-help for depression using the Medical Research
  Council complex interventions framework: a description of the modelling phase and
  results of an exploratory randomised controlled trial. *BMC Psychiatry*, 8(1), 91.
- Lovell, K., Bower, P., Gellatly, J., Byford, S., Bee, P., McMillan, D., ... Roberts, C. (2017a). Clinical effectiveness, cost-effectiveness and acceptability of low-intensity interventions in the management of obsessive-compulsive disorder: the Obsessive-Compulsive Treatment Efficacy randomised controlled Trial (OCTET). *Health Technology Assessment*, 21(37), 1-132.
- Lovell, K., Bower, P., Gellatly, J., Byford, S., Bee, P., McMillan, D., ... & Reynolds, S. (2017b). Low-intensity cognitive-behaviour therapy interventions for obsessive-

compulsive disorder compared to waiting list for therapist-led cognitive-behaviour therapy: 3-arm randomised controlled trial of clinical effectiveness. *PLoS Medicine*, *14*(6).

- Lubelczyk, R. (2015). Detoxification or supervised withdrawal. In: Trestman R, Appelbaum K, eds. Oxford Textbook of Correctional Psychiatry. Oxford: Oxford University Press.
- Luborsky, L., Singer, B., & Luborsky, L. (1975). Comparative studies of psychotherapies. Is it true that "everyone has won and all must have prizes"? Archives of General Psychiatry, 32, 995–1008.
- Luik, A. I., Bostock, S., Chisnall, L., Kyle, S. D., Lidbetter, N., Baldwin, N., & Espie, C. A. (2017). Treating depression and anxiety with digital cognitive behavioural therapy for insomnia: a real world NHS evaluation using standardized outcome measures. *Behavioural and Cognitive Psychotherapy*, 45(1), 91-96.
- Luppino, F. S., de Wit, L. M., Bouvy, P. F., Stijnen, T., Cuijpers, P., Penninx, B. W., & Zitman, F. G. (2010). Overweight, obesity, and depression: a systematic review and meta-analysis of longitudinal studies. *Archives of General Psychiatry*, 67(3), 220-229.
- Lustria, M. L. A., Noar, S. M., Cortese, J., Van Stee, S. K., Glueckauf, R. L., & Lee, J. (2013). A meta-analysis of web-delivered tailored health behavior change interventions. *Journal of Health Communication*, 18(9), 1039-1069.
- Ly, K. H., Trüschel, A., Jarl, L., Magnusson, S., Windahl, T., Johansson, R., ... & Andersson, G. (2014). Behavioural activation versus mindfulness-based guided selfhelp treatment administered through a smartphone application: a randomised controlled trial. *BMJ Open*, 4(1), e003440.
- Mahlke, C. I., Priebe, S., Heumann, K., Daubmann, A., Wegscheider, K., & Bock, T.
  (2017). Effectiveness of one-to-one peer support for patients with severe mental illness–a randomised controlled trial. *European Psychiatry*, 42, 103-110.
- Malouff, J. M., Thorsteinsson, E. B., & Schutte, N. S. (2007). The efficacy of problem solving therapy in reducing mental and physical health problems: a metaanalysis. *Clinical Psychology Review*, 27(1), 46-57.
- Mansell, W. (2008). What is CBT really and how can we enhance the impact of effective psychotherapies such as CBT? In R. House & D. Loewenthal (Eds.). *Against and For CBT: Towards and Constructive Dialogue* (pp. 19-32). Ross-on-Wye: PCCS Books.

- Margison, F., Barkham, M., Evans, C., McGrath, G., Mellor-Clark, J., Audin, K., & Connell, J. (2000). Measurement and psychotherapy: evidence-based practice and practice-based evidence. *British Journal of Psychiatry*, 177, 123-130.
- Marino, C., Child, B., & Campbell Krasinski, V. (2016). Sharing Experience Learned First hand (SELF): Self-disclosure of lived experience in mental health services and supports. *Psychiatric Rehabilitation Journal*, 39(2), 154.
- Marks, I., Shaw, S., & Parkin, R. (1998). Computer-aided treatments of mental health problems. *Clinical Psychology: Science and Practice*, 5(2), 151-170.
- Marks, I. M., Kenwright, M., McDonough, M., Whittaker, M., & Mataix-Cols, D. (2004).
  Saving clinicians' time by delegating routine aspects of therapy to a computer: a randomized controlled trial in phobia/panic disorder. *Psychological Medicine*, *34*(1), 9-17.
- Marks, I. M., Cavanagh, K., & Gega, L. (2007). *Hands-on help: Computer-aided Psychotherapy*. London: Psychology Press.
- Marks, I., & Cavanagh, K. (2009). Computer-aided psychological treatments: evolving issues. *Annual Review of Clinical Psychology*, *5*, 121-141.
- Marshall, D., Quinn, C., Child, S., Shenton, D., Pooler, J., Forber, S., & Byng, R. (2016).
  What IAPT services can learn from those who do not attend. *Journal of Mental Health*, 25(5), 410-415.
- Martín-Merino, E., Ruigómez, A., Wallander, M. A., Johansson, S., & García-Rodríguez, L.
   A. (2010). Prevalence, incidence, morbidity and treatment patterns in a cohort of patients diagnosed with anxiety in UK primary care. *Family Practice*, 27(1), 9-16.
- Mason, R., & Reeves, A. (2018). An exploration of how working in the Improving Access to Psychological Therapies (IAPT) programme might affect the personal and professional development of counsellors: an analytical autoethnographic study. *British Journal of Guidance & Counselling*, 46(6), 669-678.
- Mataix-Cols, D., Cameron, R., Gega, L., Kenwright, M., & Marks, I. M. (2006). Effect of referral source on outcome with cognitive-behavior therapy selfhelp. *Comprehensive Psychiatry*, 47(4), 241-245.
- Mathers, C. D., & Loncar, D. (2006). Projections of global mortality and burden of disease from 2002 to 2030. *Plos Medicine*, *3*(11), e442.
- Maxcy, S. J. (2003). Pragmatic threads in mixed methods research in the social sciences: The search for multiple modes of inquiry and the end of the philosophy of formalism. In L. Jason & D. Glenwick (Eds.). *Handbook of Mixed Methods in Social and Behavioral Research* (pp. 51-89). New York: Oxford University Press.

- McHugh, M. L. (2013). The chi-square test of independence. *Biochemia medica: Biochemia medica*, 23(2), 143-149.
- McLeod, J. (2011). Qualitative Research in Counselling and Psychotherapy. London: Sage.
- McManus, S., Bebbington, P., Jenkins, R., & Brugha, T. (2016). Mental health and wellbeing in England: Adult Psychiatric Morbidity Survey 2014. A survey carried out for NHS Digital by Nat Cen Social Research and the Department of Health Sciences, University of Leicester. Retrieved from https://www.basw.co.uk/resources/mental-health-and-wellbeing-england-adultpsychiatric-morbidity-survey-2014
- McMillan, D., Bee, P., Lidbetter, N., & Lukoseviciute, B. (2020). Telephone therapy is convenient and it works. Let's use it more. Retrieved from https://psyche.co/ideas/telephone-therapy-is-convenient-and-it-works-lets-use-itmore
- McPherson S., Evans C. & Richardson P. (2009) The NICE depression guidelines and the recovery model: is there an evidence base for IAPT? *Journal of Mental Health 18*, 405–414.
- Melartin, T. K., Rytsälä, H. J., Leskelä, U. S., Lestelä-Mielonen, P. S., Sokero, T. P., & Isometsä, E. T. (2002). Current comorbidity of psychiatric disorders among DSM-IV major depressive disorder patients in psychiatric care in the Vantaa Depression Study. *The Journal of Clinical Psychiatry*, 63(2), 126–134. doi:10.4088/JCP.v63n0207
- Melville, K. M., Casey, L. M., & Kavanagh, D. J. (2010). Dropout from Internet-based treatment for psychological disorders. *British Journal of Clinical Psychology*, 49(4), 455-471.
- Men's Health Forum (2020). Key data: mental health. Retrieved from http://www.menshealthforum.org.uk/key-data-mental-health.
- Mergl, R., Seidscheck, I., Allgaier, A. K., Möller, H. J., Hegerl, U., & Henkel, V. (2007). Depressive, anxiety, and somatoform disorders in primary care: prevalence and recognition. *Depression and Anxiety*, 24(3), 185-195.
- Mertens, D. M. (2005). *Research methods in education and psychology: Integrating diversity with quantitative and qualitative approaches* (2<sup>nd</sup> ed.). Thousand Oaks: Sage.

- Mewton, L., Sachdev, P. S., & Andrews, G. (2013). A naturalistic study of the acceptability and effectiveness of internet-delivered cognitive behavioural therapy for psychiatric disorders in older Australians. *PloS One*, 8(8).
- Meyer, B., Berger, T., Caspar, F., Beevers, C., Andersson, G., & Weiss, M. (2009).
  Effectiveness of a novel integrative online treatment for depression (Deprexis): randomized controlled trial. *Journal of Medical Internet Research*, *11*(2), e15.
- Mirza, I., & Jenkins, R. (2004). Risk factors, prevalence, and treatment of anxiety and depressive disorders in Pakistan: systematic review. *BMJ*, *328*(7443), 794.
- Mobbs, D., Hagan, C. C., Dalgleish, T., Silston, B., & Prévost, C. (2015). The ecology of human fear: survival optimization and the nervous system. *Frontiers in Neuroscience*, 9, 55.
- Mohr, D., Cuijpers, P., & Lehman, K. (2011). Supportive accountability: a model for providing human support to enhance adherence to eHealth interventions. *Journal of Medical Internet Research*, 13(1), e30.
- Mohr, D. C., Cheung, K., Schueller, S. M., Brown, C. H., & Duan, N. (2013a). Continuous evaluation of evolving behavioral intervention technologies. *American Journal of Preventive Medicine*, 45(4), 517-523.
- Mohr, D. C., Duffecy, J., Ho, J., Kwasny, M., Cai, X., Burns, M. N., & Begale, M. (2013b).
   A randomized controlled trial evaluating a manualized TeleCoaching protocol for improving adherence to a web-based intervention for the treatment of depression. *PloS One*, 8(8).
- Moll, S., Holmes, J., Geronimo, J., & Sherman, D. (2009). Work transitions for peer support providers in traditional mental health programs: unique challenges and opportunities. *Work*, 33(4), 449-458.
- Moller, N. P., Ryan, G., Rollings, J., & Barkham, M. (2019). The 2018 UK NHS Digital annual report on the Improving Access to Psychological Therapies programme: a brief commentary. *BMC Psychiatry*, 19(1), 252.
- Mollon, P. (2008). IAPT- an apt approach?. Psychologist, 21(10), 895-896.
- Mols, F., Schoormans, D., de Hingh, I., Oerlemans, S., & Husson, O. (2018). Symptoms of anxiety and depression among colorectal cancer survivors from the populationbased, longitudinal PROFILES Registry: Prevalence, predictors, and impact on quality of life. *Cancer*, 124(12), 2621-2628. doi:10.1002/cncr.31369
- Moore, G. F., Audrey, S., Barker, M., Bond, L., Bonell, C., Hardeman, W., ... & Baird, J. (2015). Process evaluation of complex interventions: Medical Research Council guidance. *BMJ*, 350.

- Moule, P., Armoogum, J., Dodd, E., Donskoy, A. L., Douglass, E., Taylor, J., & Turton, P. (2016). Practical guidance on undertaking a service evaluation. *Nursing Standard*, 30(45), 46-51
- Mundt, J. C., Marks, I. M., Shear, M. K., & Greist, J. M. (2002). The Work and Social Adjustment Scale: a simple measure of impairment in functioning. *The British Journal of Psychiatry*, 180(5), 461-464.
- Musiat, P., Goldstone, P., & Tarrier, N. (2014). Understanding the acceptability of e-mental health-attitudes and expectations towards computerised self-help treatments for mental health problems. *BMC Psychiatry*, 14(1), 109.
- National Collaborating Centre for Mental Health (UK. (2010). *Depression: the treatment and management of depression in adults (updated edition)*. Leicester: British Psychological Society.
- National Collaborating Centre for Mental Health (Great Britain), National Institute for Health, Clinical Excellence (Great Britain), British Psychological Society, & Royal College of Psychiatrists. (2011). *Common Mental Health Disorders: identification and pathways to care* (Vol. 123). London: RCPsych Publications.
- National Collaborating Centre for Mental Health. (2018). *The Improving Access to Psychological Therapies Manual*. UK: NCCMH.
- National Institute for Clinical Excellence (2002). Guidance on the use of computerised cognitive behavioural therapy for anxiety and depression. Technology appraisal 51.
   London: NICE.
- National Institute for Clinical Excellence (2004a). *Anxiety: management of anxiety (panic disorder, with or without agoraphobia, and generalised anxiety disorder) in adults in primary, secondary and community care.* London: NICE.
- National Institute for Clinical Excellence (2004b). *Depression: the management of depression in Primary and Secondary Care.* London: NICE.
- National Institute for Clinical Excellence (2006). *Computerised cognitive behaviour therapy for depression and anxiety (Review of Technology Appraisal 51).* London: NICE.
- National Institute for Health and Care Excellence. (2009). Depression: the treatment and management of depression in adults (update). Clinical guidelines, CG90. Retrieved from https://www.nice.org.uk/guidance/cg90
- National Institute for Health and Care Excellent. (2011). Common mental health problems: identification and pathways to care. Retrieved from

https://www.nice.org.uk/guidance/cg123/resources/common-mental-healthproblems-identification-and-pathways-to-care-pdf-35109448223173

- National Institute for Health and Care Excellence. (2018). Antidepressant treatment in adults. Retrieved from https://pathways.nice.org.uk/pathways/depression
- National Institute for Health and Care Excellence. (2019a). Evidence standards framework for digital health technologies. London: National Institute for Health and Care Excellence.
- National Institute for Health and Care Excellence. (2019b). *National Institute for Health And Care Excellence Medicines and Technologies Programme Process and methods statement for the production of IAPT evaluation in practice reports*. Retrieved from https://www.nice.org.uk/Media/Default/About/what-we-do/NICEadvice/IAPT/IEPR-process-methods.pdf
- National Institute for Health and Care Excellence (2020a). *Digital therapies assessed and accepted by the Improving Access to Psychological Therapies Programme (IAPT)*. Retrieved from https://www.nice.org.uk/Media/Default/About/what-we-do/NICE-advice/IAPT/IAB-space-from-depression.pdf
- National Institute for Health and Care Excellence (2020b). Space from Depression for treating adults with depression. Retrieved from https://www.nice.org.uk/advice/mib215/resources/space-from-depression-fortreating-adults-with-depression-pdf-2285965453227973
- Newby, J. M., Twomey, C., Li, S. S. Y., & Andrews, G. (2016). Transdiagnostic computerised cognitive behavioural therapy for depression and anxiety: a systematic review and meta-analysis. *Journal of Affective Disorders*, 199, 30-41.
- Newman, M. G., Szkodny, L. E., Llera, S. J., & Przeworski, A. (2011). A review of technology-assisted self-help and minimal contact therapies for anxiety and depression: is human contact necessary for therapeutic efficacy? *Clinical Psychology Review*, 31(1), 89-103.
- Newbigging, K., Mohan, J., Rees, J., Harlock, J., & Davis, A. (2017). Contribution of the voluntary sector to mental health crisis care in England: protocol for a multimethod study. *BMJ Open*, *7*(11), e019238.
- NHS Confederation Mental Health Network. (2013). E-mental health: What's all the fuss about? Retrieved from https://www.nhsconfed.org/resources/2013/01/e-mental-health-whats-all-the-fuss-about

- NHS Digital (2015). Improving Access to Psychological Therapies Report, March 2015 Final. Retrieved from https://digital.nhs.uk/data-therapies-report-on-the-use-of-iaptservices/march-2015-final
- NHS Digital (2017). Improving Access to Psychological Therapies (IAPT) Executive Summary (January 2017). Retrieved from https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachm ent\_data/file/610567/iapt-month-jan-2017-exec-sum.pdf
- NHS Digital. (2020). Improving access to psychological therapies data set. Retrieved from https://digital.nhs.uk/data-and-information/data-collections-and-data-sets/data-sets/improving-access-to-psychological-therapies-data-set
- NHS England. (2014). *Five Year Forward View*. Retrieved from https://www.england.nhs.uk/wp-content/uploads/2014/10/5yfv-web.pdf
- NHS England (2019). Long Term Plan. Retrieved from https://www.longtermplan.nhs.uk/wp-content/uploads/2019/08/nhs-long-term-planversion-1.2.pdf
- NHS England. (2020a). Digital therapy selection. Retrieved from https://www.england.nhs.uk/mental-health/adults/iapt/digital-therapy-selection/
- NHS England (2020b). Adult improving access to psychological therapies. Retrieved from https://www.england.nhs.uk/mental-health/adults/iapt/
- Nilsson, T., Svensson, M., Sandell, R., & Clinton, D. (2007). Patients' experiences of change in cognitive–behavioral therapy and psychodynamic therapy: a qualitative comparative study. *Psychotherapy Research*, 17(5), 553-566.
- Noar, S. M., Benac, C. N., & Harris, M. S. (2007). Does tailoring matter? Meta-analytic review of tailored print health behavior change interventions. *Psychological bulletin*, 133(4), 673.
- Norcross, J. C., & Lambert, M. J. (2018). Psychotherapy relationships that work III. *Psychotherapy*, 55(4), 303.
- Nordahl-Hansen, A., Øien, R. A., Volkmar, F., Shic, F., & Cicchetti, D. V. (2018). Enhancing the understanding of clinically meaningful results: A clinical research perspective. *Psychiatry Research*, 270, 801-806.
- Nordgreen, L. B., Carlbring, P., Linna, E., & Andersson, G. (2013). Role of the working alliance on treatment outcome in tailored internet-based cognitive behavioural therapy for anxiety disorders: randomized controlled pilot trial. JMIR Research Protocols, *2*(1), e4.

- Oakley, A., Strange, V., Bonell, C., Allen, E., & Stephenson, J. (2006). Process evaluation in randomised controlled trials of complex interventions. *BMJ*, *332*(7538), 413-416.
- Odierna, D. H., Forsyth, S. R., White, J., & Bero, L. A. (2013). The cycle of bias in health research: a framework and toolbox for critical appraisal training. *Accountability in Research*, 20(2), 127-141.
- Olthuis, J. V., Watt, M. C., Bailey, K., Hayden, J. A., & Stewart, S. H. (2016). Therapistsupported Internet cognitive behavioural therapy for anxiety disorders in adults. *Cochrane Database of Systematic Reviews*, (3).
- Omylinska-Thurston, J., McMeekin, A., Walton, P., & Proctor, G. (2019). Clients' perceptions of unhelpful factors in CBT in IAPT serving a deprived area of the UK. *Counselling and Psychotherapy Research*, *19*(4), 455-464.
- Ormandy, P., & Hulme, C. (2013). Measuring patients' preferences and priorities for information in chronic kidney disease. *Information Research*, *18*(3) 588.
- Palpacuer, C., Gallet, L., Drapier, D., Reymann, J. M., Falissard, B., & Naudet, F. (2017).
  Specific and non-specific effects of psychotherapeutic interventions for depression:
  Results from a meta-analysis of 84 studies. *Journal of Psychiatric Research*, 87, 95-104.
- Păsărelu, C. R., Andersson, G., Bergman Nordgren, L., & Dobrean, A. (2017). Internetdelivered transdiagnostic and tailored cognitive behavioral therapy for anxiety and depression: a systematic review and meta-analysis of randomized controlled trials. *Cognitive Behaviour Therapy*, 46(1), 1-28.
- Patino, C. M., & Ferreira, J. C. (2018). Internal and external validity: can you apply research study results to your patients?. *Jornal Brasileiro de Pneumologia*, 44(3), 183-183.
- Patton, M. Q. (1990). *Qualitative evaluation and research methods* (2nd ed.). Newbury Park, CA: Sage.
- Paxling, B., Lundgren, S., Norman, A., Almlöv, J., Carlbring, P., Cuijpers, P., & Andersson, G. (2013). Therapist behaviours in internet-delivered cognitive behaviour therapy: analyses of e-mail correspondence in the treatment of generalized anxiety disorder. *Behavioural and Cognitive Psychotherapy*, 41(3), 280-289.
- Paul, G. L. (1967). Strategy of outcome research in psychotherapy. *Journal of Consulting Psychology*, 31(2), 109.
- Paulhus, D. L. (1984). Two-component models of socially desirable responding. *Journal of Personality and Social Psychology*, 46(3), 598.

- Peck, D. F. (2010). The therapist–client relationship, computerized self-help and active therapy ingredients. *Clinical Psychology & Psychotherapy: An International Journal* of Theory & Practice, 17(2), 147-153.
- Perera-Delcourt, R. P., & Sharkey, G. (2019). Patient experience of supported computerized CBT in an inner-city IAPT service: a qualitative study. *The Cognitive Behaviour Therapist*, 12.
- Perrier, C. P., Etchegary, H., Palarchio, N., & Snelgrove, P. (2009). The ethics of applied research with individuals with addictions. *Journal of Ethics in Mental Health*, *4*(2), 1–6.
- Pfeiffer, P. N., Heisler, M., Piette, J. D., Rogers, M. A., & Valenstein, M. (2011). Efficacy of peer support interventions for depression: a meta-analysis. *General Hospital Psychiatry*, 33(1), 29-36.
- Pham, Q., Wiljer, D., & Cafazzo, J. A. (2016). Beyond the randomized controlled trial: a review of alternatives in mHealth clinical trial methods. *JMIR mHealth and uHealth*, *4*(3), e107.
- Pickersgill, M. (2019). Access, accountability, and the proliferation of psychological therapy: On the introduction of the IAPT initiative and the transformation of mental healthcare. *Social Studies of Science*, *49*(4), 627-650.
- Pihlaja, S., Stenberg, J. H., Joutsenniemi, K., Mehik, H., Ritola, V., & Joffe, G. (2018). Therapeutic alliance in guided internet therapy programs for depression and anxiety disorders–a systematic review. *Internet Interventions*, 11, 1-10.
- Polit, D. F., & Beck, C. T. (2009). Essentials of nursing research: Appraising evidence for nursing practice (7th ed.). Philadelphia: Lippincott Williams & Wilkins.
- Pope, C., & Mays, N. (2006). *Qualitative research in health care* (3rd ed.). Malden, Mass.: Blackwell Pub.
- Posternak, M. A., & Miller, I. (2001). Untreated short-term course of major depression: a meta-analysis of outcomes from studies using wait-list control groups. *Journal of Affective Disorders*, 66(2-3), 139-146.
- Pots, W. T., Fledderus, M., Meulenbeek, P. A., Peter, M., Schreurs, K. M., & Bohlmeijer, E. T. (2016). Acceptance and commitment therapy as a web-based intervention for depressive symptoms: randomised controlled trial. *The British Journal of Psychiatry*, 208(1), 69-77.
- Powell, J. E., & Taylor, D. (1991). Anger, depression and anxiety following heroin withdrawal. *International Journal of the Addictions*, 27(1), 25-35.

- Price, J. S. (2003). Evolutionary aspects of anxiety disorders. *Dialogues in Clinical Neuroscience*, *5*(3), 223.
- Proudfoot, J., Clarke, J., Birch, M. R., Whitton, A. E., Parker, G., Manicavasagar, V., ... & Hadzi-Pavlovic, D. (2013). Impact of a mobile phone and web program on symptom and functional outcomes for people with mild-to-moderate depression, anxiety and stress: a randomised controlled trial. *BMC Psychiatry*, *13*(1), 312.
- Proudfoot, J., Goldberg, D., Mann, A., Everitt, B., Marks, I., & Gray, J. A. (2003).
  Computerized, interactive, multimedia cognitive-behavioural program for anxiety and depression in general practice. *Psychological Medicine*, *33*(2), 217-227.
- Proudfoot, J., Ryden, C., Everitt, B., Shapiro, D. A., Goldberg, D., Mann, A., ... & Gray, J.
  A. (2004). Clinical efficacy of computerised cognitive-behavioural therapy for anxiety and depression in primary care: randomised controlled trial. *The British Journal of Psychiatry*, 185(1), 45-54.
- Pybis, J., Saxon, D., Hill, A., & Barkham, M. (2017). The comparative effectiveness and efficiency of cognitive behaviour therapy and generic counselling in the treatment of depression: evidence from the 2nd UK National Audit of psychological therapies. *BMC Psychiatry*, 17(1), 215.
- Qin, X., Phillips, M. R., Wang, W., Li, Y., Jin, Q., Ai, L., ... & Liu, L. (2010). Prevalence and rates of recognition of anxiety disorders in internal medicine outpatient departments of 23 general hospitals in Shenyang, China. *General Hospital Psychiatry*, 32(2), 192-200.
- Rachman, S. (2015). The evolution of behaviour therapy and cognitive behaviour therapy. *Behaviour Research and Therapy*, 64, 1-8.
- Ray, J. M., Kemp, L. L., Hubbard, A., & Cucciare, M. A. (2017). Developing a peer support protocol for improving veterans' engagement to computer-delivered cognitive behavioural therapy. *Behavioural and Cognitive Psychotherapy*, 45(3), 253-265.
- Rennick-Egglestone, S., Knowles, S., Toms, G., Bee, P., Lovell, K., & Bower, P. (2016, May). Health Technologies' In the Wild' Experiences of Engagement with Computerised CBT. In *Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems* (pp. 2124-2135). doi: 10.1145/2858036.2858128
- Repper, J., & Carter, T. (2010). Using personal experience to support others with similar difficulties. A Review of the Literature on Peer Support in Mental Health Services Together for Mental Wellbeing. The University of Nottingham.
- Richards, D., Duffy, D., Blackburn, B., Earley, C., Enrique, A., Palacios, J., ... & Timulak,L. (2018). Digital IAPT: the effectiveness & cost-effectiveness of internet-delivered

interventions for depression and anxiety disorders in the Improving Access to Psychological Therapies programme: study protocol for a randomised control trial. *BMC Psychiatry*, *18*(1), 59.

- Richards, D., Murphy, T., Viganó, N., Timulak, L., Doherty, G., Sharry, J., & Hayes, C. (2016). Acceptability, satisfaction and perceived efficacy of "Space from Depression" an internet-delivered treatment for depression. *Internet Interventions*, 5, 12-22.
- Richards, D., & Richardson, T. (2012). Computer-based psychological treatments for depression: a systematic review and meta-analysis. *Clinical Psychology Review*, 32(4), 329-342.
- Richards, D., Timulak, L., O'Brien, E., Hayes, C., Vigano, N., Sharry, J., & Doherty, G.
  (2015). A randomized controlled trial of an internet-delivered treatment: its potential as a low-intensity community intervention for adults with symptoms of depression. *Behaviour Research and Therapy*, 75, 20-31.
- Richards, D. A., Bower, P., Pagel, C., Weaver, A., Utley, M., Cape, J., ... & Owens, L. (2012). Delivering stepped care: an analysis of implementation in routine practice. *Implementation Science*, 7(1), 3.
- Richards, I. L., Moores, E., Witton, C., Reddy, P. A., Rippon, G., Rochelle, K. S., & Talcott, J. B. (2003). Science, sophistry and 'commercial sensitivity': Comments on 'evaluation of an exercise-based treatment for children with reading difficulties,' by Reynolds, Nicolson and Hambly. *Dyslexia*, 9(3), 146-150.
- Ritterband, L. M., Gonder-Frederick, L. A., Cox, D. J., Clifton, A. D., West, R. W., & Borowitz, S. M. (2003). Internet interventions: In review, in use, and into the future. *Professional Psychology: Research and Practice*, 34(5), 527.
- Robinson, L. A., Berman, J. S., & Neimeyer, R. A. (1990). Psychotherapy for the treatment of depression: a comprehensive review of controlled outcome research. *Psychological Bulletin*, 108(1), 30.
- Robinson, E., Titov, N., Andrews, G., McIntyre, K., Schwencke, G., & Solley, K. (2010). Internet treatment for generalized anxiety disorder: a randomized controlled trial comparing clinician vs. technician assistance. *PloS One*, 5(6), e10942.
- Robson, C., & McCartan, K. (2016). *Real world research* (4<sup>th</sup> ed.). New York: John Wiley & Sons.
- Roddis, J. K., Liversedge, H. L., Ryder, I., & Woodhouse, M. (2019). Incorporating the patient experience into clinical guidelines: recommendations for researchers and guideline developers. *BMJ Evidence-based Medicine*, 24(4), 125-126.

- Rodriguez-Pulido, F., Castillo, G., Hamrioui, S., Martin, L. D., Vazquez-Beltrán, P., de la Torre-Díez, I., & Franco-Martín, M. A. (2020). Treatment of Depression in Primary Care with Computerized Psychological Therapies: Systematic Reviews. *Journal of Medical Systems*, 44(3), 67. doi:10.1007/s10916-020-1543-7
- Rogers, E. (2003). Diffusion of innovations (5th ed.). London: Simon & Schuster.
- Rogers, E. S., Chamberlin, J., & Ellison, M. L. (1997). Measure empowerment among users of mental health services. *Psychiatric Services*, *48*(8), 1042-1047.
- Rogers E.S., Teague, G.B., Lichenstein, C., Campbell, J., Lyass, A., Chen, R., et al. (2007).
   Effects of participation in consumer-operated service programs on both personal and organizationally mediated empowerment: results of multisite study. *Journal of Rehabilitation Research Development*, 44(6), 785-799
- Rost, T., Stein, J., Löbner, M., Kersting, A., Luck-Sikorski, C., & Riedel-Heller, S. G.
  (2017). User acceptance of computerized cognitive behavioral therapy for depression: systematic review. *Journal of Medical Internet Research*, 19(9), e309.
- Roy-Byrne, P., Craske, M. G., Sullivan, G., Rose, R. D., Edlund, M. J., Lang, A. J., ... & Campbell-Sills, L. (2010). Delivery of evidence-based treatment for multiple anxiety disorders in primary care: a randomized controlled trial. *JAMA*, 303(19), 1921-1928.
- Rozental, A., Andersson, G., Boettcher, J., Ebert, D. D., Cuijpers, P., Knaevelsrud, C., ... & Carlbring, P. (2014). Consensus statement on defining and measuring negative effects of Internet interventions. *Internet Interventions*, 1(1), 12-19.
- Rozental, A., Boettcher, J., Andersson, G., Schmidt, B., & Carlbring, P. (2015). Negative effects of internet interventions: a qualitative content analysis of patients' experiences with treatments delivered online. *Cognitive Behaviour Therapy*, 44(3), 223-236.
- Rush, A. J. (2007). The varied clinical presentations of major depression disorder. *The Journal of Clinical Psychiatry*, 68(Suppl. 8), 4–10.
- Rushton, K., Ardern, K., Hopkin, E., Welsh, C., Gellatly, J., Faija, C., ... & Bee, P. (2020).
  'I didn't know what to expect': Exploring patient perspectives to identify targets for change to improve telephone-delivered psychological interventions. *BMC Psychiatry*, 20, 1-13.
- Rusticus, S. A., & Lovato, C. Y. (2014). Impact of sample size and variability on the power and type I error rates of equivalence tests: A simulation study. *Practical Assessment, Research, and Evaluation, 19*(1), 11.

- Rycroft-Malone, J., Seers, K., Titchen, A., Harvey, G., Kitson, A., & McCormack, B. (2004). What counts as evidence in evidence-based practice?. *Journal of Advanced Nursing*, 47(1), 81-90.
- Sample, I. (2020, 16 September). Stress, anxiety and depression levels soar under UK Covid-19 restrictions. *The Guardian*. Retrieved from https://www.theguardian.com/society/2020/sep/16/stress-anxiety-and-depressionlevels-soar-under-uk-covid-19-restrictions
- Saunders, J. B., Aasland, O. G., Babor, T. F., De la Fuente, J. R., & Grant, M. (1993). Development of the alcohol use disorders identification test (AUDIT): WHO collaborative project on early detection of persons with harmful alcohol consumption-II. *Addiction*, 88(6), 791-804.
- Scanlon, C., & Adlam, J. (2010). The Recovery Model or the modelling of a coverup? *Groupwork*, 20(3), 100-114.
- Schneider, A. J., Mataix-Cols, D., Marks, I. M., & Bachofen, M. (2005). Internet-guided self-help with or without exposure therapy for phobic and panic disorders. *Psychotherapy and psychosomatics*, 74(3), 154-164.
- Schmidt, I. D., Forand, N. R., & Strunk, D. R. (2019). Predictors of dropout in Internetbased cognitive behavioral therapy for depression. *Cognitive Therapy and Research*, 43(3), 620-630.
- Schneider, L. H., Hadjistavropoulos, H. D., & Faller, Y. N. (2016). Internet-delivered cognitive behaviour therapy for depressive symptoms: an exploratory examination of therapist behaviours and their relationship to outcome and therapeutic alliance. *Behavioural and Cognitive Psychotherapy*, 44(6), 625.
- Schröder, J., Berger, T., Meyer, B., Lutz, W., Späth, C., Michel, P., ... & Moritz, S. (2018). Impact and change of attitudes toward Internet interventions within a randomized controlled trial on individuals with depression symptoms. *Depression and Anxiety*, 35(5), 421-430.
- Scogin, F. R., Hanson, A., & Welsh, D. (2003). Self-administered treatment in stepped-care models of depression treatment. *Journal of Clinical Psychology*, 59(3), 341-349.
- Seale, B. (2016). Patients as partners. Building collaborative relationships among professionals, patients, carers and communities. London: The King's Fund.
- Self Help Services (2021). Your information: how it is used and how to gain access to it. Retrieved from https://www.selfhelpservices.org.uk/privacy/
- Serrano-Blanco, A., Palao, D. J., Luciano, J. V., Pinto-Meza, A., Luján, L., Fernández, A., ... & Haro, J. M. (2010). Prevalence of mental disorders in primary care: results from

the diagnosis and treatment of mental disorders in primary care study (DASMAP). *Social Psychiatry and Psychiatric Epidemiology*, 45(2), 201-210.

- Shandley, K., Austin, D., Klein, B., Pier, C., Schattner, P., Pierce, D., & Wade, V. (2008). Therapist-assisted, internet-based treatment for panic disorder: can general practitioners achieve comparable patient outcomes to psychologists? *Journal of Medical Internet Research*, 10(2), e14.
- Sharry, J., Davidson, R., McLoughlin, O., & Doherty, G. (2013). A Service-Based Evaluation of a Therapist-Supported Online Cognitive Behavioral Therapy Program for Depression. *Journal of Medical Internet Research*, 15(6), E121.
- Shinohara, K., Honyashiki, M., Imai, H., Hunot, V., Caldwell, D. M., Davies, P., ... & Churchill, R. (2013). Behavioural therapies versus other psychological therapies for depression. *Cochrane Database of Systematic Reviews*, (10).
- Sijbrandij, M., Kunovski, I., & Cuijpers, P. (2016). Effectiveness of internet-delivered cognitive behavioral therapy for posttraumatic stress disorder: A systematic review and meta-analysis. *Depression and Anxiety*, *33*(9), 783-791.
- Silfvernagel, K., Carlbring, P., Kabo, J., Edström, S., Eriksson, J., Månson, L., & Andersson, G. (2012). Individually tailored internet-based treatment for young adults and adults with panic attacks: randomized controlled trial. *Journal of Medical Internet Research*, 14(3), e65.
- Simmonds-Buckley, M., Bennion, M. R., Kellett, S., Millings, A., Hardy, G. E., & Moore, R. K. (2020). Acceptability and Effectiveness of NHS-Recommended e-Therapies for Depression, Anxiety, and Stress: Meta-Analysis. *Journal of Medical Internet Research*, 22(10), e17049.
- Simon, G. E., Katon, W. J., VonKorff, M., Unützer, J., Lin, E. H., Walker, E. A., ... & Ludman, E. (2001). Cost-effectiveness of a collaborative care program for primary care patients with persistent depression. *American Journal of Psychiatry*, 158(10), 1638-1644.
- Simon, G. E., & Ludman, E. J. (2009). It's time for disruptive innovation in psychotherapy. *The Lancet*, 374(9690), 594-595. doi:10.1016/s0140-6736(09)61415-x
- Simons, H. (2009). Case study research in practice. London: SAGE publications.
- Slavich, G. M., & Irwin, M. R. (2014). From stress to inflammation and major depressive disorder: a social signal transduction theory of depression. *Psychological Bulletin*, 140(3), 774.

- Slawson, N. (2016, 24 May). Nicky Lidbetter: 'My anxiety has been a motivator.' *The Guardian*. Retrieved from https://www.theguardian.com/society/2016/may/24/icky-lidbetter-my-anxiety-has-been-motivator-mental-health
- So, M., Yamaguchi, S., Hashimoto, S., Sado, M., Furukawa, T. A., & McCrone, P. (2013). Is computerised CBT really helpful for adult depression? A meta-analytic reevaluation of CCBT for adult depression in terms of clinical implementation and methodological validity. *BMC Psychiatry*, 13(1), 113.
- Solomon, D. A., Keller, M. B., Leon, A. C., Mueller, T. I., Lavori, P. W., Shea, M. T., ... & Endicott, J. (2000). Multiple recurrences of major depressive disorder. *American Journal of Psychiatry*, 157(2), 229-233.
- Solomon, P. (2004). Peer support/peer provided services underlying processes, benefits, and critical ingredients. *Psychiatric Rehabilitation Journal*, 27(4), 392.
- Somers, J. M., Goldner, E. M., Waraich, P., & Hsu, L. (2006). Prevalence and incidence studies of anxiety disorders: a systematic review of the literature. *The Canadian Journal of Psychiatry*, 51(2), 100-113.
- Smith, J., & Noble, H. (2014). Bias in research. Evidence-based Nursing, 17(4), 100-101.
- Spijkerman, M. P. J., Pots, W. T. M., & Bohlmeijer, E. T. (2016). Effectiveness of online mindfulness-based interventions in improving mental health: A review and metaanalysis of randomised controlled trials. *Clinical Psychology Review*, 45, 102-114.
- Spitzer, R. L., Kroenke, K., Williams, J. B., & Löwe, B. (2006). A brief measure for assessing generalized anxiety disorder: the GAD-7. Archives of Internal Medicine, 166(10), 1092-1097.

Stake, R. E. (2005). Qualitative case studies. In N. K. Denzin and Y. S. Lincoln. Handbook of Qualitative Research (3rd edn). Thousand Oaks. CA: Sage.

- Steel, Z., Marnane, C., Iranpour, C., Chey, T., Jackson, J. W., Patel, V., & Silove, D.
  (2014). The global prevalence of common mental disorders: a systematic review and meta-analysis 1980–2013. *International Journal of Epidemiology*, 43(2), 476-493.
- Steimer T. (2002). The biology of fear and anxiety-related behaviors. *Dialogues in clinical neuroscience*, *4*(3), 231–249.
- Stevanovic, D., Jancic, J., & Lakic, A. (2011). The impact of depression and anxiety disorder symptoms on the health-related quality of life of children and adolescents with epilepsy. *Epilepsia*, 52(8), e75-e78.
- Sweeney, A., Fahmy, S., Nolan, F., Morant, N., Fox, Z., Lloyd-Evans, B., ... & Slade, M. (2014). The relationship between therapeutic alliance and service user satisfaction in

mental health inpatient wards and crisis house alternatives: a cross-sectional study. *PLoS One*, *9*(7), e100153.

- Sweet, T (2019). Third sector perspective: Procurement: the case for collaboration. *Healthcare Counselling and Psychotherapy*. Page 3.
- Teachman, B. A. (2014). No appointment necessary: Treating mental illness outside the therapist's office. *Perspectives on Psychological Science*, *9*(1), 85-87.
- Terry, N., & Reeves, A. (2015). How do counsellors and psychotherapists understand diet and nutrition as part of the therapy process? A heuristic study. *Counselling and Psychotherapy Research*, 15(4), 309-319.
- Thew, G. R. (2020). IAPT and the internet: the current and future role of therapist-guided internet interventions within routine care settings. *The Cognitive Behaviour Therapist*, 13.
- Third Sector (2020). Charities have a role in public services, but not as surrogate businesses. Retrieved from https://www.thirdsector.co.uk/charities-role-public-services-not-surrogate-businesses/management/article/1455038
- Thompson, C., Kinmonth, A. L., Stevens, L., Pevele, R. C., Stevens, A., Ostler, K. J., ... & Cooper, D. (2000). Effects of a clinical-practice guideline and practice-based education on detection and outcome of depression in primary care: Hampshire Depression Project randomised controlled trial. *The Lancet*, 355(9199), 185-191.
- Thornicroft, G. (2007). Most people with mental illness are not treated. *The Lancet*, 370(9590), 807–808. doi: 10.1016/S0140-6736(07)61392-0
- Thornicroft, G. (2018). Improving access to psychological therapies in England. *The Lancet*, *391*(10121), 636-637.
- Titov, N., Andrews, G., Schwencke, G., Solley, K., Johnston, L., & Robinson, E. (2009). An RCT comparing effect of two types of support on severity of symptoms for people completing Internet-based cognitive behaviour therapy for social phobia. *Australian* & New Zealand Journal of Psychiatry, 43(10), 920-926.
- Titov, N., Andrews, G., Davies, M., McIntyre, K., Robinson, E., & Solley, K. (2010). Internet treatment for depression: a randomized controlled trial comparing clinician vs. technician assistance. *PloS One*, 5(6), e10939.
- Titov, N., Dear, B. F., Schwencke, G., Andrews, G., Johnston, L., Craske, M. G., & McEvoy, P. (2011). Transdiagnostic internet treatment for anxiety and depression: a randomised controlled trial. *Behaviour Research and Therapy*, 49(8), 441-452.
- Titov, N., Fogliati, V. J., Staples, L. G., Gandy, M., Johnston, L., Wootton, B., ... & Dear,B. F. (2016). Treating anxiety and depression in older adults: randomised controlled

trial comparing guided v. self-guided internet-delivered cognitive-behavioural therapy. *BJPsych Open*, *2*(1), 50-58.

- Toft, T., Fink, P. E. R., Oernboel, E. V. A., Christensen, K. A. J., Frostholm, L., & Olesen, F. (2005). Mental disorders in primary care: prevalence and co-morbidity among disorders. Results from the functional illness in primary care (FIP) study. *Psychological Medicine*, 35(8), 1175-1184.
- Tolentino, J. C., & Schmidt, S. L. (2018). DSM-5 criteria and depression severity: implications for clinical practice. *Frontiers in Psychiatry*, 9, 450.
- Tomasino, K. N., Lattie, E. G., Ho, J., Palac, H. L., Kaiser, S. M., & Mohr, D. C. (2017). Harnessing peer support in an online intervention for older adults with depression. *The American Journal of Geriatric Psychiatry*, 25(10), 1109-1119.
- Topooco, N., Riper, H., Araya, R., Berking, M., Brunn, M., Chevreul, K., ... & Kleiboer, A. (2017). Attitudes towards digital treatment for depression: a European stakeholder survey. *Internet Interventions*, 8, 1-9.
- Treweek, S., & Zwarenstein, M. (2009). Making trials matter: pragmatic and explanatory trials and the problem of applicability. *Trials*, *10*(1), 1-9.
- Tyrer, P. (2001). The case for cothymia: mixed anxiety and depression as a single diagnosis. *The British Journal of Psychiatry*, *179*(3), 191-193.
- Tyrer, P. (2009). Are general practitioners really unable to diagnose depression?. *The Lancet*, 9690(374), 589-590.
- Twomey, C., O'Reilly, G., Byrne, M., Bury, M., White, A., Kissane, S., ... & Clancy, N.
  (2014). A randomized controlled trial of the computerized CBT programme,
  MoodGYM, for public mental health service users waiting for interventions. *British Journal of Clinical Psychology*, *53*(4), 433-450.
- Twomey, C., O'Reilly, G., & Byrne, M. (2015). Effectiveness of cognitive behavioural therapy for anxiety and depression in primary care: a meta-analysis. *Family Practice*, 32(1), 3-15.
- Van Ballegooijen, W., Cuijpers, P., Van Straten, A., Karyotaki, E., Andersson, G., Smit, J.
  H., & Riper, H. (2014). Adherence to Internet-based and face-to-face cognitive behavioural therapy for depression: a meta-analysis. *PloS One*, 9(7), e100674.
- Van der Zanden, R., Kramer, J., Gerrits, R., & Cuijpers, P. (2012). Effectiveness of an online group course for depression in adolescents and young adults: a randomized trial. *Journal of Medical Internet research*, 14(3), e86.
- van Stolk, C., Hofman, J., Hafner, M., & Janta, B. (2014). Psychological wellbeing and work: improving service provision and outcomes. *Rand Health Quarterly*, *4*(1).

- Van Straten, A., Hill, J., Richards, D. A., & Cuijpers, P. (2015). Stepped care treatment delivery for depression: a systematic review and meta-analysis. *Psychological Medicine*, 45(2), 231-246.
- Vasilica, C., & Ormandy, P. (2017). Methods for studying information provision, networking and communication in patient support groups. In C. Urquhart, F. Hamad, D. Tbaishat, & A. Yeoman (Eds.), *Information Systems: Process and Practice* (pp. 205-233). London: Facet Publishing.
- Vermani, M., Marcus, M., & Katzman, M. A. (2011). Rates of detection of mood and anxiety disorders in primary care: a descriptive, cross-sectional study. *The Primary Care Companion to CNS Disorders*, *13*(2), PCC.10m01013. doi:10.4088/PCC.10m01013
- Wade, D. T., & Halligan, P. W. (2017). The biopsychosocial model of illness: a model whose time has come. *Clinical Rehabilitation*, 31(8), 995-1004. doi:10.1177/0269215517709890
- Waller, R., & Gilbody, S. (2009). Barriers to the uptake of computerized cognitive behavioural therapy: a systematic review of the quantitative and qualitative evidence. *Psychological Medicine*, 39(5), 705-712.
- Wang, P. S., Angermeyer, M., Borges, G., Bruffaerts, R., Chiu, W. T., De Girolamo, G., ...
  & Kessler, R. C. (2007). Delay and failure in treatment seeking after first onset of mental disorders in the World Health Organization's World Mental Health Survey Initiative. *World Psychiatry*, 6(3), 177.
- Weaver, T., Madden, P., Charles, V., Stimson, G., Renton, A., Tyrer, P., ... & Paterson, S. (2003). Comorbidity of substance misuse and mental illness in community mental health and substance misuse services. *The British Journal of Psychiatry*, 183(4), 304-313.
- Wells, A., & Fisher, P. (Eds.). (2015). Treating Depression: MCT, CBT, and Third Wave Therapies. New York: John Wiley & Sons.
- Whitfield, G., & Williams, C. (2004). If the evidence is so good–why doesn't anyone use them? A national survey of the use of computerized cognitive behaviour therapy. *Behavioural and Cognitive Psychotherapy*, 32(1), 57-65.
- Widdowson, M. D. J. (2011). Case study research methodology. International Journal of Transactional Analysis Research, 2(1), 25-34.
- Wilhelm, S., Weingarden, H., Ladis, I., Braddick, V., Shin, J., & Jacobson, N. C. (2020). Cognitive-behavioral therapy in the digital age: presidential address. *Behavior Therapy*, 51(1), 1-14.

Williams, C. (n.d.). Living Life to the Full. Retrieved from https://llttf.com/

- Williams, C. H. J. (2015). Improving Access to Psychological Therapies (IAPT) and treatment outcomes: Epistemological Assumptions and Controversies. *Journal of Psychiatric and Mental Health Nursing*, 22(5), 344-351.
- Wims, E., Titov, N., & Andrews, G. (2008). The Climate Panic program: an open trial of Internet-based treatment for panic disorder. *E-Journal of Applied Psychology*, 4(2).
- Wootton, B. M., Titov, N., Dear, B. F., Spence, J., & Kemp, A. (2011). The acceptability of Internet-based treatment and characteristics of an adult sample with obsessive compulsive disorder: an Internet survey. *PLoS One*, 6(6).
- World Health Organization. (1993). International Classification of Diseases ICD 10 Classification of Mental and Behavioural Disorders: Diagnostic Criteria for Research. Geneva: World Health Organization.
- World Health Organization. (2017a). Depression and other common mental disorders: global health estimates (No. WHO/MSD/MER/2017.2). Geneva: World Health Organization
- World Health Organization. (2017b). ICD-10 online versions. Retrieved from https://www.who.int/classifications/icd/icdonlineversions/en/
- World Health Organization. (2018a). International classification of diseases for mortality and morbidity statistics (11th Revision). Retrieved from https://icd.who.int/browse11/l-m/en
- World Health Organisation. (2018b). Classification of digital health interventions v1. 0: a shared language to describe the uses of digital technology for health (No. WHO/RHR/18.06). Geneva: World Health Organization.
- Wright, J. H., Wright, A. S., Albano, A. M., Basco, M. R., Goldsmith, L. J., Raffield, T., & Otto, M. W. (2005). Computer-assisted cognitive therapy for depression: maintaining efficacy while reducing therapist time. *American Journal of Psychiatry*, *162*(6), 1158-1164.
- Wright, J. H., Mishkind, M., Eells, T. D., & Chan, S. R. (2019). Computer-Assisted Cognitive-Behavior Therapy and Mobile Apps for Depression and Anxiety. *Current Psychiatry Reports*, 21(7), 62.
- Yanos, P. T., & Ziedonis, D. M. (2006). The patient-oriented clinician-researcher: advantages and challenges of being a double agent. *Psychiatric Services*, 57(2), 249-253.
- Yardley, L., & Bishop, F. L. (2015). Using mixed methods in health research: Benefits and challenges. *British Journal of Health Psychology*, 20(1), 1-4.

- Yin, R. K. (2006). Mixed methods research: Are the methods genuinely integrated or merely parallel? *Research in the Schools*. *13*(1), 41-47.
- Yin, R. K. (2009). Case study research: Design and methods. London: SAGE.
- Zhai, L., Zhang, Y., & Zhang, D. (2015a). Sedentary behaviour and the risk of depression: a meta-analysis. *British Journal of Sports Medicine*, 49(11), 705-709.
- Zhai, L., Zhang, H., & Zhang, D. (2015b). Sleep duration and depression among adults: A meta-analysis of prospective studies. *Depression and Anxiety*, 32(9), 664-670.
Appendices

# Appendix 1

# Papers included in the Portfolio of Published Works

## Paper One

Cavanagh, K., Seccombe, N., Lidbetter, N., & Bunnell, D. (2011a). Supported, service-user led, computerised cognitive behavioural therapy (CCBT) self-help clinics. *Journal of Public Mental Health*, 10(4), 225-233. <u>https://doi.org/10.1108/17465721111188241</u>

## Paper Two

Cavanagh, K., Seccombe, N., & Lidbetter, N. (2011b). The Implementation of Computerized Cognitive Behavioural therapies in a Service User-Led, Third Sector Self Help clinic. *Behavioural and Cognitive Psychotherapy*, 39(4), 427-442. <u>https://doi.org/10.1017/S1352465810000858</u>

## Paper Three

Lidbetter, N., & Bunnell, D. (2013). Self Help Services: helping people to help themselves. *Mental Health and Social Inclusion*, 17(2), 76-81. <u>https://doi.org/10.1108/20428301311330126</u>

## **Paper Four**

Elison, S., Ward, J., Davies, G., Lidbetter, N., Hulme, D., & Dagley, M. (2014). An outcomes study of eTherapy for dual diagnosis using Breaking Free Online. *Advances in Dual Diagnosis*, 7(2), 52-62. <u>https://doi.org/10.1108/ADD-11-2013-0025</u>

**Paper Five** 

## Research

# BMJ Open Feasibility of a UK community-based, eTherapy mental health service in **Greater Manchester: repeated-measures** and between-groups study of 'Living Life to the Full Interactive', 'Sleepio' and 'Breaking Free Online' at 'Self Help Services'

Sarah Elison,<sup>1</sup> Jonathan Ward,<sup>1</sup> Chris Williams,<sup>2</sup> Colin Espie,<sup>3</sup> Glyn Davies,<sup>1</sup> Stephanie Dugdale,<sup>1</sup> Kathryn Ragan,<sup>4</sup> Leanne Chisnall,<sup>4</sup> Nicky Lidbetter,<sup>4</sup> Keith Smith<sup>4</sup>

#### ARSTRACT

Objectives There is increasing evidence to support the effectiveness of eTherapies for mental health, although limited data have been reported from community-based services. Therefore, this service evaluation reports on feasibility and outcomes from an eTherapy mental health service

Setting 'Self Help Services', an Increasing Access to Psychological Therapies (APT) oTherapy service in Greater Manchester,

Participants 1068 service users referred to the service for secondary care for their mental health difficulties. Interventions Participants were triaged into one of three eTherapy programmes: "Living Life to the Full Interactive" for low mood, stress and anxiety; "Sleepio' for insomnia; and 'Breaking Free Online' for substance misuse, depending on clinical need.

Primary outcomes measures Standardised psychometric assessments of depression, arodety and social functioning, collected as part of the IAPT Minimum Data Set, were conducted at baseline and post-treatment. Results Data indicated baseline differences, with the Breaking Free Online group having higher scores for depression and arodety than the Living Life to the Full Interactive (depression Cl 1.27 to 3.21, p<0.0001; anxiety CI 077 to 1.72, p<0.0001) and Sleepio (depression CI 1.19 to 4.52, p<0.0001; anxiety CI 2.16 to 5.23, p<0.0001) groups. Promising improvements in mental health scores were found within all three groups (all p<0.0001), as were significant reductions in numbers of service users reaching clinical threshold scores for mental health difficulties (p<0.0001). Number of days of engagement was not related to change from baseline for the Living Life to the Full or Sleepio programmes but was associated with degree of change for Breaking Free Online. Conclusion Data presented provide evidence for feasibility of this eTherapy delivery model in supporting service users with a range of mental health difficulties

### Strengths and limitations of this study

- Large sample size
- Data from community-based mental health service and service users
- High ecological validity Examines feasibility of a novel treatment modality within a community-based mental health service
- Outcomes have implications for reducing waiting list times for mental bealth services
- Unequal sample sizes across three eTherapy groups Lack of control/comparison group Lack of randomisation to treatment arms
- Lack of longer-term follow-up data

and suggest that eTherapies may be a useful addition to treatment offering in community-based services.

### INTRODUCTION

For individuals in the UK with mental health difficulties, waiting list times remain a significant barrier to accessing psychosocial support.1 Data indicate that since the 2008 recession, funding for menual health provision in some regions of the UK has been cut by as much as 32%,2 a trend that has occurred alongside increasing prevalence of common mental health difficulties such as anxiety and depression3 and increased demand for mental health service.4 Despite the introduction of the Increasing Access to Psychological Therapies (IAPT) programme in England, which was intended to alleviate pressures on primary mental health services,5 as many as 1 in 10

To cite: Elison S, Ward J, Williams C, of at. Feasibility of a UK community-based, eThorapy mental health service in Greator Manchester: repeated measures and between-groups study of 'Living Life to the Full Interactive', 'Sloepio' and 'Breaking Free Online' at 'Self Help Services', BMJ Open 2017;7:0016392. dot 10.1136/ bmjopen-2017-016392

 Prepublication history for this paper is available online. To view these files please visit the journal online (http://dx.doi. org/10.1136/bmjopen-2017-0163921

Received 13 February 2017 Revised 9 May 2017 Accepted 7 June 2017



Breaking Free Group, Manchesler, UK <sup>3</sup>Institute of Health and Wellbeing, University of Glasgow, UK "Nutfield Department of Clinical Neurosciences, University of Oxford, UK Self Holp Services, UK

Correspondence to Dr. Sarah Elison; selison@ breakingfreegroup.com



Elison S, et al. BMJ Open 2017;7:e016392. doi:10.1136/bmjogen-2017-016392

patients may still be waiting for over a year for face-to-face psychosocial treatment.<sup>6</sup> The IAPT approach is designed to widen access to lower-intensity interventions for mild to moderate depression and anxiety, which are delivered by specially trained Psychological Wellbeing Practitioners (PWPs). These PWPs work alongside higher-intensity mental health professionals such as high-intensity therapists and Clinical Psychologists, who provide support to individuals with more complex needs.<sup>7</sup> Although the IAPT approach has reduced waiting list times within primary mental health services, IAPT services themselves are now facing significant oversubscription.<sup>6</sup>

One possible solution to the growing issue of waiting lists in mental health services are 'computer-assisted therapies' (CATs) or 'cTherapies.' These therapies deliver evidencebased psychosocial interventions and behavioural change techniques through digital technologies such as web and mobile applications. Such interventions have the potential to deliver highly individualised treatment, by tailoring intervention content to the specific needs of the individual, and there is now a growing literature demonstrating the clinical and cost-effectiveness of tailorable eTherapy programmes for the treatment of a wide range of health difficulties.<sup>89</sup>

Three examples of such eTherapy programmes are 'Living Life to the Full Interactive' (www.llttf. com/).<sup>1011</sup> a programme for low mood, stress and anxiety, and depression associated with physical health problems; 'Sleepio' (www.sleepio.com),<sup>12-45</sup> a sleep improvement programme; and 'Breaking Free Online' (www.breakingfreeonline.com),16-20 a programme that helps people overcome substance misuse difficulties. All three of these programmes have been delivered to service users via an innovative eTherapy service in Greater Manchester, 'Self Help Services', a service-user-led mental health charity that provides primary care mental health services across the north of England. A number of National Health Service (NHS) Trusts have commissioned services provided by the charity as part of the IAPT initiative, and therefore, this study reports on the feasibility of the delivery of these three eTherapies by Self Help Services in community-based mental health treatment settings, via the use of clinical outcomes data collected for service evaluation purposes in the IAPT Minimum Data Set.

#### Advantages and disadvantages of delivering eTherapies for mental health

There are a number of advantages to providing cTherapies, such as significantly reducing waiting times within healthcare services and being more cost-effective than one-to-one therapy, given that multiple users can access an cTherapy simultaneously.<sup>21-23</sup> Providing interventions as eTherapy also ensures optimal treatment fidelity as therapeutic techniques are delivered using a computer in a highly standardised manner,<sup>21,24</sup> without the human-related variance in delivery often seen in traditional human-facilitated interventions.<sup>29,25</sup> However, despite these advantages of delivering cTherapies, there are still some limitations with the approach. Some studies have demonstrated adherence to be low, with numbers of service users dropping out of treatment being high.<sup>27-29</sup> Additionally, when such interventions are provided with minimal or no practitioner support, there is little opportunity for a positive therapeutic alliance to be built between the service user and and a practitioner, which may reduce effectiveness.<sup>30,51</sup>

The introduction of digital health interventions such as eTherapies into existing healthcare systems is also often perceived as 'disruptive', meaning it can take considerable time for such innovations to be implemented and incorporated into standard practice.<sup>32-54</sup> Additionally, research has demonstrated that information technology infrastructure within NHS mental health services may be inadequate for the effective delivery of eTherapies.<sup>3</sup> Nevertheless, a recent report by the King's Fund highlighted the potential of eTherapies as part of effective practice,<sup>37</sup> with some eTherapy programmes having been demonstrated to be tailorable to the need of the individual and clinically effective, with the National Institute for Health and Care Excellence now recommending such approaches for anxiety and depression.38 This is due to the fact that despite difficulties with implementation, some eTherapies have crossed the divide between research-based innovation and implementation in clinical settings, including those eTherapies delivered at Self Help Services, namely Living Life to the Full Interactive, Sleepio and Breaking Free Online.

## Mental health eTherapy provision at Self Help Services

Living Life to the Full Interactive<sup>10</sup> <sup>11</sup> is a licensed eTherapy programme demonstrated as being effective in helping people cope with low mood, stress and anxiety via the inclusion of techniques informed by cognitive behaviour therapy (CBT) principles.<sup>20 40</sup> The programme is an online interactive self-help skills package, comprising a number of modules covering areas of life and wellbeing commonly affected by low mood and stress. It uses both interactive text and video formats to accommodate different learning styles and provides psychoeducation alongside more practical CBT techniques such as relaxation and guidance on how to make life changes.

Sleepio is an online sleep improvement programme demonstrated as being effective in helping people with insomnia,<sup>12-15</sup> which can be used as a self-help programme. It comprises intervention techniques informed by CBT principles and provides users with 6 weeks of access to tailored clinical content, and 12 weeks of support from an online community. The programme includes 10 online tools and a 'library' of articles written by sleep experts, in addition to a personal 12-week sleep diary. Users are encouraged to log on once a week for a personalised 20 min session with an avatar that guides them through a personalised programme.

Breaking Free Online is an online treatment programme that has a growing evidence base to support its effectiveness in helping people overcome difficulties

Elison S. at al. BMJ Open 2017:7:e016392. doi:10.1136/bmiopen-2017-016392

BMJ Open: first published as 10.1138/bmjopen-2017-016392 on 20 July 2017. Downloaded from http: ul/bmjopen.bmj.com/ on 4 March 2019 by guest. Protected by copyright

3

with alcohol and drugs,<sup>16-20</sup> which can be delivered as CAT or self-help, and targets 39 different substances, including alcohol. The programme provides multiple interactive psychosocial interventions, drawing on CBT, mindfulness and relapse prevention techniques, via a six-domain biopsychosocial model, the Lifestyle Balance Model,<sup>41</sup> which conceptualises various aspects of functioning associated with substance misuse and comorbid mental health difficulties.

Although the three eTherapy programmes provided by Self Help Services each have different primary clinical targets, they all contain cognitive-behavioural interventions that are likely to be generally helpful to individuals in addressing underlying anxiogenic or depressogenic thinking and unhelpful behaviour patterns. For example, Living Life to the Full Interactive was developed specifically for individuals experiencing low mood, stress and anxiety but contains clinical techniques that could also be effective for addressing associated issues such as poor sleep. Additionally, previous research conducted at Self Help Services with individuals using Breaking Free Online for their substance misuse demonstrated significant improvements in depression (r-0.59), anxiety (r-0.63) and general social functioning (r-0.68).1 6 Addi tionally, outcomes studies of Sleepio indicate that as well as resulting in improved sleep for users, the programme also facilitates general improvements in mental health (d--0.33) and workplace functioning (d-0.77).12

### Aims of the study

Over the past 4 years, Living Life to the Full Interactive, Sleepio and Breaking Free Online have been delivered to service users via the novel eTherapy delivery model developed by Self Help Services, which has provided an additional, digital treatment modality within the Greater Manchester IAPT service provision. Therefore, this service evaluation sought to explore feasibility and outcomes of the Self Help Services eTherapy delivery model and its potential to provide a useful addition to traditional IAPT treatment offerings. This is done via examination of psychosocial outcomes for service users engaging with each of the three eTherapy programmes using the IAPT Minimum Data Set, which is intended to facilitate service evaluation and development, and measures depression, anxiety and social functioning.

Guidance from the Medical Research Council (MRC) recommends that alongside examining clinical effectiveness of complex interventions via randomised controlled trials (RCIs), feasibility studies allow examination of acceptability of a new intervention, service user compliance and different delivery approaches. These are all important considerations that can impact on recruitment and retention of service users and, ultimately, clinical outcomes.<sup>45–44</sup> Additionally, the MRC framework recommends that feasibility and pitoting work be conducted, both within research and community-based treatment delivery settings, in order to contribute to further development of clinical content of such interventions, and development of appropriate delivery models within the healthcare system. In this way, this service evaluation takes a pragmatic approach by examining feasibility of delivering eTherapies in a community mental health service, using clinical outcomes data from service users, as opposed to data collected within the highly controlled context of a research study such as a clinical trial, where ecological validity may be low.<sup>6546</sup>

## METHOD

## Design

This study had a pre-test post-test design, using standardised psychometric assessments from the IAPT Minimum Data Set, to examine mental health and social functioning outcomes, in three separate groups of service users accessing different eTherapy treatment programmes in a community-based mental health service.

## Participants

Participants were 1068 service users receiving support for a range of mental-health-related issues from Self Help Services, an eTherapy service, between 2011 and 2015. Inclusion criteria included any service user over the age of 18 years accessing one of the three eTherapy programmes provided by Self Help Services (Living Life to the Full Interactive, Sleepio or Breaking Free Online), who had completed their eTherapy treatment period, provided post-treatment assessment data and consented for their anonymised data to be used for service evaluation purposes at the start of their treatment. Self Help Services provides services across Greater Manchester and some areas of Liverpool, including self-help and peer support groups, face-to-face counselling and a mental health crisis centre, alongside access to a eTherapies.

Services users had either been referred to Self Help Services by a healthcare practitioner or self-referred. On entering the service, they completed an initial assessment comprising a battery of standardised psychometric assessments, which forms the IAPT Minimum Data Set, and a consultation with a Self Help Services practitioner to establish their principal area of difficulty: depression, anxiety and/or stress, sleep disruption or problems with alcohol or drugs. These initial assessments were conducted either face to face or via telephone depending on the service users' preference. Practitioners were all trained to provide guidance and support to service users using the eTherapy programmes, with some practitioners also having lived experience of mental health difficulties. Following a collaborative discussion between the service user and practitioner, service users were triaged into the most appropriate of the three eTherapies and supported in setting up an account on the relevant programme.

## Procedure

Service users entered the service and were evaluated as above during a 4-year period between 2011 and 2015. Primary outcome measures came from the Minimum Data Set of standardised psychometric assessments completed

Elison S. et al. BMJ Open 2017;7:e016392. doi:10.1136/bmiceon-2017-016392

in NHS IAPT services throughout England to facilitate service evaluation, and included:

- A. The Patient Health Questionnaire (PHQ-9)<sup>47</sup>: This nine-item scale measures levels of depression and contains validated clinical norms, with a possible score range of 0–27. Internal reliability of the PHQ-9 has been found to be excellent (α–0.89), with test–retest reliability also being excellent (r=0.84). Score ranges for severity of depression are: 0–4, 'minimal'; 5–9, 'mild'; 10–14, 'moderately severe'; 20–27 'severe.'
- B. The General Anxiety Disorder Scale (GAD-7)<sup>48</sup>. This seven-item scale measures levels of anxiety and also contains validated clinical norms and has a possible score range of 0–21. Factor analyses revealed the GAD-7 to have a one-dimensional factor structure with item factor loadings ranging between 0.69 and 0.81, with internal consistency being excellent ( $\alpha$ =0.92). Score ranges for severity of anxiety are: 0–4, 'minimal'; 5–9, 'mild'; 10–14, 'moderate': 15–21, 'severe.'
- C. The Work and Social Adjustment Scale (WASA)<sup>49</sup>: This five-item scale measures levels of social impairment and has a possible score range of 0–40. Cronbach's α measure of internal scale consistency ranged from 0.70 to 0.94, with test–retest correlation being 0.73.

This baseline assessment was completed with a total of 1787 service users. Once the baseline assessment had been completed, service users were triaged into the most appropriate eTherapy programme and provided with full access. They were then followed up with a telephone call once a week from the service coordinator and were offered the opportunity to come into the service for face-to-face support if required. Then, following a period of engagement with the eTherapy programme each service user was triaged into, each service user was contacted to arrange a time to complete the last treatment session, during which the same measures were completed a second time as part of a post-treatment assessment. Of the 1786 service users completing the baseline assessment, a total of 1068 (59.8%) started treatment and provided post-treatment assessment data, with 719 (40.2%) completing the baseline assessment but not completing treatment and providing post-treatment assessment. Reasons for service users not completing treatment and providing post-treatment data were: 216 (38.5%) disengaged/dropped out from the service following baseline assessment, 251 (34.9%) were discharged from the service following baseline assessment because treatment was not required; and 252 (35%) were referred to another service, for example, higher-intensity IAPT or non-IAPT services. Comparisons between those service users providing post-treatment assessment data and those who did not, indicated no significant baseline differences between the groups in terms of psychometric assessment scores (PHQ-9, p=0.234; GAD-7, p=0.061; WASA, p=0.54).

#### Data analyses

6

BNJ

Shapiro-Wilk tests revealed data from the main outcomes measures (PHQ-9, GAD-7, WASA), both at baseline and post-treatment, to be non-normally distributed (all p<0.05); therefore non-parametric tests were run to analyse data. Two main sets of analyses were conducted. Kruskal-Wallis analyses of variance (ANOVA) were conducted to examine baseline differences between the three eTherapy groups on the mental health and social functioning assessment contained within the IAPT Minimum Data Set. Analyses of covariance (ANCOVA) were also conducted to examine whether eTherapy group assignment was predictive of the extent to which scores for post-treatment mental health and social functioning differed from baseline scores, that is, the degree of change in functioning, whilst controlling for the participant characteristics of age and gender. Additionally, separate within-group, repeated-measures Wilcoxon signed-rank ANOVA tests were conducted to examine changes in the same psychometric outcomes from baseline to post-treatment assessment within each of the three eTherapy groups. Pearson's effect sizes (r) were calculated using test statistics from ANOVA and ANCOVA tests run in SPSS (Z) and sample size (n) using the following formula<sup>50</sup>:

$$r = \frac{Z}{\sqrt{n}}$$

### RESULTS

Clinical outcomes data from a total of 1068 service users from the Self Help Services eTherapy service were included in the analyses, with 866 (81%) having accessed Living Life to the Full Interactive, 85 (8%) having accessed Sleepio and 117 (11%) having accessed Breaking Free Online. Across the entire sample, 679 (63.6%) were female; by group, 572 (66.1%) of those allocated to Living Life to the Full Interactive were female, compared with 60 (70.6%) to Sleepio and 47 (40.2%) of Breaking Free Online users. Across the whole sample, mean age was 37.38 years (range 16-79 years; SD, 11.98), with a mean age of 36.11 years (range 16-73 years; SD, 11.51) for Living Life to the Full Interactive users, 45.21 years (range 20-79 years; SD, 15.04) for Sleepio users, and 41.21 years (range 19-60 years; SD, 9.80) for Breaking Free Online users.

Time periods of engagement with the service varied, with some service users engaging in treatment for longer periods than others, depending on their need. Engagement periods for the whole sample ranged from 4 days to 288 days (0.64–41.14 weeks), with a median engagement treatment period of 62.39 days (IQR–40.18). For each individual eTherapy group, engagement periods were as follows: for Living Life to the Full Interactive users, 4–288 days (0.64–41.14 weeks) with a median of 66.29 days (IQR–43.06); for Sleepio users, 29–148 days (4.19– 21.08 weeks) with a median of 66.35 days (IQR–39.06);

Elison S. et al. BMJ Open 2017;7:e016392. doi:10.1136/bmiepen-2017-016392

Table 1 Comparison of baseline psychometric outcomes for the three eTherapy programmes						
	Breaking Free Online mean (SD)	Living Life to the Full Interactive mean (SD)	Sleepio mean (SD)	z	p Value	r
PHQ-9 baseline	14.20 (6.43)	11.89 (4.77)	11.34 (5.12)	12.00	<0.0001	0.36
GAD-7 baseline	12.18 (5.75)	11.32 (3.98)	8.49 (5.00)	9.80	<0.0001	0.31
WASAS baseline	16.84 (9.74)	16.14 (7.91)	15.65 (9.12)	0.56	0.569	0.00

and for Breaking Free Online users, 6–205 days (0.92– 29.35 weeks) with a median of 58.29 days (IQR=48.64). At the end of each service user's period of engagement with the service, the same battery of assessment measures was completed at point of discharge, which was between 4 days and 288 days following the baseline assessment.

When the three eTherapy groups were compared using Kruskal-Wallis ANOVAs on their psychometric assessment scores for mental health and social functioning at baseline, some significant differences between the groups were found (table 1). The Breaking Free Online group was found to have significantly higher scores on the PHQ-9 assessment for baseline depression (Breaking Free Online mean-14.20) than the other two eTherapy groups (Living Life to the Full Interactive mean-11.89, CI 1.27 to 3.21, p<0.0001; Sleepio mean-11.34; CI 1.19 to 4.52, p<0.0001). The Breaking Free Online group was also found to have significantly higher scores at baseline on the GAD-7 assessment for anxiety than the other two eTherapy groups (Breaking Free Online mean-12.18; Living Life to the Full Interactive mean-11.32, CI. 077 to 1.72, p<0.0001; Sleepio mean-8.49, CI 2.16 to 5.23, p<0.0001).

In addition to comparing the three eTherapy groups at baseline, degrees of psychometric score change from baseline to post-treatment were also compared across the three eTherapy groups, with baseline scores on each of the psychometric assessments being regressed against post-treatment scores, while controlling for age and gender. ANCOVAs revealed no significant differences between the groups in terms of degree of change in any of the baseline psychometric assessments including the following: PHQ-9 F-2.373, p-0.094; GAD-7 F-3.239, p-0.052; WASAS F-0.164, p-0.848. Additionally, when outcomes from baseline to post-treatment were compared within each of the three eTherapy groups, using Wilcoxon signed-rank tests, significant reductions in scores on PHQ-9, GAD-7 and WASAS were found within all three groups, with p values across all three eTherapy groups being <0.0001. In addition to conducting these Wilcoxon analyses, effect sizes were calculated. In all three eTherapy groups and for each of the psychometric assessments, effect sizes were in the moderate to large range. See table 2 for full details of these within-group analyses.

Given that service users accessed each eTherapy programme for varying lengths of time (4–288 days), regression analyses were conducted to examine whether number of days engaging with each programme was associated with degree of change in depression, anxiety and

Elison S. of al. BMJ Open 2017;7:e016392. doi:10.1136/bmjopen-2017-016392

social functioning from baseline to post-treatment assessment. These regression analyses revealed that length of period of engagement (in days) was not related to degree of change from baseline for service users accessing either the Living Life to the Full Interactive or Steepio groups. However, number of days of engagement was associated with degree of change in scores for depression, anxiety and social functioning from baseline to post-treatment assessment for those service users accessing Breaking Free Online. For Breaking Free Online users, the greater the number of days of engagement with the programme, the greater the reduction in scores for depression, anxiety and social impairment (table 3).

Alongside statistically significant within-group changes in scores on PHQ-9, GAD-7 and WASAS (table 2), the percentages of service users reaching clinical thresholds scores for mild, moderate and severe depression and anxiety at baseline were compared with the percentages reaching each threshold at post-treatment (see tables 4 and 5). Chi-square analyses demonstrated that within each of the three eTherapy groups, the percentages of service users reaching threshold scores for clinically relevant depression and anxiety (a score of 5 $\leq$ ) after treatment reduced significantly from baseline: Living Life to the Full Interactive (PHQ-9  $\chi^2$ -260.30, p<0.0001; GAD-7  $\chi^2$ -105.44, p<0.0001), Breaking Free Online (PHQ-9  $\chi^2$ -68.77, p<0.0001; GAD-7  $\chi^2$ -45.88, p<0.0001), Sleepio (PHQ-9  $\chi^2$ -57.24, p<0.0001; GAD-7  $\chi^2$ -44.23, p<0.0001). Specifically, the percentages of service users

		z	p Value	r
PHQ-9	Breaking Free Online	-6.771	<0.0001	0.63
	Living Life to the Full Interactive	-21.450	<0.0001	0.73
	Sleepio	-7.226	<0.0001	0.78
GAD-7	Breaking Free Online	-6.449	<0.0001	0.60
	Living Life to the Full Interactive	-21.463	<0.0001	0.73
	Sleepio	-6.365	< 0.0001	0.69
WASAS	Breaking Free Online	-5.558	<0.0001	0.51
	Living Life to the Full Interactive	-13.729	<0.0001	0.47
	Sleepio	-4.967	<0.0001	0.54

Table 3 Regression analyses demonstrating associations between number of days of engagement with each eTherapy programme and degree of change in psychometric scores

		F	β	Р	95% CI
Breaking Free Online	PHQ-9	34.387	-0.341	<0.0001	-0.214 to 0.039
	GAD-7	28.396	-0.330	<0.0001	-0.193 to 0.027
	WASAS	26.648	-0.190	0.020	-0.170 to 0.176
Living Life to the Full Interactive	PHQ-9	168.814	-0.038	0.185	-0.039 to 0.013
	GAD-7	95.392	-0.041	0.178	-0.036 to 0.016
	WASAS	127.071	-0.021	0.475	-0.037 to 0.058
Sleepio	PHQ-9	31.996	-0.130	0.129	-0.065 to 0.056
	GAD-7	40.010	-0.138	0.083	-0.060 to 0.047
	WASAS	30.856	-0.050	0.556	-0.147 to 0.097

in the minimal and mild categories of symptom severity increased after treatment, while the percentages of service users in the categories between moderate and severe symptom severity decreased (see tables 4 and 5 for score ranges for each severity category).

#### DISCUSSION

Given that eTherapy is still a relatively novel treatment modality within the UK mental health sector, this service evaluation explored the feasibility of a novel treatment delivery model developed by one of the UK's only eTherapy mental health services: Self Help Services. This service evaluation used outcomes from community-based service users receiving treatment via three eTherapy programmes provided by Self Help Services: Living Life to the Full Interactive for low mood, stress and anxiety; Sleepio for insomnia; and Breaking Free Online for substance misuse. Statistically significant within-group reductions in scores for anxiety, depression and social impairment were demonstrated for each of the three eTherapy programmes. Additionally, regardless of eTherapy group allocation, degrees of reduction in mental health scores were comparable across each of the three eTherapy programmes, (table 1) with eTherapy group assignment not being predictive of degree of change in depression, anxiety or social functioning scores between baseline and post-treatment assessment.

Despite the equivalent outcomes across the three eTherapy groups, there were some significant differences between the groups in terms of scores on the assessment measures and hence the severity of their mental health difficulties. The Breaking Free Online group was found to have significantly higher baseline scores for depression and anxiety than the Living Life to the Full Interactive and Sleepio groups, and significantly higher scores for anxiety at post-treatment assessment. These findings may make sense, as the Breaking Free Online group may have presented with more complex difficulties than the Living Life to the Full Interactive and Sleepio groups, given the extent to which substance use can impair mental health51 52 and the often chaotic lifestyle that is common for individuals with drug and alcohol difficulties.41.53 The significantly higher post-treatment anxiety for the Breaking Free Online group may also be explained by the fact that, in the early stages of substance misuse recovery, many individuals may experi-ence a temporary worsening of mental health symptoms<sup>5455</sup> when the anti-depressant effects of previously consumed drugs are removed, particularly in the case of opiates.36

When clinical threshold scores for depression and anxiety were examined, there were reductions across all three groups in percentages of service users reaching threshold scores for clinically significant anxiety and depression at post-treatment assessment. These findings reinforce the statistically significant reductions in scores for mental health found across the three eTherapy groups. Moreover, it is important to note that anxiety and depression scores reduced in association with all programmes, despite these symptoms not being the principal clinical targets of the Breaking Free Online and Sleepio programmes.

Table 4 Changes in percentages of service users reaching clinical threshold scores for depression by eTherapy group							
	PHQ-9	baseline threshold	(%)	PHQ-9 post-treatment threshold (%)			
PHQ-9 score thresholds	Breaking Free Online	Living Life to the Full Interactive	Sleepio	Breaking Free Online	Living Life to the Full Interactive	Sleepio	
Minimal (range 0-4)	7.7	6.9	10.6	37.6	40.5	42.4	
Mild (range 5–9)	21.4	23.1	20	15.4	33.5	38.7	
Moderate (range 10-14)	20.5	40.3	44.7	22.2	15.5	11.8	
Moderately severe (range 15-19)	26.5	24.2	18.8	11.1	7.8	2.4	
Severe (range 20–27)	23.9	5.5	5.9	13.7	2.7	4.7	

6

Elison S, et al. BMJ Open 2017;7:e016392. doi:10.1136/bmjopen-2017-016392

6

Table 5 Changes in percentages of service users reaching clinical threshold scores for anxiety by eTherapy group								
	GAD-7 baseline threshold (%)			GAD-7 post-treatment threshold (%)				
GAD-7 thresholds	Breaking Free Online	Living Life to the Full Interactive	Sleepio	Breaking Free Online	Living Life to the Full Interactive	Sleepio		
Minimal (range 0-4)	11.1	3.6	27.1	38.5	43.3	58.9		
Mild (range 5–9)	26.5	41.1	43.4	28.2	37.9	23.5		
Moderate (range 10-14)	28.2	40.2	20	15.4	11.3	14.1		
Severe (range 15-21)	34.2	15.1	10.6	17.9	7.5	3.5		

The findings from this service evaluation would appear to support the feasibility and effectiveness of eTherapy programme delivery in mental health services, and would appear to support findings from previous research, including outcomes studies demonstrating effectiveness of each of the three eTherapy programmes,<sup>10</sup> <sup>11</sup> <sup>15</sup> <sup>14</sup> <sup>16</sup> <sup>17</sup> and findings related to delivery of eTherapies more generally. For example, eTherapies have been demonstrated as being useful additions to adult mental health services,<sup>58</sup> services for children and young people<sup>57 58</sup> and also treatment provision delivered by social care workers.50 However, some authors have advised caution around the potential of eTherapies, as many may not be grounded in psychological theory and may lack a solid evidence base. Additionally, in some cases, during the development process, the challenges of implementation and uptake may not have been adequately considered, meaning that some eTherapics may not fulfil their promise of widening access to treatment.<sup>61</sup> However, despite this caution, a recent survey of NHS mental health services showed that the provision of eTherapies is an emerging and growing trend, meaning that it is becoming increasingly important to put structures in place to ensure that only evidencebased eTherapies are commissioned and delivered.

### Limitations to the study

Although the findings from this study are promising, there are a number of limitations that merit discussion. Firstly, the sample sizes across the three programmes varied, with the Living Life to the Full Interactive group (n-866) being considerably larger than the Sleepio and Breaking Free Online groups (n-85 and n-117, respectively). However, given that Self Help Services is an eTherapy service for individuals with mental health problems that are common among the general population, it is unsurprising that Living Life to the Full Interactive is accessed by more service users, as it is the only eTherapy programme of the three included in this service evaluation that is designed specifically for addressing mental health difficulties, such as low mood, stress and anxiety. Additionally, Living Life to the Full Interactive is a more established programme and has been provided in mental health services for a number of years, in contrast to Sleepio and Breaking Free Online, which are still relatively new eTherapy programmes.

Further limitations are the lack of follow-up data from participants and the fact that the study did not include

Elison S, et al. BMJ Open 2017;7:e016392. doi:10.1136/bmjopen-2017-016392

randomisation and control groups. However, although methodologies such as RCIs are an important part of the development and evaluation process for complex interventions, <sup>14</sup> <sup>63</sup> <sup>64</sup> there is now a growing literature to suggest that additional methodologies, employed alongside RCIs, may be required for evaluating complex, multi-component interventions such as cTherapies for mental-health-related conditions.<sup>44</sup> <sup>65–68</sup> When eTherapies are evaluated via RCT designs, there may be some methodological limitations due to the tailorability of these programmes, which though may enhance clinical effectiveness,<sup>69–71</sup> may result in within-group variation in terms of the personalised sets of intervention strategies that each user may complete.

This study was also restricted to analyses of data routinely collected for service evaluation purposes at IAPT services via the Minimum Data Set, meaning that the authors were not able to make decisions around which psychometric measures should be used. It may be that there are limitations with the psychometric properties of the measures that have been selected for inclusion in the Minimum Data Set, and therefore, other measures may be more appropriate. Additionally, some data that would have been informative were not available in the Minimum Data Set, such as around severity of insomnia for the Sleepio group and severity of substance dependence and substance consumption for the Breaking Free Online group. The Minimum Data Set also does not record whether service users have received face-to-face or telephone support during each contact with Self Help Services, and some information was not available around reasons for service users disengaging with the service or, reasons why service users engaged with each of the programme for the time periods they did. For example, it would have been informative to understand service user satisfaction with each programme, or why they stopped using each one.

An additional factor that could be viewed as a limitation is that service users engaged with the eTherapy programmes for varying lengths of time, between 4 days and 288 days. However, eTherapy programmes are designed to offer such patient-centred flexibility, with this being associated with their potential to be clinically effective.<sup>21</sup> Additionally, regression analyses revealed that number of days of engagement did not appear to be associated with degree of change in scores for depression, anxiety and social impairment, from baseline to treatment assessment, for

the Living Life to the Full Interactive and Sleepio users. However, there did appear to be a significant association between number of days of engagement with Breaking Free Online and degree of improvement in depression, anxiety and social impairment from baseline to post-treatment assessment. This may be explained by the fact that the Breaking Free Online group had more severe mental health problems at baseline and were being treated for a particularly challenging mental-health-related condition, that being substance misuse and comorbid mental health difficulties.

Finally, just over 40% of service users who initially engaged with the service did not provide post-treatment data, although for the majority of these service users, they were either discharged due to the service not being required or were referred to more appropriate services. Aurition is common in eTherapies<sup>20 72 75</sup> with this now a focus for research in the digital health sector more generally. The problem of autrition may also be an issue in mental health interventions,<sup>74 75</sup> with drop-out from psychological therapies being associated with poorer outcome for service users."

### Implications of the findings

The data presented here demonstrate that the Self Help Services eTherapy model may have the potential to inform future mental health service delivery, given the encouraging clinical outcomes reported and the potential cost-effectiveness of such an approach. In addition, as there are now significant and lengthening waiting listseven for IAPT services that were originally intended to reduce waiting times for mental health services<sup>6</sup> widening service provision to incorporate eTherapies may increase access to evidence-based psychosocial treatment for large numbers of people who could benefit from it.

PWPs, who are trained specifically to deliver lower-intensity interventions within IAPT services, may be ideally placed to incorporate the delivery of eTherapies into their current roles. This is because they are trained to conduct assessments, build a therapeutic alliance with service users, work collaboratively with them to identify areas in which they wish to see change, deliver assisted self-help and provide information about other services that may be beneficial to each individual service user's recovery.77 By expanding their therapeutic repertoire to include the provision of eTherapies, PWPs would be able to deliver comprehensive, evidence-based programmes that are highly standardised and not subject to the variation in fidelity of delivery that is common to more traditional psychosocial interventions,2 given that all clinical content is delivered via the computer.

Clearly, it is important to provide access to psychosocial interventions to address the increasingly pressing issue of waiting times for mental health services and to ensure these interventions are effective and evidence based. This service evaluation has demonstrated that evidence-based eTherapy programmes can be effective, using clinical outcomes data from service users in a community-based BWJ

mental health service, as opposed to data from a highly controlled study, enhancing ecological validity and generalisability of findings. Future research is planned to explore the longer-term clinical outcomes of providing eTherapies as part of an IAPT stepped-care model and potential waiting list time and cost implications of such a service.

Acknowledgements The authors would like to thank the service users, volunteers and eTherapy co-ordinators at Self Holp Services for their participation in the service, in addition to Nic Seccombe (Self Help Services) for his internatics assistance and Dr Andrew Jones (institute of Population Health, University of Manchester) for his statistical analysis advice.

Contributors SE led on study design, data analyses, Morature review and drafting the manuscript. JW developed the Breaking Free Celline treatment programme and co-wrole the manuscript. CW developed the LMmg Life to the Full interactive treatment programme and provided comments on the manuscript. CE developed the Sleepie treatment programme and provided comments on the manuscript. CD developed the Breaking Free Online treatment programme and provided comment on the manuscript. SD assisted with literature review and data analyses. KR provided support to service users in accessing the three eTherapy programmes, collected outcomes data and assisted with access to the data. LC p rovided support to service users in accessing the three of herapy programmes and collected outcomes data. NL provide permissions for accessing the data, organised access to the data and provided comments on the manuscript. KS extracted data for analyses and provided these to the lead author.

Competing interests SE, 6D and SD are employed by Breaking Free Group, where the Breaking Free Online programme was developed. JW and 6D are authors of the Breaking Free Online programme. JW is founder and a director and shareholder of Breaking Free Group, CW is President of BABCP, the lead body for CBT in the UK, and is also author of Living Life to the Full Interactive as well as a range of other CBT-based resources that address anxiety, depression and other disorders, which are available commercially. He receives royalties for these and is sharehelder and director of a company that commercialises these resources. CE is co-founder and CMO of Big Health Ltd (Sloepk), where he is a sharoholder and receives remuneration from the company

Patient consent Dotail has been removed from this case description/these case descriptions to ensure anonymity. The oditors and reviewers have seen the detailed information available and are satisfied that the information backs up the case the authors are making.

Ethics approval This was a service evaluation using existing, non-identifiable service user data from Self Help Services, which were fully aborymised by Self Help Services before being provided to the lead author for analyses, and so external ethical approval was not required.

Provenance and peer review. Not commissioned; externally peer reviewed.

Data sharing statement. No additional data are available as all data provided by Self Holp Services for the perposes of the study are reported in the manuscript. Self Holp Services collects additional confidential data on their service users, though was not required for this study.

Open Access This is an Open Access article distributed in accordance with the Creative Commons Attribution Non Commercial (CC BY-NC 4.0) Icense, which pormits others to distribute, remit, adapt, build upon this work non-commercially, and license their derivative works on different terms, provided the original work is property cited and the use is non-commercial. See: http://creativecomi mons.org/ Iconses/by-nc/4.0/

© Article author(s) (or their employer(s) unless otherwise stated in the text of the article) 2017. All rights reserved. No commercial use is permitted unless otherwise expressly granted.

#### REFERENCES

- We need to talk: getting the right therapy at the right time. London: MIND, 2014. 1.
- Minto, 2014. Docherty M, Thernicroft G. Specialist mental health services in England in 2014: overview of funding, access and levels of care. Int J Mant Health Syst 2015;3:34. 2.

Elson S. at al. BMJ Doon 2017;7:e016392. doi:10.1136/bmicpon-2017-016392

B

Open:

first

published

25

6

1136/bmjopen-2017-016392

9

8

ŝ

2017.

Downloaded

from http://bm

lobeu

3

.00m

9 4 March

2 19 by

guest.

Protected

Ţ

copyright

## 6

- 3. National Institute for Health and Care Excellence. Common mental National Institute for Health and Care Excellence. Common mental health problems: identification and pathways to care: clinical guidelines. London: NICE, 2011. Available from. https://www.nice. erg.uk/guidance/eg123/chapter/introduction. TKs F. Briefing: mental health under pressure London: the King's Fund; 20152017;05 https://www.kingshurd.org.uk/sites/files/d/field/ field\_publication\_file/mental-health-under-pressure-nov15\_0.pdf. Clark DM. Implementing NICE guidelines for the psychological treatment of depression and anxiety disorders: the MPT experience. Int *Hew Psychiatry* 2011;23:318–27. We still need to talk: getting the right therapy at the right time. MindLondon, 2013. Bower P, Gilbody S. Stepped care in psychological therapies: access, effectiveness and efficiency, narrative literature review. Br J. Psychiatry 2005;186:11–17.

- 6
- 7. Psychiatry 2005;166:11–17. Andersson G, Cuipers P. Internet-based and other computerized
- 8.
- 9
- 10.
- Audenson G, Cuijpers P, Internet-based and other computerized psychological treatments for adult depression: a meta-analysis, Cogn Behav Ther 2009;38:108–206. Musiat P, Tarrier N, Collateral outcomes in e-meetal health: a systematic review of the evidence for added benefits of computerized cognitive behavior therapy interventions for mental health. Psychol Med 2014;44:3137–50. Williams C, Whitfield G, Written and computer-based self-help treatments for depression. Br Med Bull 2001;57:133–44. Richards D, Richards A, Barkham M, et al. PHASE: a 'health technology' approach to psychological treatment in primary mental health care. Primary Health Care Research and Development 2002;3:159–68. 11. 2002;3:159-68
- k S, Luik AI, Espie CA. Sleep and productivity benefits of 12 digital cognitive behavioral therapy for insomnia: a randomized controlled trial conducted in the workplace environment. J Occup
- Contraction and Contraction of the Workplace devicements, J. Coccup Environ Med 2016;35:663-9.
  Espise CA, Hames P, McKinstry B. Use of the internet and mobile media for delivery of cognitive behavioral insomnia therapy. Sloop Med Clin 2013;8:407–19. 13.
- Espie CA, Kyle SD, Miller CB, et al. Attribution, cognition and
- Espis CA, Kyle SD, Miller CB, et al. Attribution, cognition and psychopathology in persistent insomnia disorder: outcome and mediation analysis from a randomized placebo-controlled trial of enfine cognitive behavioural therapy. Skept Med 2014;15:913–7. Coulson NS, Smedley R, Bostock S, et al. The pros and corns of getting engaged in an online social community embedded within digital cognitive behavioral therapy for insomnia: survey among users. J Med Internet Res 2016;11:668. Elison S, Ward J, Davies G, et al. An outcomes study of el herapy for dual diagnosis using Breaking Free Online. Adv Daal Diagn 2014;7:52–62. Elison S, Davies G, Ward J, et al. Sub-group analyses of a heterogeneous samele of service users accessing computer assist reterogeneous samele of service users accessing computer assist
- 16.
- Etison S, Davies G, Ward J, et al. Sub-group analyses of a heterogeneous sample of service users accessing computer-assisted therapy (CAI) for substance dependence using Breaking Free Online. Journal of Medical Internet Resoarch 2015;2v13.
   Etison S, Davies G, Ward J et al. An outcomes evaluation of computerised treatment for problem dividing using breaking Free Online Alcoholism Treatment Guarterly. 2015;33, 118–96.
   Etison S, Humphrays L, Ward J, et al. A pilot outcomes evaluation of computering treatment for problem dividing using breaking Free Online Alcoholism Treatment Guarterly. 2015;33, 118–96.
   Etison S, Humphrays L, Ward J, et al. A pilot outcomes evaluation for computer assisted therapy for substance misuse an evaluation of Breaking Free Online. Journal of Substance Dise 2013;19:1–6.
   Etison S, Weston S, Davies G, et al. Findings from mixed methods feasibility and effectiveness evaluations of the "Breaking Free Online" treatment and recovery programme for substance misuse in prisons. Drugs: Education, Prevention and Policy 2015;23:1–10.
   Carrol KM, Boursavillo BJ. Computer assisted therapy as an adjunct to standard care for addiction. Drug Alcohol Depend 2010;10:200–7.
   Denster D, Unicken R, Londenkin M, Cost-off Depend 2010;10:200–7. 17. 18.
- 19
- 20.
- 21. 22
- 2010:110:200-7
- 2010; 10:200-7, Hedman E, Ljótsson B, Lindefors N, Cognitive behavior therapy via the internet: a systematic review of applications, clinical efficacy and cost-effectiveness. *Expert Rev Pharmacencer Outcomes Res* 1999;16:201-2014. 23. 2012:12:745-64.

- Carroll KM, Integrity T. Treatment integrity and dissemination: rethinking fidelity via the stage model. *Chrical Psychology: Science and Practice* 2013;20:99–108.
   Bornell B, Sepinwall D, Ernst D, et al. A new tool to assess treatment fidelity and evaluation of treatment fidelity across 10 years of health behavior research. *J Consult Clin Psychol* 2005;73:882–60.
   Porepletchikova F, Kazdin AE. Treatment integrity and therapoulic change: issues and research recommendations. *Clinical Psychology: Science and Psychol* 2005;12:365–63.
   Gerbards SA, Abma TA, Arntz A, et al. Improving adherence and effectiveness of computerised cognitive behavioural therapy without

Elson S, of al. BMJ Open 2017;7:e016392. doi:10.1136/bmjopen-2017-016392

- support for depression: a qualitative study on patient experiences. J Affect Disord 2011;129:117–25. Mohr DC, Cuigers P, Lehman K. Supportive accountability: a model for providing human support to enhance adherence to el-lealth interventions. J Med Internet Res 2011;13:e00. Donkin L, Christenson H, Naismith SL, et al. A systematic review of 28.
- 29.
- 30.
- merventions. J Med Internet Hes 2011;13:s00.
   Donkin L, Christonson H, Naismith SL, et al. A systematic review of the impact of adherence on the effectiveness of e-therapies. J Med Internet Hes 2011;13:e52.
   Newman MG, Szkodry LE, Llera SJ, et al. A review of technology-assisted solf-help and minimal contact therapies for drug and alcohol abuse and smoking addiction: is human contact necessary for therapeutic efficacy? *Clin Psychol Rev* 2011;31:178–80.
   Newman MG, Szkodry LE, Llera SJ, et al. A review of technology-assisted solf-help and minimal contact therapies for anoise; and depression: is human contact necessary for therapeutic efficacy? *Clin Psychol Rev* 2011;31:178–80.
   Newman MG, Szkodry LE, Llera SJ, et al. A neview of technology-assisted solf-help and minimal contact therapies for anoise; and depression: is human contact mecassary for therapeutic efficacy? *Clin Psychol Rev* 2011;31:89–103.
   Elson S, Ward J, Duvies G, et al. Implementation of computerassisted therapy for substance misuse: a qualitative study of Brevking Free Online using Boger's diffusion of innovation theory. *Drugs Alcohol Today* 2014;14:207–18.
   Barnett J, Vasikiou K, Djemi F, et al. Understanding innovators' experiences of barriers and facilitations in implementation and diffusion of beabhcare service innovations: a qualitative study. *BMC Health Serv Fins* 2011;11:242.
   Innovation NHS, Health. Wealth: accelerating adoption and diffusion in the NHS. London: Department of Health, 2011.
   Kenicer D, McClay CA, Wilams C. A national survey of beoth neces in one officiancy officiancy officiancy in placement and a beful service officiancy adoption in the NHS. London: Department of the other service officiancy and officiance on the other service officiancy adoption in the NHS. London: Department of Health, 2011. 31.
- 32
- 33.
- 34.
- Expandence of Health, 2011. Kenicer D, McClay CA, Williams C. A national survey of health service infrastructure and policy impacts on access to computerised CBT in Scotland. *BMC Med Inform Decis Mak* 35. 2012:12:1-5
- Andrewse H, Kenicer D, McClay G-A, et al. A national survey of the infrastructure and IT policies required to deliver computerised cognitive behavioural therapy in the English NHS. *BMJ* Open region or computerioural therapy in the English NHS. 263 2013:3:e002277
- 2013:3:e002277. Fund King's. The digital revolution: eight technologies that will change health and care 2016, 2016, updated Available from, http:// www.kingstund.org.uk/publications/articles/eight-technologies-will-change-health-and-care. NCE: Computational cognitive behaviour therapy for depression and arxiety review of technology. London: National Institute for Health and Clinical Eccellence 2009. Bock AI: Cognitive therapy: past, present, and future. J Consult Clin Psychol 1903;61:194–8. Bock JS: cognitive therapy: past, present, and heater. Clinical Bock JS: cognitive therapy: past, present, and heater. J Consult Clin Psychol 1903;61:194–8.
- Beck JS. Cognitive behavior therapy: basics and beyond: Guilford 40.
- 41.
- Beck JS: Cogmere example of the probability of the
- 2015;10. Espis CA, Kyle SD, Williams C, et al. A randomized, placebo-controlled trial of online cognitive behavioral therapy for chronic insomnia disorder delivered via an automated media-rich web application. Sleep 2012;35:769-. Moore GF, Audrey S, Barker M, et al. Process evaluation of complex interventions: Medical Research Council guidance. BMJ 2015;30:0:h1298. 42
- 43.
- 44.
- 45
- 2015;390:h1268. Craig P, Dioppe P, Macintyre S, et al. Developing and evaluating complex interventions: the New Medical Research Council guidance. *BMJ* 2006;337:a1665-a65.. MacPhoneon H, Pragmattic clinical trials. Complement Ther Med 2004;12:136-40 https://doi.org/. Curran GM, Bauer M, Mittman B, et al. Effectiveness implementation hybrid designs: combining elements of clinical effectiveness and implementation research to enhance public health impact. Med Care 2012;9:0:17-29. 46 2012:50:217-26
- Kroenke K, Spitzer RL, Williams JBW. The PHQ-9. J Gen Intern Med 2001;16:608–13. 47. 48
- 2001;10:606–13. Spitzer RL, Kroenke K, Williams JB, et al. A brief measure for assessing generalized anxiety disorder: the CAD-7. Arch Intern Med 2006;106:1002–7. Mundt JC, Marks IM, Shear MK, et al. The work and Social
- 49. Adjustment Scale: a simple measure of impairment in functioning. Br J Psychiatry 2002;180:481-4. Field A. Discovering statistics using SPSS. Third Edition. London: Search 2000.
- 50 Sage, 2009. Buckley PE Prevalence and consequences of the dual diagnosis of substance abuse and severe mental illness. *J Can Psychiatry* 51.
- 2006;67 Suppl 7:5-0. Drake RE, O'Neal EL, Wallach MA. A systematic review of psychosocial research on psychosocial interventions for people with 62

- co-occurring severe mental and substance use disorders. J Subst Abuse Treat 2008;34:123-38. Smith K, Flatley J. Drug Misuse Declared: findings from the 2010/11 british crime survey England and Wales. London: Home Office, 2011. 53. 2011.
- 54.
- 55
- 2011. Lubelczyk R. Dotoxification or supervised withdrawal. In: Trestman R. Appelbaam K. eds. Oxford Textbook of Correctional Psychiatry. Oxford: oxford University Press, 2015. Powell JE, Taylor D, Anger TD, Anger, depression, and anxiety following hearin withdrawal. Int J Addict 1982;27:25-35. Agular de Ancos F, Verdeje-García A, Cevenino A, et al. Dysregulation of emotional response in current and abstiment becoin users: negative heightening and positive blanting. Psychopharmacology 2009;198:159-06. 56.
- 2008;198:150-68. Stasiak K, Fleming T, Lucasson MF, et al. Computer-based and online thorapy for depression and aroioty in children and addescents. J Child Actobesc Psychopharmascol 2016;26:235-45. Boydell KM, Hodgins M, Pignatiello A, et al. Using technology to deliver mental health services to children and youth: a scoping review. J Can Acad Child Adolesc Psychiatry 2014;23:87-99. Ramsey AT, Montgomery K. Technology-based interventions in social work practice: a systematic review of mental health interventions. Soc Work Health Care 2014;53:883-99. Leich S, Flatt S, Ace based psychological interventions: friend or 57.
- 58.
- 59.
- 60.
- Sec Profit Health Care 2014;53:383-99. Leigh S, Flatt S. App-based psychological interventions: friend or loo? Evid Based Ment Health 2015;18:97-9. Andonson G, Titov N. Advantages and finitiations of Internet-based interventions for common mental disorders. World Psychiatry 2014;13:4–11. 61.
- 2014;13:4-11.
   Bernion MR, Hardy G, Meore RK, et al. E-therapies in England for stress, anxiety or depression: what is being used in the N457 A survey of mental health services. *BMJ Open* 2017;7:e014844.
   Proudloot J, Ryden C, Everitt B, et al. Clinical efficacy of comparitorisod cognitive-behavioural thomapy for arxieity and depression in primary care: randomised controlled trial. *Br J Psychiatry* 2004;105:46-64.
   Carnoll KM, Ball SA, Martino S, et al. Computus-assisted delivery of cognitive-behavioral therapy for addiction: a randomized trial of CBT4CBT. *Am J Psychiatry* 2008;165:981-8.

- arch AIDS
- Padian NS, McCoy SI, Balkus JE, et al. Weighing the gold in the gold standard: challenges in HIV prevention research. AD 2010;24:621–35.
   Victora CG, Habicht JP, Bryce J. Evidence-based public health: moving beyond randomized trials. Am J Public Health 2004;34:400–5.
- 67.
- 68.
- 2004;34:400-5. Kaptichuk JJ, The double-blind, randomized, placebo-controlled trist: gold standard or golden call? J Cân Epidemiol 2001;54:541-9. Mohr DC, Schueller SM, Riley WT, et al. Trials of intervention principles: evaluation mothods for evolving Behavioral intervention fechnologies. J Med Intermet Res 2015;17:e168. Near SM, Benac CN, Harris MS. Does tailoring matter? Meta-analytic review of tailored print health behavior change interventions. Psychol Bull 2007;133:673-93. Lustria ML, Near SM, Cortese J, et al. A meta-analysis of web-delivered tailored health behavior change interventions. J Health Commun 2013;18:1039-89. 69.
- 70.
- 71.
- 72.
- 73.
- 75.
- 76.
- 77.
- 78.
- Lustin ML, Noar SM, Cortese J, et al. A meta-analysis of web-delivered tailored health behavior change interventions. J Health Commun 2013;18:1008–69.
   Krebe P, Prochaska JO, Rossi JS. A meta-analysis of computer-talored interventions for health behavior change. Prov Med 2010;51(3-4):214-21.
   Khadjesai Z, Murray E, Kalsitzaki E, et al. Impact and costs of incentives to reduce attition in online trials: two randomized controlled trials. J Med Internet Res 2011;13:e26.
   Eysenbach G. The law of attition. J Med Internet Res 2005;7:e11.
   Genring HE, Townseed L, Elkins J, et al. Mata-analysis of dropout from cognitive behavioral therapy: mognitude, timing, and moderators. J Commut Cito Psychol 2016;93:1106–22.
   Delgadillo J, McMillan D, Lucock M, et al. Early changes, attrition, and doso-response in low intensity psychological interventions. Br J Chin Psychol 2014;52:114-30.
   NHS. Psychological methoing thest practic gales. Improving Access to Psychological Therapise. London: NHS, 2010.
   Wildt JH, Weight AS, Albano AM, et al. Computer-assisted cognitive therapy for depression: maintaining difficucy while reducing therapist time. Am J Psychological Therapise. London: NHS, 2010.
   Wight JH, Weight AS, Albano AM, et al. Computer-assisted cognitive therapy for depression: maintaining efficacy while reducing therapist time. Am J Psychological Therapise. London: NHS, 2010.
   Wight JH, Weight AS, Albano AM, et al. Computer-assisted cognitive therapy for depression: maintaining efficacy while reducing therapist time. Am J Psychological Therapise. London: NHS, 2010.

BMJ

Open:

11S

published

2

5

.1136/bmjopen-2017-016392

9

3 È

2017.

Downloaded from http://bmjopen.bmj

00m

on 4 March

2019 by guest.

Protected

by copyright

10

## Paper Six

Luik, A. I., Bostock, S., Chisnall, L., Kyle, S. D., Lidbetter, N., Baldwin, N., & Espie, C. A. (2017). Treating depression and anxiety with digital cognitive behavioural therapy for insomnia: a real world NHS evaluation using standardized outcome measures. *Behavioural and Cognitive Psychotherapy*, 45(1), 91-96. https://doi.org/10.1017/s1352465816000369

## Paper Seven

Gellatly, J., Chisnall, L., Seccombe, N., Ragan, K., Lidbetter, N., & Cavanagh, K. (2018). @ Home eTherapy service for people with common mental health problems: an evaluation. *Behavioural and Cognitive Psychotherapy*, 46(1), 115-120 <u>https://doi.org/10.1017/S1352465817000297</u>

Article removed due to copyright restrictions.

The Author Accepted Manuscript version is available open access at http://sro.sussex.ac.uk/id/eprint/67496/

# Appendix 2

# Letters from collaborating authors confirming contribution



21st October 2020

This letter is to confirm the contribution of Nicky Lidbetter to the following paper:

Cavanagh, K., Seccombe, N., Lidbetter, N., & Bunnell, D. (2011). Supported, service-user led, computerised cognitive behavioural therapy (CCBT) self-help clinics. Journal of Public Mental Health, 10(4). https://doi.org/10.1108/1746572111188241

This paper documents the delivery of supported, service-user led, computerised CBT in self-help clinics. It describes the ground-breaking delivery model pioneered by Self Help Services under Nicky's leadership and summarises relevant outcome data. Nicky was involved in the preparation, writing and revisions of the manuscript.

Yours sincerely

Kate Cavanagh Professor of Clinical Psychology

School of Psychology | University of Sussex | Name of Building | Brighton BN1 9QH| United Kingdom kate.cavanagh@sussex.ac.uk www.sussex.ac.uk/psychology



21st October 2020

This letter is to confirm the contribution of Nicky Lidbetter to the following paper:

Cavanagh, K., Seccombe, N., Lidbetter, N., & Bunnell, D. (2011). Supported, service-user led, computerised cognitive behavioural therapy (CCBT) self-help clinics. Journal of Public Mental Health, 10(4). https://doi.org/10.1108/17465721111188241

This paper documents the delivery of supported, service-user led, computerised CBT in self-help clinics. It describes the ground-breaking delivery model pioneered by Self Help Services under Nicky's leadership and summarises relevant outcome data. Nicky was involved in the preparation, writing and revisions of the manuscript.

Yours sincerely

Kate Cavanagh Professor of Clinical Psychology

School of Psychology | University of Sussex | Name of Building | Brighton BN1 9QH| United Kingdom kate.cavanagh@sussex.ac.uk www.sussex.ac.uk/psychology Breaking Free Group Williams House Manchester Science Park Lloyd Street North Manchester M15 6SE



Thursday 22<sup>nd</sup> October 2020

Dear Sir/Madam,

I am writing to you to confirm that in her role as CEO of 'Self-Help Services', Nicky provided approval for the Breaking Free Group research team to access a database containing anonymised clinical data from service users at Self-Help Services. This was done in order to facilitate the publication of two studies – references for these studies are listed below. Nicky also provided comments on draft manuscripts outlining the findings of these studies, ahead of their submission for publication in the journals Advances in Dual Diagnosis and BMJ Open.

- Elison, S., Ward, J., Davies, G., Lidbetter, N., Dagley, M., & Hulme, D. (2014). An outcomes study of eTherapy for dual diagnosis using Breaking Free Online. Advances in Dual Diagnosis, 7(2), 52-62.
- Elison, S., Ward, J., Williams, C., Espie, C., Davies, G., Dugdale, S., Ragan, K., Chisnall, L., Lidbetter, N., & Smith, K. (2017). Feasibility of a UK community-based, eTherapy mental health service in Greater Manchester: repeated-measures and between-groups study of 'Living Life to the Full Interactive', 'Sleepio' and 'Breaking Free Online' at 'Self Help Services'. *BMJ Open*, 7(7), 1-10.

Your faithfully,

51.

Dr Sarah Elison-Davies, PhD Research Director, Breaking Free Group

Breaking Free Online Limited Registered office: Innovation Birmingham Campus, Faraday Wharf, Holt Street, Birmingham Science Park, Birmingham, 87 488 Registered in England No. 07441522





Sleep and Circadian Neuroscience Institute, Oxford Molecular Pathology Institute, Sir William Dunn School of Pathology South Parks Road, Oxford OX1 3RE, United Kingdom Web: www.ndon.ox.ac.uk | Tel: +44(0)1865 618661 | Email: colin.espie@ndon.ox.ac.uk PA: toria.summers@ndon.ox.ac.uk

Professor Colin A Espie BSc, MAppSci, PhD, DSc, FRSM, FBPsS, CPsychol, CSci Professor of Sleep Medicine Senior Research Fellow, Somerville College

Our ref: CE/ts

15th September 2020

To whom it may concern:

Dear Sir/Madam

Treating Depression and Anxiety with Digital Cognitive Behavioural Therapy for Insomnia: A Real World NHS Evaluation Using Standardized Outcome Measures

I am writing to confirm that Nicola Lidbetter contributed to the design of the study; leading on the site's strategic delivery of the project at Self Help Services. Nicola also contributed to the final draft of the manuscript and played a critical role in the dissemination of the paper to relevant service user and professional populations.

Yours faithfully

Professor Colle A. Espie BSC, MAppSci, PhD, DSc, FBPsS, CPsychol, CSci Professor of Sleep Medicine Senior Research Fellow, Somerville College Emeritus Professor of Clinical Psychology, University of Glasgow





Division of Nursing, Midwifery and Social Work The University of Manchester Jean McFarlane Building Oxford Road Manchester M13 9PL

tel +44(0)161 306 0260 fax +44(0)161 306 7707 email: nmsw@manchester.ac.uk www.nursing.manchester.ac.uk

## 21st October 2020

## eTherapy peer-reviewed paper contribution

Gellatly, J., Chisnall, L., Seccombe, N., Ragan, K., Lidbetter, N., & Cavanagh, K. (2017). @Home eTherapy Service for People with Common Mental Health Problems: an Evaluation. *Behavioural and Cognitive Psychotherapy*. <u>https://doi.org/10.1017/S1352465817000297</u>

I am writing with respect to the peer-reviewed journal article above, and specifically the contributions that were made by Nicola Lidbetter.

The piece of work conducted was evaluative and was instigated by Nicola who worked closely with myself, other University academics and the eTherapy team to refine aims and objectives of the research, ensuring compliance with service governance regulations. Nicola contributed to the preparation, writing and revisions of the manuscript. She also played a critical role in dissemination of the paper to relevant service user and professional populations.

I would be pleased to provide any further information if required.

Joooto Genany

Dr Judith Gellatly Programme Manager/Research Fellow The University of Manchester



The University of Manchester, Oxford Road, Manchester M13 9PL Royal Charter Number: RC000797